Investigating heart rate variability as a novel non-invasive measure of blood glucose level in type 1 and type 2 diabetes.

by Luke Jarman¹

BMedSc(Hons)

February 2023

Thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy, Faculty of Science, University of Technology Sydney.

Principle supervisor:	Associate Professor Sara Lal ^{1, 2}
Co-supervisors:	Associate Professor Roderick Clifton-Bligh ³
	Dr Najah Nassif ²
	Professor Ann Simpson ²

1. Neuroscience Research Unit, School of Life Sciences, University of Technology Sydney.

2. School of Life Sciences and Centre for Health Technologies, University of Technology Sydney.

3. Kolling Institute of Medical Research, Royal North Shore Hospital, St Leonards.

CERTIFICATE OF ORIGINAL AUTHORSHIP

I, Luke Jarman, declare that this thesis is submitted in fulfilment of the requirements for the award of Doctor of Philosophy (Science), in the School of Life Sciences at the University of Technology Sydney.

This thesis is wholly my own work unless otherwise referenced or acknowledged. In addition, I certify that all information sources and literature used are indicated in the thesis.

This document has not been submitted for qualifications at any other academic institution.

This research is supported by an Australian Government Research Training Program.

Production Note: Signature: Signature removed prior to publication.

Date: 14-Feb-2023

Acknowledgements

My first and biggest acknowledgement goes to my supervisor, Associate Professor Sara Lal. Your strong guidance throughout this PhD has not only improved me as a researcher, but also as a person. You have opened doors for me and helped me develop as a professional amidst a world of challenges, including a global pandemic. Thank you for the feedback on my written work as well as helping develop my oral presentation skills. I won't soon forget the lessons I learnt under your supervision.

I'd also like to thank my co-supervisors, Associate Professor Rory Clifton-Bligh, Dr Najah Nassif, and Professor Ann Simpson, who did not hesitate to rise to the occasion when the demands of this PhD were high. Thank you to Rory as well as Jean Doyle and Liza Nery for your monumental efforts with launching the study at Royal North Shore Hospital, as well as your help with recruitment. Thank you to Ann and Najah for your constant feedback and engagement at research meetings. It is to you all that I owe the sense of accomplishment I feel for this work.

To my colleagues Taryn Chalmers, Ty Lees, and Jaymen Elliott: you have passed on to me enough technical and scientific knowledge to fill a library, and I have done my best to fit some of it in this thesis. There is no chapter which you did not influence in some way, and for that I am incredibly grateful. A big thanks also to Shamona Maharaj, Ardalan Eslami, and George Kalatzis, who have all contributed in their own way, whether it be through peer support, writing advice, or buying me pizza. I valued these equally. It has been my pleasure to watch all my colleagues graduate before me, as each of you have reached out to help me even after moving on from the research unit. I hope to join you all soon as a doctor!

Thank you to Diabetes Australia, who facilitated with the recruitment of participants by allowing me to post an advertisement on their website. I'd like to also acknowledge the Australian Government for granting me the Research Training Program Stipend, which provided me with financial support whilst I conducted this PhD.

To my partner Taryn Hayhurst: my words of appreciation for you could fill another thesis, but this time I'll try to keep it short. Thank you for keeping me motivated, keeping me sane, and being with me every step of this journey.

To those who volunteered their time, patience, and blood to this scientific research: thank you. You had a simple desire to make life better for those that came after you, even if it was only a chance, and I hope you got something out of this research. For those of you who are about to read on, I hope you do too.

Table of Contents

Declaration	i
Acknowledgements	ii
Table of Contents	iii
List of Tables	viii
List of Figures	ix
List of Abbreviations	xi
Abstract	xii
Chapter 1. Introduction	1
1.1 Diabetes Mellitus: A Disease of Civilisation	1
1.2 Epidemiology	3
1.3 Prevalence	4
1.4 Classification	5
1.4.1 Type 1 Diabetes	6
1.4.2 Type 2 Diabetes	7
1.5 Complications	9
1.5.1 Short-Term	9
1.5.1.1 Hyperglycaemia	9
1.5.1.2 Hypoglycaemia	
1.5.1.3 Impaired Awareness of Hypoglycaemia	
1.5.2 Long-Term	
1.5.2.1 Macrovascular	15
1.5.2.2 Microvascular	15
1.5.2.3 Diabetic Autonomic Neuropathy	16
1.5.3 Neurophysiology of Blood Glucose Level and Autonomic Activity	17
1.6 Importance of Glycaemic Management	
1.6.1 Glycaemic Variability	
1.6.2 Oxidative Stress and Hyperglycaemia	19

1.6.	.3	Treatment	
1	.6.3.1	Type 1 Diabetes Treatment	
1	.6.3.2	Type 2 Diabetes Treatment	
1.6.	.4	Homeostasis	
1.6.	.5	Metabolic State	
1.7	Inva	sive Glucose Monitoring: The Current Standard	
1.7.	.1	Blood Glucose Assessment	
1	7.1.1	Strengths	
1	7.1.2	Limitations	
1.7.	.2	Continuous Glucose Monitoring	
1	7.2.1	Strengths	
1	7.2.2	Limitations	
1.7.	.3	Additional Measures	
1	7.3.1	Glucose Tolerance Test	
1	7.3.2	HbA _{1c} Assessment	
1.7.	.4	Achieving Glycaemic Targets	
1.7.	.5	Barriers	40
1	7.5.1	Invasive Technologies	41
1.7.	.6	Ideal Monitoring	
1.8	Non	-Invasive Glucose Monitoring: The Future Standard	
1.8	.1	Absorption Spectroscopy	
1.8	.2	Fluorescence	
1.8	.3	Reverse Iontophoresis	45
1.8	.4	Summary	45
1.9	Gluc	ose Prediction: Novel Applications	46
1.9.	.1	Based on Alternative Measures	46
1	9.1.1	Neural Regulation of Blood Glucose	47
1.10	Hea	rt Rate Variability: An Autonomic Marker	50
1.1	0.1	Applications in Diabetes	52

1.1	0.2	Applications in Glucose Monitoring	. 53
1.11	Basi	is for Research	. 53
1.12	Aim	S	. 59
1.13	Нур	otheses	. 59
Chapter	2.	UTS Pilot Study	. 60
2.1	Met	hods	. 60
2.1.	.1	Subjects	. 60
2.1.	.2	Exclusion Criteria	. 60
2.1.	.3	Blood Pressure	. 61
2.1.	.4	Blood Glucose Assessment	. 62
2.1.	.5	Heart Rate Variability	. 64
2	2.1.5.1	Recording R-R Interval Data by Electrocardiogram	. 64
2	2.1.5.2	2 Extrapolation of Frequency-Domain Measures	. 66
2.1.	.6	Follow-up	. 69
2	.1.6.1	Meal and Additional Assessments	. 70
2	2.1.6.2	2 Food Diary	. 70
2.1.	.7	Statistical Analysis	. 72
2.2	Res	ults	. 73
2.2.	.1	Glucose Level and Blood Pressure	.74
2.2.	.2	Heart Rate Variability Measures	. 77
2.2.	.3	Summary of Results	. 79
2.3	Disc	cussion	. 80
2.3.	.1	Dynamics of Heart Rate Variability	. 80
2.3.	.2	Dynamics of Postprandial Blood Glucose	. 83
2.3.	.3	Strengths and Limitations	. 85
Chapter	3.	UTS T2D Study	. 87
3.1	Met	hods	. 87
3.1.	.1	Subjects	. 87
3.1.	.2	Pre-study Requirements	. 88

3.1.3	Expansion of Previous Methodology	88
3.1.4	Recording R-R Interval Data by Heart Rate Monitor	89
3.1.4.1	Selection of HRV Intervals to be Analysed	90
3.1.5	Postprandial Assessment	91
3.1.6	HbA _{1c} Assessment	92
3.1.7	Statistical Analysis	92
3.2 Res	ults	94
3.2.1	Demographic Differences: Non-Diabetes vs. Diabetes	94
3.2.2	Physiological Differences: Non-Diabetes vs. Diabetes	95
3.2.2.1	Males vs. Females	99
3.2.3	Correlation Analysis	105
3.2.3.1	Partial Correlations	106
3.2.4	Multiple Linear Regression	111
3.2.5	Summary of Results	113
3.3 Disc	cussion	114
3.3.1	Sample Comparison	114
3.3.1.1	l Sex Comparison	116
3.3.2	Correlation Analysis and Confounders	117
3.3.3	Heart Rate Variability Associations with Fasting Blood Glucose Level	117
3.3.3.1	Non-Diabetes Sample	117
3.3.3.2	2 Diabetes Sample	118
3.3.4	Heart Rate Variability Associations with Postprandial Blood Glucose Level	119
3.3.4.1	Non-Diabetes Sample	120
3.3.4.2	2 Diabetes Sample	120
3.3.5	General Relationship of Blood Glucose Level and Autonomic Activity	121
3.3.5.1	Autonomic Response to Food Intake	122
3.3.6	Strengths and Limitations	122
Chapter 4.	RNSH T1D Study	127
4.1 Met	hods	127

4.1.1	Subjects	127
4.1.2	Pre-study Requirements	128
4.1.3	Continuous Glucose Monitoring	129
4.1.4	Recording of R-R Interval Data by Heart Rate Monitor	129
4.1.5	Meals, Medications, and Physical Activity Diary	130
4.1.6	Selection of Glycaemic Excursions and R-R Interval Data	130
4.1.7	Measures of Heart Rate Variability	131
4.1.8	Statistical Analysis	132
4.2 Res	ults	133
4.2.1	Preliminary Correlation Analysis	134
4.2.2	Hypoglycaemia	136
4.2.3	Hyperglycaemia	140
4.2.4	Summary of Results	145
4.3 Disc	ussion	146
4.3.1	Preliminary Analysis of the Sample	146
4.3.2	Changes in Autonomic Activity during Hypoglycaemia	147
4.3.2.1	Clinically Significant Hypoglycaemia	149
4.3.2.2	Neurophysiology of Hypoglycaemia	151
4.3.3	Changes in Autonomic Activity during Hyperglycaemia	152
4.3.3.1	Neurophysiology of Hyperglycaemia	153
4.3.4	Strengths and Limitations	154
Chapter 5.	Conclusion	157
5.1 Futu	are Directions	159
Chapter 6.	References	163

List of Tables

Table 1.1 Ranges of blood glucose levels that are considered 'optimal, 'impaired', 'pre-diabetes'
or 'diabetes', as determined by fasting blood glucose and 2-hour glucose tolerance test
Table 1.2 Summary of key findings from studies assessing frequency-domain heart rate
variability parameters and blood glucose levels in groups with diabetes and groups without
diabetes
Table 2.1 Blood glucose and blood pressure means and standard deviations recorded in the UTS
Pilot Study sample (n=14)75
Table 2.2 Heart rate variability measures means and standard deviations recorded in the UTS
Pilot Study sample (n=14)77
Table 3.1 Means and standard deviations for demographic differences between the group
without diabetes (n=29) and the group with diabetes (n=27)
Table 3.2 Means and standard deviations for physiological differences between the group
without diabetes (n=29) and group with diabetes (n=27)
Table 3.3 Demographic differences between males (n=15) and females (n=14) without diabetes,
as well as males (n=13) and females with diabetes (n=14)100
Table 3.4 Physiological differences between males (n=15) and females (n=14) without diabetes,
as well as males (n=13) and females with diabetes (n=14)101
Table 3.5 Partial Pearson's correlations between heart rate variability measures and blood
glucose level, accounting for age, body mass index, and kilojoule intake as covariates in the
group without diabetes (n=29) and the diabetes group (n=27)107
Table 3.6 Regression analysis for postprandial blood glucose and the significantly correlated
heart rate variability measures in the group without diabetes (n=29)111
Table 3.7 Regression analysis for fasting blood glucose and the significantly correlated heart
rate variability measures in the diabetes group (n=27)112
Table 3.8 Regression analysis for postprandial blood glucose and the significantly correlated
heart rate variability measures in the diabetes group (n=27).
Table 4.1 Means and standard deviations for demographic and physiological features of the
sample (n=15)134
Table 4.2 Mean heart rate variability measures ($n=22$) at 10 minutes prior, during, and 10
minutes after hypoglycaemia
Table 4.3 Mean heart rate variability measures ($n=33$) at 10 minutes prior, during, and 10
minutes after hyperglycaemia

List of Figures

Figure 1.1 Total number of global deaths from leading causes in 2000 and 20192
Figure 1.2 Prevalence of diabetes in major regions between 1980 and 20144
Figure 1.3 Interaction of traditional and non-traditional risk factors in the progression of
cardiovascular disease
Figure 1.4 Summary of the short and long-term complications of diabetes and how they relate to
diabetes awareness and long-term management14
Figure 1.5 Flow chart for the pathogenesis of tissue damage in chronic hyperglycaemia
Figure 1.6 Anatomical structures involved in glucose, insulin, and glucagon regulation
Figure 1.7 Hormonal regulation of blood glucose level in people without diabetes
Figure 1.8 Measurement of interstitial glucose concentration using a continuous glucose
monitoring system
Figure 1.9 Line graph comparing the glucose levels recorded by a continuous glucose
monitoring system, a glucometer, and a laboratory glucose analyzer
Figure 1.10 Current methods for invasive and non-invasive measurements of blood glucose 44
Figure 1.11 Sympathetic and parasympathetic innervation of the human body
Figure 1.12 Electrocardiogram waveform with QRS complex and R-R intervals
Figure 2.1 Example of participant with an automated blood pressure monitor
Figure 2.2 Example of a participant undergoing a blood glucose assessment
Figure 2.3 Einthoven electrode placement for the 10-minute electrocardiogram recording 65
Figure 2.4 Plotting the time-series graph from the electrocardiogram waveform
Figure 2.5 Frequency spectrogram derived from fast Fourier transform
Figure 2.6 Flow chart for extrapolation of heart rate variability from electrocardiogram data69
Figure 2.7 Protocol summary for the UTS Pilot Study
Figure 2.8 Effect of electrocardiogram noise on extrapolating heart rate variability measures. 74
Figure 2.9 Blood glucose, blood pressure, and heart rate data for the three assessments (n=14).
Figure 2.10 Change in the mean blood glucose level and mean low frequency, high frequency,
and total power between the three time points (n=14)78
Figure 3.1 Application of BodyGuard 2 heart rate monitor
Figure 3.2 Timing of assessments in the UTS T2D Study
Figure 3.3 Comparison of systolic and diastolic blood pressure between participants without
diabetes (n=29) and diabetes participants (n=27)
Figure 3.4 Comparison of blood glucose level and kilojoule intake between the group without
diabetes (n=29) and the diabetes group (n=27)

Figure 3.5 Comparison of heart rate variability measures between non-diabetes (n=29) and
diabetes groups (n=27)
Figure 3.6 Differences in blood pressure between males $(n=15)$ and females $(n=14)$ without
diabetes as well as males (n=13) and females with diabetes (n=14)102
Figure 3.7 Differences in blood glucose level between males (n=15) and females (n=14) without
diabetes, as well as males (n=13) and females with diabetes (n=14)103
Figure 3.8 Differences in heart rate variability measures between males (n=15) and females
(n=14) without diabetes104
Figure 3.9 Differences in heart rate variability measures between males with diabetes (n=13)
and females with diabetes (n=14)104
Figure 3.10 Scatter plots for postprandial blood glucose level and significantly correlated heart
rate variability measures in the sample without diabetes (n=29)108
Figure 3.11 Scatter plots for fasting blood glucose level and significantly correlated heart rate
variability measures in the diabetes sample (n=27)109
Figure 3.12 Scatter plots for postprandial blood glucose level and significantly correlated heart
rate variability measures in the diabetes sample (n=27)110
Figure 4.1 Scatter plots of glucose level versus various heart rate variability measures for
Participant 01 (n=96)135
Figure 4.2 Change in heart rate variability before and after hypoglycaemia in three events 136
Figure 4.3 Comparison of mean low frequency, high frequency, and total power at three time
points during hypoglycaemia (n=22)138
Figure 4.4 Comparison of mean low to high frequency ratio and normalised low frequency
power at three time points during hypoglycaemia (n=22)139
Figure 4.5 Comparison of time-domain measures at three time points during hypogylcaemia
(n=22)140
Figure 4.6 Change in heart rate variability before and after hyperglycaemia in three events141
Figure 4.7 Comparison of mean low frequency, high frequency, and total power at three time
points during hyperglycaemia (n=33)143
Figure 4.8 Comparison of mean low to high frequency ratio and normalised low frequency
power at three time points during hyperglycaemia (n=33)144
Figure 4.9 Comparison of time-domain measures at three time points during hyperglycaemia
(n=33)145
Figure 4.10 Proportion of asymptomatic hypoglycaemia at different definitions

List of Abbreviations

BGL = Blood glucose level	p = P-value	
BMI = Body mass index	PSD = Power spectral density	
BP = Blood pressure	QRS = QRS complex (electrocardiogram	
bpm = Beats per minute	waveform)	
CGM = Continuous glucose monitoring	r = Correlation coefficient	
ECG = Electrocardiogram	RMSSD = Root mean square of successive	
HLA = Human leukocyte antigen.	RNSH = Royal North Shore Hospital	
HF = High frequency	ROS = Reactive oxygen species.	
HFnu = Normalised high frequency	RR = R-R interval	
HREC = Human Research Ethics Committee	s = Second	
HRV = Heart rate variability	\mathbf{s}^2 = Second squared	
Hz = Hertz	SDNN = Standard deviation of NN intervals.	
kg/m ² = Kilogram per square metre	SMBG = Self-monitoring of blood glucose	
kJ = Kilojoule	T1D = Type 1 diabetes	
LAQ = Lifestyle Appraisal Questionnaire	T2D = Type 2 diabetes	
LF = Low frequency	UTS = University of Technology Sydney	
LF:HF = Low to high frequency ratio	VLF = Very low frequency	
LFnu = Normalised low frequency		
min = Minutes		
mmHg = Milligrams of mercury		
mmol/L = Millimoles per litre		
ms = Milliseconds		
ms ² = Milliseconds squared		

Abstract

Diabetes mellitus is costly to both individuals and health systems and affects almost 500 million adults worldwide. The scientific literature strongly advocates that maintaining blood glucose level (BGL) within the optimal range as the best means of reducing diabetes complications and improving quality of life and health outcomes for those living with diabetes. Optimal management and monitoring of BGL may be significantly improved by combining data on an individual's heart rate variability (HRV) – a non-invasive autonomic marker – with data from currently-available invasive glucose monitoring systems. As part of this PhD candidature and thesis, groups of people with type 1 diabetes, type 2 diabetes, or without any chronic illness, were recruited to investigate correlations between HRV measures and BGL. Correlation analysis demonstrated that multiple different HRV measures were significantly and inversely correlated with BGL measured in a fasting and postprandial state. Multiple linear regression analyses determined that HRV measures account for 27-55% of the total variation in BGL measured in different metabolic states. However, further research is needed. Continuous glucose monitoring (CGM) remains the gold-standard for measurement of BGL in diabetes, however HRV represents a promising area which can add to detection of diabetes and glycaemic events in tandem with CGM.

Chapter 1. Introduction

With the advent of the 21st century, new challenges have emerged for medical science. Driven by global shifts in culture and influence, as well as technological advancements related to food, travel, and entertainment, the lifestyles of modern humans have strayed from those of our Palaeolithic ancestors. Eaten and Konner (1985) were among the first to suggest that though the habits and diets of humans have evolved dramatically over time, the genetic traits passed down with each generation have not experienced the same pressures of natural selection as our Palaeolithic predecessors.¹ As such, the genome of modern humans is more suited to the huntergatherer lifestyles that were adapted over 12,000 years ago,² resulting in a 'gene-culture' misalignment in the 21st century. In light of this, there is growing recognition in the scientific community that the discordance between modern and Palaeolithic lifestyles is associated with the current rise in 'diseases of civilisation', though the extent to which this is true is debated within the literature.^{3, 4} Regardless, diseases attributed to modern ways of life are emerging at alarming rates and represent a significant burden to society, and this must be addressed.

1.1 Diabetes Mellitus: A Disease of Civilisation

There is a growing awareness of the rise of non-communicable diseases related to modern civilisation. In the 1970s, epidemiologists observed an increase in the prevalence of obesity, cardiovascular diseases, and type 2 diabetes (T2D) among indigenous populations that had renounced their traditional lifestyles in favour of 'Westernised' lifestyles,^{5, 6} and recent research conducted by Balick and colleagues (2019) supports these findings.⁷ This causal relationship was reinforced by O'dea (1984) who demonstrated that Australian Aborigines with T2D who temporarily reverted to their highly-active hunter-gatherer lifestyle and low energy density diet showed significant improvements in blood pressure, glucose tolerance, and lipid profiles.⁸ This has also been ratified by more recent research conducted by Frassetto and colleagues (2009).9 Despite this evidence, adherence to traditional hunter-gatherer diets, which are protective factors against diseases of civilisation, remains in decline.¹⁰ The literature is generally critical of the habits and diets characteristic of Western civilisation, defined by a high-fat, high-sugar diet and a largely sedentary lifestyle lacking in meaningful levels of physical activity.^{10, 11} Though some authors contend that trends in physical inactivity have changed little over the last 30-40 years,¹² there is a consensus that the widely popular Western diet is an underlying cause of many emerging diseases of civilisation.^{13, 14} Because of this spread, these diseases of civilization are not endemic to Western nations, and many have reached epidemic proportions worldwide.15,16

In the past 20 years, non-communicable diseases have overtaken infectious and parasitic diseases as the leading causes of mortality worldwide (Figure 1.1).¹⁷ Although advancements in medical science have generally improved life expectancy and mortality rates worldwide,^{18, 19} the burden of disease remains high.²⁰ This may also be attributed to the fact that humans are living longer on average, and age is a risk factor for many lifestyle-related diseases.²¹ Due to commercial pressures to consume products high in fat and sugar, as well as social pressures to engage in cigarette and alcohol consumption, non-communicable diseases have replaced infectious diseases as the main causes of mortality worldwide, and will present a major challenge to medical science for the foreseeable future.²² Without adoption of better lifestyles and behaviours, which would require significant cultural changes, it is projected that the burden of lifestyle diseases will continue to grow. T2D is but one of many conditions associated with civilisation. However, it is the fourth biggest cause of lifestyle-related mortality worldwide,²² and is considered one of the main causes of premature illness and mortality.^{23, 24} The objective of this initial chapter is to establish why current management of diabetes mellitus – both type 1 diabetes (T1D) and T2D – is inadequate given its scope, and how the research conducted as part of this PhD candidature may improve it.



Figure 1.1 Total number of global deaths from leading causes in 2000 and 2019.

Figure 1.1 illustrates the change in the number of deaths from the leading causes of death in the year 2000 (blue) compared to 2019 (red). Communicable diseases are indicated by the blue lines, and include lower respiratory infections, neonatal conditions, and diarrheal diseases. All others shown are non-communicable diseases and are indicated by pink lines. Non-communicable diseases caused higher mortality rates in 2019 compared to 2000, and communicable diseases caused lower rates, indicating a shift in the causes of mortality in recent decades. Adapted from World Health Organization (2020).¹⁷

1.2 Epidemiology

Diabetes mellitus - referred to hereon as diabetes - is a group of metabolic disorders characterised by chronic hyperglycaemia, or chronically-elevated levels of blood glucose.²⁵ Though diabetes has been introduced in this thesis as a Disease of Civilization, this is only true for T2D, which constitutes the vast majority of diabetes cases. Other types, such as T1D which will be distinguished in a later section, are not typically considered 'lifestyle' disorders. However, all types of diabetes share a common end point of dysglycaemia, or abnormal blood glucose, and this is the focus of the thesis. For this reason, the term 'diabetes' is used in this thesis in the context of dysglycaemia and refers to both T1D and T2D. Ranges for blood glucose level (BGL) indicative of diabetes are provided in Table 1.1. In the past, discrepancies in the classification of diabetes between the American Diabetes Association (ADA) and the World Health Organization (WHO) - two leading global authorities on diabetes - caused inconsistencies in epidemiological data.²⁶ Estimates of some types of diabetes were affected more severely as they were sensitive to the identifying criteria.²⁷ Now, WHO and ADA use consistent criteria for diagnosing diabetes, providing a gold-standard approach for identification.^{28, 29} These criteria are the most widely-accepted ranges for optimal and suboptimal BGL, also referred to as plasma glucose concentration.

Table 1.1 Ranges of blood glucose levels that are considered 'optimal, 'impaired', 'pre-diabetes' or 'diabetes', as determined by fasting blood glucose and 2-hour glucose tolerance test. Fasting is defined as no caloric intake for at least 8 hours. Individuals need to meet criteria for both fasting and 2-hour glucose tolerance ranges for diagnosis. * = 2-hour glucose tolerance test is standardised by 75 grams of glucose solution. **mmol/L** = Millimoles per litre. Adapted from the American Diabetes Association (2016) and the World Health Organization (2016).^{28, 29}

Diagnosis	Fasting blood glucose (mmol/L)	2-hour glucose tolerance test (mmol/L) *
Optimal	4.0 – 5.5	≤ 7.7
Impaired fasting glucose	6.1 - 6.9	≤ 7.7
Pre-diabetes	5.6 - 6.9	7.8 - 11.1
Diabetes	≥ 7.0	≥ 11.1

Epidemiological evidence suggests that fasting BGL has been on the rise since 1980 (n=2,700,000), increasing by 0.07 millimoles per litre (mmol/L) in men and 0.09 mmol/L in women per decade on average.³⁰ The current burden of this is already high – the global health expenditure for diabetes was estimated to be 850 billion USD in 2017.³¹ On an individual level, diabetes is also the cause of significant emotional distress and financial burden,³² which is due

to a combination of medical costs and reduced capacity for work.³³ Health expenditure for people living with diabetes is higher for individuals with complications, such as retinopathy and kidney failure, though this depends on various factors such as the type of diabetes and it's duration.³⁴ To better understand the scope of diabetes, the following sections will discuss the specific attributes which need to be addressed, such as the rising prevalence, concerns with classification, short and long term complications, and limitations of current treatment.

1.3 Prevalence

According to a systematic literature review of 221 data sources conducted by the International Diabetes Federation, the percentage of people affected by diabetes is expected to increase from 8.4% in 2017 to 9.9% by 2045.³¹ Figure 1.2 shows that total cases have been steadily rising for the past 30 years. Note the large increase in the Eastern Mediterranean region – the origin of the Mediterranean diet, one of the healthiest modern diets^{35, 36} In their 2011 paper, Musaiger reasoned that this was because many people living in the Mediterranean region, particularly those in Arab countries, are renouncing their traditional diet in favour of the Western diet.³⁷ Prevalence may also be rising as a result of increased screening in some regions.



Figure 1.2 Prevalence of diabetes in major regions between 1980 and 2014.

Figure 1.2 portrays the rising prevalence of diabetes across six major regions worldwide, from 1980 to 2014. Diabetes in the South-East Asia Region, of which Australia is part, has been steadily rising over the past 30 years. The Eastern Mediterranean Region is undergoing the highest increase in new cases of diabetes worldwide. The black dotted line indicates average prevalence of diabetes across all regions. Image from World Health Organization (2016).²⁹

Luke Jarman | 4

Data on the prevalence of diabetes in Australia is reported every six months by the National Diabetes Services Scheme, an Australian Government initiative which is administered by Diabetes Australia. As of the 31st of December 2021, there were 1,426,245 people with diabetes registered. Of those, 9% had T1D, 87% had T2D, 3% had gestational diabetes, and less than 1% had some other type of diabetes.³⁸ Per census data provided by the Australia Bureau of Statistics, the population of Australia at the end of 2021 was 25,766,605.³⁹ As such, at the end of 2021, 5.54% of all Australians had some type of diabetes. Other studies report lower rates of T1D, such as a recent study in the United States which found that 5.8% of all diabetes cases were T1D and 90.9% were T2D.⁴⁰ This discrepancy may be due to many cases of 'insulin-treated T2D' being misclassified as 'T1D' in the Australian Health Survey and National Diabetes Services Scheme. Discrepancies may also arise in cases of 'double diabetes', where people with T1D may also develop metabolic syndrome, which shares clinical features with T2D such as insulin resistance.⁴¹ As such, an individual with T1D may register as both someone with T1D and T2D in the prevalence reports. This raises an important issue with diabetes classification in the literature: many cases of diabetes do not fit perfectly into a single category, and there is some overlap between the different types of diabetes.⁴² Future studies should recognise these problems with diabetes classification when approaching study design and when distinguishing sample groups.

1.4 Classification

Both ADA and WHO recognise two main types of diabetes which constitute majority of cases: T1D and T2D.^{28, 29} Largely, they are caused by defects in multiple genes and, in the case of T2D, also lifestyle risk factors such as obesity.⁴³ Less common forms of diabetes exist, such as: monogenic forms of diabetes which relate to a defect in a single gene; secondary diabetes which can result from cystic fibrosis or other medical conditions; and latent autoimmune diabetes in adults.⁴⁴ There is an emphasis in the literature on separating non-classical forms of diabetes from T1D and T2D, as they have distinguishable clinical manifestations and treatments.⁴⁵ Given that T1D and T2D are the most common types of diabetes and collectively account for over 95% of all types of diabetes, these will be the focus of this thesis, and any research conducted as part of the PhD candidature will focus on T1D and T2D. This chapter will also provide evidence as to why T1D and T2D should be studied independently, including the relevance of their distinct aetiology and pathophysiology.⁴⁶

1.4.1 Type 1 Diabetes

T1D and T2D are characterised by a dysfunction of insulin which results in long-term hyperglycaemia. For T1D, this dysfunction is due to a total or near-total deficiency in insulin production.⁴⁷ In short, insulin is a hormone which promotes glucose uptake by somatic cells to reduce levels of glucose circulating in the blood. As a result, T1D is associated with chronically-elevated levels of plasma glucose, as the body lacks insulin-related mechanisms to maintain glucose homeostasis.⁴⁸ The exact mechanisms of T1D pathophysiology are still widely debated, and this has inhibited the development of effective preventive measures.⁴⁹ However, there is a general consensus that insulin depletion is caused by immune-mediated destruction of insulin-producing β -cells in the pancreas.^{50, 51} Various genetic, epigenetic, and exogenous risk factors have been implicated in the development of T1D, however causative mechanisms are poorly-understood.^{52, 53} This may be because of the sheer number of regions of the human genome which confer susceptibility to the condition, which make it difficult to isolate a single cause, as well as the fact that a human pancreas cannot be studied non-invasively in living subjects.⁵⁴

The human leukocyte antigen (HLA) complex is comprised of 128 genes densely clustered on chromosome-6, and variations in this complex can explain 35-50% of the genetic predisposition to T1D.⁵⁵ Familial clustering is prominent in T1D, and the HLA haplotypes which predispose one to T1D are the strongest genetic association to T1D identified to date. However, genetic studies have not been able to identify specific genes of the HLA complex which cause T1D, and there are various non-HLA genes implicated in the pathogenesis of T1D.⁵⁶ Increased expression of HLA-1 antigens in the insulin-producing beta cells of the pancreas has been associated with the onset of T1D. It is hypothesised in recent work that HLA-1 antigens increase the action of influent CD8⁺ T cells specific to defined islet antigens.⁵⁷ In this autoimmune model, CD8⁺ T cells target the beta cells of the pancreas and destroy them, leading to a deficiency in insulin production and thus dysglycaemia.

Studies on hereditary links in T1D, including the inheritance of susceptibility genes, are criticised as being underpowered and thus insufficient to detect complex interactions between genes.⁵⁸ Exogenous or environmental agents represent a divided area of study within the literature, as those agents that have been identified so far vary significantly. These include: infection by enteroviruses or retroviruses, pathological gut flora, consumption of certain milk proteins, and exposure to environmental pollutants.⁵⁹ A range of genetic and environmental factors are involved in the pathogenesis of T1D, and future research may need to focus on developing current understandings so that more therapeutic options for T1D may be explored.⁶⁰

1.4.2 Type 2 Diabetes

As with T1D, T2D is associated with a dysfunction of insulin that causes chronically high levels of blood glucose. By comparison, T2D is associated with a progressive impairment of insulin secretion and efficacy,²⁸ which leads to profound, chronic elevations in BGL.⁶¹ Associated with the pathogenesis of T2D is insulin resistance, a term which refers to the way in which long term over-production of insulin, which may be exacerbated by high carbohydrate load of diets, leads to a progressive loss of sensitivity to insulin. The risk factors for developing T2D are more well-known than T1D and, as such, T2D is theoretically easier to prevent compared to T1D.⁶² This also facilitates health programs and strategies to target groups at risk of T2D, such as overweight individuals and indigenous populations. Other groups at risk include the elderly, smokers, and heavy drinkers, though generally preventative programs are not targeted at these populations in the context of preventing diabetes.⁶³ There is also a large genetic component to T2D, and studies in this area need to improve sample sizes, diversify populations, and improve phenotyping and sub-phenotyping.⁶⁴

The traditional risk factors for T2D are well-established, and include increased waist circumference, obesity, smoking, high blood pressure, and physical inactivity.^{65, 66} Large-scale clinical trials have shown promising results in reducing the prevalence of T2D in at-risk groups by targeting these specific lifestyle-related risk factors.^{67, 68} There are consistent findings that proper management of blood glucose and blood pressure, as well as smoking cessation and weight loss, are important strategies for managing T2D and its complications such as cardiovascular disease.^{69, 70} Recent literature also advocates the importance of non-traditional risk factors for cardiovascular disease (Figure 1.3), which are also implicated in T2D. These include insulin resistance, glycaemic variability, thrombogenic factors, and others.⁷¹ In particular, insulin resistance and glycaemic variability are non-traditional risk factors for cardiovascular disease which are also important in the prevention of microvascular and macrovascular complications of diabetes. Future studies may need to consider the complex interactions of these risk factors in the pathogenesis of T2D.





Figure 1.3 shows the complex relationships between the various risk factors present in T2D that are also implicated in cardiovascular disease. Traditional risk factors include: 'increased blood pressure', which has a direct correlation with cardiovascular disease; 'visceral obesity'; and certain 'genetic factors'. The other risk factors featured in this figure are considered non-traditional and represent an emerging awareness of the complex interactions between risk factors and disease onset. Adapted from Martín-Timón et al. (2014).⁷¹

Evidently, the two main types of diabetes share few similarities in terms of pathophysiology. Due to the spread of poor lifestyles, traditional risk factors for T2D have reached epidemic proportions. Though the mechanisms of its aetiology are debated, T1D is also rising. Reasons for this increase in prevalence are also debated, as correlations between T1D and proposed risk factors such as air pollution and certain infections have been inconclusive,^{72, 73} supporting the need for further research. Though T1D constitutes a smaller proportion of all cases of diabetes at only 9%, this statistic has remained static in recent years. This indicates the prevalence of T1D is also increasing in line with the T2D epidemic,⁴⁰ which is driven by the rise of traditional risk factors for T2D such as obesity and hypertension.⁴³ According to the literature, this level of comorbidity and interconnectedness between lifestyle-related diseases and risk factors is difficult to quantify. Additionally, living with diabetes is a risk factor for developing complications of diabetes. More specifically, the severity of complications of diabetes is associated with the amount of time an individual lives with suboptimal levels of BGL.⁷⁴ This chapter will now explore the short and long-term complications of diabetes.

1.5 Complications

Complications of diabetes may arise in the short-term and the long-term, and are the direct consequence of hyperglycaemia.⁷⁵ Maintaining optimal BGL reduces the risk for these complications. However, for reasons that will be discussed later in this chapter, many people living with diabetes are unable to optimise their BGL sufficiently. Some reasons include intercurrent illness and infection, or as a consequence of treatment for dysglycaemia. Acute symptoms can arise when BGL rises above the optimal upper limit of 7.7 mmol/L, including acute hyperglycaemia, diabetic ketoacidosis, and the hyperglycaemic hyperosmolar state, and when BGL drops below the lower limit of 4.0 mmol/L, which is known as hypoglycaemia. Roughly one third of all hospitalisations caused by diabetes are due to unmanaged diabetes conditions, representing a significant burden to health care systems.⁷⁶

1.5.1 Short-Term

As one of the defining features of diabetes, persistent hyperglycaemia is the most common symptom experienced by people living with diabetes.⁷⁷ Acute symptoms of hyperglycaemia are generally mild, and include feelings of hunger and thirst, as well as frequent urination.⁷⁸ The lack of seriousness of these symptoms, combined with the fact that they generally develop slowly over time in T2D, contributes to unawareness of diabetes. In turn, the lack of awareness that these symptoms may be related to diabetes contributes to the large percentage of untreated hyperglycaemia.

1.5.1.1 Hyperglycaemia

In cases of unmanaged or undiagnosed diabetes, the deficiency of insulin activity can result in very high levels of glycaemia, which can cause two serious acute complications: diabetic ketoacidosis, which is particularly prevalent in T1D, and the hyperglycaemic hyperosmolar state, which is the more common presentation for people with T2D. Ketoacidosis is a complication of acute insulin deficiency and is a common acute complication for people living with T1D.⁷⁹ It is associated with high levels of circulating ketone bodies (hyperketonaemia) which can cause metabolic acidosis, and people usually experience excessive hunger, thirst, and weight loss.⁸⁰ In their systematic review of 19 studies, Marcovecchio (2017) concluded that the prevalence of diabetic ketoacidosis was higher in women, younger people, and in people treated with insulin injections.⁸¹ Most present with a BGL greater than 16.7 mmol/L, and treatment involves insulin therapy to correct the hyperglycaemia, as well as replacing lost fluids.⁸² The

literature is critical of additional supplement treatments such as bicarbonate infusion, as they are almost never required. For example, a systematic review of 12 randomized trials conducted by Chua, Schneider, and Bellomo (2011) demonstrated that bicarbonate infusion did not improve treatment outcomes in people with diabetic ketoacidosis.⁸³ Another acute complication of hyperglycaemia is the hyperglycaemic hyperosmolar state, and treatment involves insulin therapy and correction of fluid deficit.⁸² Though ketoacidosis does occur in people with T2D, the more common presentation of acute hyperglycaemia is the hyperglycaemic hyperosmolar state. Diagnosis of this condition is when BGL is greater than 33.3 mmol/L with no metabolic acidosis or ketonemia, and the mortality rate is between 10-20%, or 10 times higher than the rate in people with diabetic ketoacidosis.⁸⁴ Stringent monitoring and regulation of BGL is important for prevention.

1.5.1.2 Hypoglycaemia

Just as acute complications can arise from BGL rising above the optimal range, complications can be caused by BGL falling below the optimal lower limit of 4.0 mmol/L. This is more common in cases of diabetes where there is a total dependence on insulin to be supplied exogenously, or externally.⁸⁵ Such cases are referred to as 'insulin-dependent diabetes', and do not necessarily refer to T1D, though most cases of T1D are insulin dependent, as well as the more severe cases of T2D. As a result of this trend, severe hypoglycaemia – defined as an event requiring third party assistance – occurs more commonly in T1D compared to T2D. The Global HAT study (n=27,585) reported that people with T1D experience hypoglycaemia at a rate of 73.3 events per year, compared to 19.3 per year for people with T2D. In terms of severe hypoglycaemia, people with T1D experience 4.9 events per year compared to 2.5 for insulin-treated T2D patients.⁸⁶

People with insulin-dependent diabetes are educated on how to measure insulin doses and selfadminister them to maintain their BGL in a euglycaemic, or optimal, range. However, due to the multitude of day-to-day factors which influence BGL, including exercise and food intake, this euglycaemic target can be difficult to maintain. It is commonplace for exogenous insulin to reduce BGL below the optimal range, causing hypoglycaemia. This is sometimes referred to as iatrogenic hypoglycaemia because it results from an error in medical treatment. Richard and colleagues (2019) cite three main causes of severe hypoglycaemia in diabetes as missed meals, incorrect use of antidiabetic medication, and mismatch between antidiabetic medication and carbohydrate intake.⁸⁷ This is reinforced by Tourkmani and colleagues (2018), who also add physical exercise and alcohol consumption as common causes.⁸⁸ Confusion, difficulty concentrating, and dizziness are known neuroglycopenic symptoms of hypoglycaemia, and occur because glucose is the major source of energy for the brain. Sweating, heart palpitations, and trembling are the autonomic symptoms of hypoglycaemia, and usually occur before the onset of neuroglycopenic symptoms and can act as a 'warning system'.⁸⁹ Coma and death result from prolonged severe hypoglycaemia, and loss of consciousness may precede these at undefined levels of BGL.⁹⁰ As such, the literature advocates that people living with T1D be aware of the autonomic signs and symptoms of hypoglycaemia and be prepared with a plan of action,⁸⁸ for example having a snack containing 15g of carbohydrate and a protein source in between meals to prevent repeated hypoglycaemia.⁸⁵ The literature emphasizes the importance of optimizing glycaemia without compromising the safety of people with severe hypoglycaemic events, though this is a difficult balance to maintain.⁹¹

1.5.1.3 Impaired Awareness of Hypoglycaemia

Impaired awareness of hypoglycaemia, also known as hypoglycaemia unawareness, is a complication of diabetes that results from repeated iatrogenic hypoglycaemia. It involves reduced responsiveness of the autonomic nervous system to low BGL, resulting in a loss of the body's natural alarm system for hypoglycaemia.⁹² People with this impaired awareness are six times more likely to experience severe hypoglycaemic events compared to those with optimal awareness.⁹³ In a study of n=98 young adults with T1D, conducted by Paes and colleagues (2020), hypoglycaemia unawareness was observed in 28% of the subjects.⁹⁴ In a cross-sectional study on T2D participants undergoing insulin therapy (n=2,350), Meijel and colleagues (2018) observed impaired awareness of hypoglycaemia in nearly 10% of the sample.⁹⁵ Evidently, this unawareness affects a significant proportion of people with diabetes, roughly one in four people with T1D, and roughly one in ten of those with T2D on insulin therapy. This is especially concerning when considering the threat level of hypoglycaemia. Frequent hypoglycaemia leading to hypoglycaemia unawareness is reversible in the short term through stringent avoidance of hypoglycaemia for 2-3 weeks.⁹² However, as demonstrated in the HypoCOMPass trial (n=96), awareness of hypoglycaemia was restored after 24-weeks without compromising glycaemic target, and thus autonomic function may take up to 24 weeks or longer to be restored.96

One hypothesis is that iatrogenic hypoglycaemia, caused by pharmacotherapy and other factors, is a major cause of hypoglycaemia-associated autonomic failure. According to Cryer (1992), the reduced ability of the autonomic nervous system to activate or respond to critically low BGL reduces the ability of people who rely on autonomic symptoms to self-monitor their own BGL,

leading to recurrent severe hypoglycaemia.⁹⁷ Under ideal circumstances, people can recognize the onset of sweating and heart palpitations as early warning indicators of hypoglycaemia. The absence of these indicators creates a cycle where unmanaged hypoglycaemia predisposes one to subsequent instances of hypoglycaemia, further removing defenses against detecting it. This mechanism is still largely misunderstood, and a recent review published by Rickels (2019) determined that there was a lack of consensus on this pathological mechanism.⁹⁸ Though autonomic failure is involved in hypoglycaemia unawareness, it is unlikely to be related to cardiac autonomic neuropathy, a long-term complication of diabetes. It has been shown that only subject age and duration of diabetes is associated with hypoglycaemia unawareness, and cardiac autonomic neuropathy is not.⁹⁴ Research indicates that people commonly make errors when estimating BGL in very high ranges, such as above 22.0 mmol/L, or in the very low ranges, such as below 3.0 mmol/L, due to the presence of neuroglycopenic symptoms which affect concentration and alertness.⁹⁹ As such, self-diagnosis of low BGL can be unreliable, and future research may aim to improve this.

Evidently, the threat of BGL falling below the optimal lower limit is of great concern. A simple solution to this problem is to implement a system of continuous monitoring as part of standard diabetes management. In theory, continuous monitoring is currently the most reliable means of detecting hypoglycaemia early and facilitating self-intervention,¹⁰⁰ but in practice it is expensive and usually reserved for T1D. A more detailed discussion of the strengths and limitations of current glucose monitoring will be provided in a later section. It should be noted at this point that the literature advocates strongly for stringent monitoring of glycaemic events, as it is also associated with improved long-term outcomes,¹⁰¹ which will be explored in Section 1.5.2.

1.5.2 Long-Term

In the long-term, unmanaged BGL can also lead to life-threatening conditions, as well as increased morbidity and mortality. For hypoglycaemia, the long-term consequences are poorly understood, and this an area of research that is severely lacking. Generally, the main long-term complications of hypoglycaemia are impaired awareness of hypoglycaemia and hypoglycaemia-associated autonomic failure, as well as predisposition to cardiovascular events.¹⁰² Fortunately, almost all episodes of hypoglycaemia occur in people that are taking medication and are thus aware of the risk of iatrogenic hypoglycaemia, so there is increased opportunity to intervene.⁸⁷

As for long-term complications of hyperglycaemia, there is an abundance of research in the literature, and it defines these complications as the result of various factors. Based on estimates provided by Beagley and colleagues (2014), 46% of all people living with diabetes worldwide

are undiagnosed or unaware of their condition, and are thus likely to be untreated cases.¹⁰³ The lack of diagnosis is due to progressive increase in BGL over time in people with T2D. As the condition progresses, the individual had an adaptive physiological response to the high levels of glucose which become the new 'normal' for them. As such, symptoms may be absent despite high BGL.⁸². Even in individuals with symptoms, medical intervention may not be a priority for symptoms such as increased thirst and hunger. However, sustained hyperglycaemia causes serious complications in the long term. Among those who are aware of their condition (56% of total diabetes cases worldwide), glycaemic management is better, but not optimal. In an Australian study, roughly half of a large population of diagnosed diabetes cases were unable to maintain their BGL at the optimal target level, such as through medications and lifestyle interventions.¹⁰⁴ The reasons for this will be provided in a later section. At this point, it is only important to note that long-term complications are common in diabetes and the main reason is because of persistent hyperglycaemia. Specifically, when hyperglycaemia remains at chronically high levels for longer than 2 years, causing long-term microvascular and macrovascular damage to the cardiovascular system, kidneys, nerves, and retinas (Figure 1.4). There is abundant evidence from a range of populations which show that duration of diabetes - or the length of time since diagnosis - is strongly correlated with complications of diabetes, and that this correlation with levels of HbA_{1c} is a measure of longer term glycemia.¹⁰⁵⁻¹⁰⁷ The next section will describe these complications and their severity in more detail.

Figure 1.4 Summary of the short and long-term complications of diabetes and how they relate to diabetes awareness and long-term management.



Figure 1.4 presents a flow-chart of how sustained glycaemia relates to prevalence of diabetes complications. An estimated 46% of all cases of diabetes worldwide affect people who are unaware of their condition. In these cases, treatment is unlikely. Without effective intervention, chronic hyperglycaemia, which is inherently part of diabetes, leads to serious, life-threatening diabetes complications. Of those who are aware, the effectiveness of their interventions is correlated with less severe complications. Those who are being treated with antidiabetic medication or glucose-lowering drugs in general should be aware of the risks of hypoglycaemia. *Euglycaemic target refers to HbA_{1c}, which is a biomarker for long-term glycaemia described in a later section. Image adapted from information presented in previous sections.

1.5.2.1 Macrovascular

Macrovascular complications affect arterioles, capillaries, and venules, which play an important role in supplying blood to organs. As such, the main macrovascular complications of diabetes are cardiovascular diseases, and include stroke, peripheral vascular disease, and coronary heart disease.⁷⁴ There is some debate over whether microvascular complications precede macrovascular complications in terms of pathogenesis and development, or whether they are distinct entities which progress separately.¹⁰⁸ Al-Wakeel and colleagues (2009) and Krentz and colleagues (2007) concluded that both types of complications develop simultaneously.^{109, 110} It is also possible that macrovascular and microvascular complications develop simultaneously and independently, though more research is needed to clarify the interaction between risk-factors of diabetes and it's complications.

1.5.2.2 Microvascular

End-stage renal disease, retinopathy, and neuropathy are the common microvascular complications of diabetes.¹¹¹. An estimated 80% of all end-stage renal disease cases are caused by diabetes or hypertension, and of those related to diabetes, 91% were due to T2D.⁷⁴ Figure 1.5 summarises the general process by which hyperglycaemia leads to tissue damage in certain organs. Pettus and colleagues (2019) determined that suboptimal glycaemia in T1D adults (n=31,430) was associated with higher prevalence of acute complications, such as severe hypoglycaemia and diabetic ketoacidosis, as well as chronic complications, such as neuropathy and nephropathy.¹¹² Brownlee (2005) has suggested that certain cells in the renal nephrons, eyes, and neurons of the autonomic nervous system cannot regulate their internal glucose concentrations efficiently, and thus are susceptible to damage from hyperglycaemia.¹¹³ Consequently, the progression of diabetes is associated with a decline in autonomic modulation of the heart and cardiovascular disease.¹¹⁴⁻¹¹⁶ Clinical research has shown that autonomic function is significantly diminished in diabetes compared to people without diabetes.¹¹⁷ There is widespread acceptance that chronic hyperglycaemia causes irreparable damage to nerves of the autonomic nervous system, and diabetes is the most common cause of neuropathy.¹¹⁸





Figure 1.5 outlines the process of hyperglycaemia-induced tissue damage. There is a strong genetic component to diabetes, which contributes to hyperglycaemia as well as cellular damage. Hyperglycaemia is associated with changes in cellular metabolism and structure. *Other accelerating factors include hypertension and hyperlipidaemia, which act independently to exacerbate hyperglycaemia and accelerate tissue damage. Adapted from Brownlee (2005).¹¹³

1.5.2.3 Diabetic Autonomic Neuropathy

As with most adverse complications of diabetes, diabetic autonomic neuropathy is more common with chronic hyperglycaemia, and there is a strong agreement in the literature that duration of diabetes is correlated with increased risk for diabetic autonomic neuropathy.^{107, 119} This reinforces the current understanding that untreated hyperglycaemia accelerates tissue damage in diabetes. Degeneration of the autonomic nervous system is caused by the accumulation of oxidising agents which compromise vascular supply to the nerves.¹²⁰ Consequently, diabetic autonomic neuropathy is associated with the development of diabetic foot disease, ulcers, and vascular disease,^{121, 122} and longitudinal research published by Morbach and colleagues (2012) reports that the prognosis of subjects with foot ulcers is especially poor.¹²³ In their literature review, Vinik, Erbas, and Casellini consider diabetic autonomic neuropathy to be one of the least understood complications of diabetes.¹¹⁵ One of the problems that has been identified in the treatment of diabetic autonomic neuropathy is the lack of simple methods for evaluating autonomic function.124 Based on the MONA LISA Hypothesis (Most Obesities k/Nown Are Low In Sympathetic Activity), the measurement of sympathetic activity – a component of the autonomic nervous system that can be partially measured by low frequency heart rate variability – may be useful in monitoring the progression of complications.¹²⁵ Lagani and colleagues (2018) advocate for the development of prognostic tools, such as models which include clinical data on age, sex, smoking habits, blood pressure and HbA_{1c}, as these are strongly correlated with diabetes complications.¹²⁶ Hyperglycaemia is a leading cause of diabetic autonomic neuropathy, and people with diabetes who are affected by diabetic autonomic neuropathy have increased mortality.127, 128

1.5.3 Neurophysiology of Blood Glucose Level and Autonomic Activity

The autonomic nervous system modulates glucose metabolism and BGL through its two counter regulatory branches.¹²⁹ The sympathetic nervous system acts to increase BGL through direct innervation of the liver and skeletal muscles to increase glucose production, as well as through activating the adrenal medulla and pancreas to produce epinephrine and glucagon, respectively.¹²⁹ During exercise or exposure to cold, the sympathetic nervous system may also be involved in increasing glucose uptake by skeletal muscles, despite unchanged BGL.¹³⁰ This is likely related to survival mechanisms, rather than conventional glucose homeostasis. The autonomic nervous system can also directly innervate the liver in response to pancreatic glucagon or insulin release, and the effects of direct sympathetic neural innervation of the liver are stronger compared to parasympathetic innervation.¹³¹ The parasympathetic nervous system effectively decreases BGL by increasing insulin release from the pancreas, as well as by directly innervating the liver.¹³² Evidently, autonomic innervation of endocrine organs is an important component of glucose regulation. Activation of the vagus nerve, the main parasympathetic nerve, leads to increased gastric acid secretion to facilitate digestion, and also leads to decreased heart rate, a staple of parasympathetic control.¹²⁹ Insulin and glucagon, as well as epinephrine and norepinephrine, cause neural activation of the liver. Epinephrine is released by the adrenal medulla and the sympathetic nerve terminals, and rapidly increases BGL by promoting liver glucose production and inhibiting insulin-mediated glucose uptake in skeletal muscles. By comparison, norepinephrine contributes little to liver glucose production, but is involved in glucose uptake and use in skeletal muscles independently of insulin.¹³²

Over the past 20 years, there has been a shift in the fundamental ways in which diabetes affects quality of life. Mortality rates for people with diabetes are declining, and there have been decreases in the incidence of classic complications such as cardiovascular diseases and lower limb amputation.¹³³ Current understandings of the underlying mechanisms of diabetes complications are insufficient, and future research should consider how this affects the development of better therapeutic options and treatment outcomes.¹³⁴ Based on the information provided in this section, improving access to technologies that can evaluate autonomic function in everyday clinical practice may be invaluable in the early detection, and thus early intervention, of diabetic tissue damage, as the peripheral nervous system is an early target of hyperglycaemic damage.¹³⁵ The role of the autonomic nervous system in glucose regulation is an important component of this PhD candidature, and will be discussed in a later section. Now that the nature and severity of diabetes complications have been established, it is clear these represent a significant burden to both individuals and national health systems.

1.6 Importance of Glycaemic Management

There is a strong consensus in the literature that stringent management of BGL, which involves maintaining BGL within the target glycaemic range, significantly reduces the risk and severity of diabetes complications.¹³⁶⁻¹³⁸ This has been reinforced by the ADVANCE trial¹³⁹ (n=11,140) and the follow-up study of the Veterans Affairs Diabetes Trial¹⁴⁰ (n=1,791), both of which demonstrated a reduction in microvascular and macrovascular complications of diabetes in groups randomly assigned to intensive glucose management compared to less intensive. Large-scale longitudinal studies have reinforced this understanding as well. In the United Kingdom Prospective Diabetes Study (n= 4,209), T2D subjects undergoing intensive insulin therapy to manage their BGL over 10 years showed a 24% reduction in microvascular disease, a 15% reduction in myocardial infarction, and a 13% reduction in all-cause mortality compared to the group treated with a simple dietary restriction.¹⁴¹ The results from the Diabetes Control and Complications Trial and its long-term observational follow-up, the Epidemiology of Diabetes Interventions and Complications study (n=1,441), also demonstrated that stringent glycaemic management, facilitated by intensive insulin therapy and vigilant glucose monitoring, substantially reduce the long-term complications of diabetes in people living with T1D.¹⁴²

1.6.1 Glycaemic Variability

There is growing recognition that glycaemic variability, attributed to large, acute fluctuations in BGL, such as those associated with glucose spikes after a meal, is also relevant in the pathogenesis of autonomic complications. Glycaemic variability indicates the presence of glycaemic excursions,¹⁴³ and BGL falling outside the optimal limits leads to complications in the short and long term, as discussed. As a therapeutic target, it is under-recognized, and recent literature reviews advocate for the importance of targeting glycaemic variability as a means of reducing the risk of complications of diabetes.¹⁴⁴ In the Verona Diabetes Study, higher variability in long term fasting BGL was shown to be associated with high total mortality, as well as higher mortality from cardiovascular disease and cancer.¹⁴⁵ Greater glycaemic variability is associated with lower autonomic function in T2D, and stringent glucose monitoring and intervention is crucial in reducing glycaemic variaiblity.¹⁴⁶ Nusca and colleagues (2018) contend that the two major aspects of glycaemic variability are as follows: the magnitude of glucose fluctuations, including the height from the nadir to the peak; and the length of time between glucose fluctuations. Both are important in glycaemic variability, which is associated with coronary heart disease and other vascular complications.¹⁴⁷ Damage from glycaemic variability occurs at a cellular level, and there are several pathways in which this is theorized to occur.

1.6.2 Oxidative Stress and Hyperglycaemia

A popular long-held theory is that complications of diabetes, particularly neuropathy, are the result of an increase in oxidative stress.¹⁴⁸ To elaborate, there are four key processes which link hyperglycaemia and neuropathy in diabetes, and they all involve excess production of reactive oxygen species (ROS) by the electron transport chain in the mitochondria. The electron transport chain is a major source of ROS, including superoxide, hydroxyl radical, and hydrogen peroxide, and these arise from oxidative phosphorylation, glucose autoxidation, and multiple enzymatic processes, which are all heightened in hyperglycaemia.¹⁴⁹ This is because high BGL leads to glucose molecules accumulating in the cells of the muscles, adipose, and pancreas, where these cellular processes which produce ROS are increased. ROS are likely helpful in cell signaling, acting as second messengers in intracellular signaling pathways. However, in excess, these cause damage to the membranes, protein structure, lipids, and DNA of cells, and thus contribute to oxidative stress.¹⁴⁹ As such, chronic hyperglycaemia leads to degeneration of autonomic nerves by way of affecting the biochemistry of free radicals.

Oxidative stress causes harm to many types of cells, however, there are certain types of cells which show signs of damage sooner than others in response to chronic hyperglycaemic conditions. Clinical studies have shown that parasympathetic neuropathy can have a relatively early onset in T2D.¹⁵⁰ It has been proposed that increased oxidative stress, caused by hyperglycaemia, affects the parasympathetic nerves first because their myelinated preganglionic fibres are longer and thicker compared to sympathetic nerves.¹⁵¹ This is related to the work of Ewing and Clarke (1985) who proposed a battery of five non-invasive reflex tests for early detection of autonomic neuropathy in diabetes. Since parasympathetic neuropathy occurs early in T2D, Ewing and Clarke (1985) proposed the 'Ewing battery' which includes the Valsalva manoeuvre, lying to standing heart rate response, deep breathing heart rate response, postural blood pressure change, and sustained handgrip test.¹⁵² Worsening of autonomic function can be detected using the Ewing battery, a gold-standard in clinical practice, and early detection is important in diabetes as autonomic neuropathy is linked to myocardial ischemia, coronary artery disease, and stroke.¹⁵³

Autonomic neuropathy in diabetes also presents low-grade chronic inflammation. Cytokines such as interleukin-1, 6, and 8, as well as C-reactive protein, TNF- α , and monocyte chemoattractant protein-1, contribute to inflammatory reactions.¹⁵⁴ These are produced by activated immune cells and resident macrophages and adipocytes, and play a key role in inflammatory signalling. Activation of receptor-mediated inflammatory signalling leads to increased oxidative stress, which as discussed contributes to degeneration of autonomic nerves, particular those of the parasympathetic nervous system.¹⁵⁵ Several therapeutic drugs have

demonstrated reductions in microvascular complications in T2D by reducing inflammation. For example, melatonin reduces elevated levels of proinflammatory cytokines and increases blood circulation to the nerves.¹⁵⁶ Targeting inflammation signalling in chronic hyperglycaemia has important clinical implications for reducing the risk and severity of autonomic neuropathy. Given the background of diabetes complications and the cellular processes involved, it is clear that stringent glycaemic management is a priority in reducing the current cost of diabetes on individuals and society. Now that it is clear why glycaemic management is important, the next sections will explain how this is possible and discuss current limitations.

1.6.3 Treatment

Adopting the best course of treatment of diabetes is complicated by a range of factors. Primarily, therapies will differ based on the type of diabetes, as pathological differences between T1D and T2D lead to profound differences in treatment.⁴⁶ The severity of the diabetes as well as other comorbidities are also key factors to consider. Although therapeutic strategies differ significantly, T1D and T2D are both characterised by high levels of blood glucose that cannot be controlled by the body's own insulin. Left untreated, chronic hyperglycaemia leads to debilitating, life-threatening complications in the long-term. As such, it is possible to generalize the main objective of diabetes treatment as such: maintain target glycaemia by use of antidiabetic mediation in combination with lifestyle modification.¹⁵⁷ The extent of diet and exercise changes, as well as the nature of the pharmacotherapy, can differ significantly among people with diabetes.

Most cells in the human body require insulin signalling to take in glucose from the blood. In conditions defined by insulin dysfunction, such as diabetes, the lack of signalling causes glucose to reach high levels in the blood.¹⁵⁸ Therefore, pharmaceutical insulin therapy is a common method for supplying exogenous insulin to people living with diabetes who cannot use or produce their own. This usually applies to those living with T1D, as insulin production is limited, but it may also apply to end-stage T2D where insulin resistance has progressively worsened, and pancreatic insulin supply may also be limited. According to Riddle and colleagues in their 2021 consensus report, pharmaceutical therapies targeting glycaemic management in diabetes have improved greatly in recent years.¹⁵⁹ This section aims to describe the role of medications in dealing with dysglycaemia (Figure 1.6), as well as explain why insulin therapy is important in diabetes management.





Figure 1.6 illustrates the production, regulation, and role of insulin in maintaining glucose homeostasis. Following the absorption of glucose-containing nutrients in the gastrointestinal tract, glucose sensors in the liver detect an increase in blood glucose concentration. In a person without diabetes, this is followed by the release of insulin from the pancreas to promote glucose uptake by somatic cells, such as those in skeletal muscle, and thus reduce blood glucose to an optimal level. In diabetes, reduced insulin activity causes glucose levels to rise, without a mechanism for maintaining homeostasis. Adapted from Grayson, Seeley & Sandoval (2013).⁴⁸

1.6.3.1 Type 1 Diabetes Treatment

As T1D results in a total or near-total deficit of insulin, the condition is managed with insulin therapy paired with frequent monitoring of BGL to remain vigilant of hypoglycaemia.¹⁶⁰ Insulin therapy can be achieved with 3-4 injections of basal (slow-acting) and prandial (fast-acting) insulin per day, or continuous insulin infusion.¹⁶¹ Automated insulin delivery, combined with currently available CGM, significantly improves the length of time people with T1D can have target glycaemia. With T1D, complete remission is possible through pancreas transplant, though novel research is also exploring the potential for gene therapy to restore insulin production in T1D and result in remission.¹⁶² Regardless of the method, the main objective of T1D treatment is clear: maintain BGL in the optimal range for as long as possible and reduce glycaemic variability. Among common diabetes treatment strategies, more stringent methods are correlated with improved glycaemic management; however, based on a meta-analysis of 37 articles, there is little data on whether engagement with treatment improves quality of life in diabetes.¹⁶³ Future research may need to consider how current treatment methods do not necessarily lead to better quality of life.

In T1D, the effects of a pancreas transplant on long-term complications is an area of research that is lacking, though available literature suggests that a transplant is effective in restoring euglycaemia in T1D and improving quality of life.¹⁶⁴ However, the availability of donors and the lifelong immunosuppression therapy are major factors which limit the viability of transplant surgery for most people living with T1D.¹⁶⁵ Surgery in general is invasive and is generally not applicable in most cases. There is growing interest in islet cell transplant and stem cell generation of islet cells prior to gene therapy, though currently this is not used in clinical settings. Transplanting islet cells has shown improvements in insulin independence up to five years following the transplant, with minimal complications.¹⁶⁶ 20-year follow-up data is available for a cohort of n=29 females and n=20 males.¹⁶⁷ At the end of the follow-up, from the year 2000 to 2020, 86% of the subjects had no albuminuria, 12% had microalbuminuria, and 2% had macroalbuminuria. Two subjects (4.08%) died during the follow-up period. Since pancreas transplantation has been in clinical use for longer, there is 30-year follow-up data for subjects. In a cohort of n=2,796 subjects with T1D, simultaneous pancreas-kidney transplant improved survival and resulted in an almost twofold lower 10-year mortality rate.¹⁶⁸ Data from the International Pancreas Transplant Registry of n=18,159 pancreas transplants demonstrated that simultaneous pancreas-kidney transplanting is superior to pancreas transplants alone in terms of survival. At 5-year, 10-year, and 20-year follow-up, survival rate was 80%, 68%, and 45%, respectively, for simultaneous pancreas-kidney transplants versus 59%, 39% and 12%, respectively, for pancreas transplants alone.¹⁶⁹

Transplants restore euglycemia in people with T1D, and in turn these reduce the progression of diabetes complications, such as retinopathy. However, large epidemiological data has shown that early metabolic control is critical in the overall development of complications of diabetes, despite the short-term benefits provided by intensive glycaemic control. This is hypothesised to be due to 'metabolic memory', a phenomenon where cellular pathways, such as excessive cellular ROS at the level of glycated-mitochondrial proteins, maintain stress signalling and contribute to long-term diabetes complications.¹⁷⁰ Metabolic memory is one issue which has been used to justify the importance of early intervention and prevention of diabetes, as early control is critical in preventing the progression of complications.

1.6.3.2 Type 2 Diabetes Treatment

Conversely, management of T2D typically involves improving physical activity and diet in combination with pharmacotherapy.⁶⁷ Lifestyle changes include: daily physical activity;¹⁷¹ dietary changes and restrictions,¹⁷² such as minimising processed foods and high energy density foods;¹⁷³ and reducing alcohol consumption and smoking.¹⁷⁴ The objective of these lifestyle interventions is to improve glycaemic management in the short term and restore insulin sensitivity in the long term, which in theory can result in remission from T2D.¹⁷⁵ Such treatment is considerably less invasive and more affordable compared to surgery, though it is important to note that sustained remission of T2D is very rare, even with intensive lifestyle and pharmaceutical intervention,¹⁷⁶ though it is more common with surgery.¹⁷⁷ Remission is described as achieving target glycaemia at least three months after ceasing antidiabetic medication, as the person's natural homeostatic processes have been restored and optimal glycaemia can be achieved without pharmacotherapy.¹⁵⁹

Lifestyle interventions are often difficult to maintain, and weight that is initially lost can be regained over time as people with T2D age. As such, pharmacotherapy is considered a first-line therapy, and some drugs are even used in the prevention of T2D. A common therapeutic agent in the treatment of prediabetes and T2D is metformin, an oral glucose-lowering drug which helps to draw glucose out of the bloods by enhancing insulin action in the liver and skeletal muscles.¹⁷⁸ Metformin use is associated with reduced oxidative stress in T2D subjects compared with subjects treated with lifestyle interventions¹⁷⁹ or with sulfonylureas.¹⁸⁰ Albumin in the blood is an important antioxidant, and it has been proposed that an added benefit of metformin use is the protection it provides to serum albumin. By this process, people with T2D treated with metformin show reduced oxidative stress and glycation.¹⁸⁰ Metformin use also provides sustained cardiovascular benefits, which are unlikely related to its glucose-lowering effects
since this sustained benefit has not been observed with sulfonylureas or insulin therapy.¹⁸¹ Clinical research has suggested that metformin may improve sympathovagal balance in T2D.¹⁸²

For most cases of T2D, maintaining euglycaemia – or optimal levels of BGL – involves medication and lifestyle interventions. However, there is data which supports more non-conventional options. According to a meta-analysis of 10 studies (n=17,532) conducted by Billeter and colleagues (2018), metabolic surgery such as bariatric surgery prevents microvascular complications to a better extent in T2D than treatment by medication, which includes insulin therapy and glucose-lowering medications.¹⁸³ Chang and colleagues (2014) confirm in their meta-analysis and review (n=161,756) that bariatric surgery works to slow or even reverse the progression of T2D by restricting hunger, and thus also caloric intake.¹⁸⁴ However, as with any surgery the invasive nature of bariatric surgery provides a risk for serious complications, limiting the application of this treatment.

Insulin therapy is required when non-insulin therapy, such as metformin or other glucoselowering drugs, is unable to maintain glucose homeostasis and euglycaemia.¹⁸⁵ In T2D, multiple drugs may be used to achieve euglycemia, and insulin can be added on to multiple oral therapies, and in T1D treatment is typically just insulin therapy. In a person without diabetes, homeostasis is the process by which BGL is maintained within a narrow range (Figure 1.7), between 4.0 and 7.7 mmol/L.¹⁸⁶ As with other homeostatic systems, such as core body temperature, BGL must be maintained within its optimal range to ensure an individual's health.¹⁸⁷ Intensive insulin therapy reduces the risk of complications in the long-term, but researchers warn that the risks of severe hypoglycaemia in the short-term may outweigh this.¹⁸⁸ In insulin-dependent diabetes, it is important to consider diet and medication guidelines set by the physician to reduce the incidence of hypoglycaemia, as glucose monitoring systems cannot always be relied on for the detection of hypoglycaemia. The reasons for this unreliability will be explored in depth in a later section which discusses the applications of glucose monitoring.



Figure 1.7 Hormonal regulation of blood glucose level in people without diabetes.

Figure 1.7 portrays the role of glucagon and insulin in maintaining glucose levels within a optimal range. In a fasting state, where nutritional sources of glucose are restricted for at least eight hours, homeostatic processes seek to maintain an individual's BGL above 4.0 mmol/L through the breakdown of stored glycogen in the muscles and liver into glucose. BGL is maintained below 7.8 mmol/L during any metabolic state by releasing insulin, which increases the uptake of glucose by somatic cells. **BGL** = Blood glucose level. **mmol/L** = Millimoles per litre. Adapted from Roder and colleagues (2016).¹⁸⁹

To summarise, a dysfunction in the role of insulin reduces the ability of people with diabetes to optimise their BGL when it rises above 7.7 mmol/L. Treatment of diabetes revolves heavily around an individual's metabolic state, such as whether they are fasting, eating, exercising, or at rest, to name a few. Some examples of this have been discussed already, such as the need to take antidiabetic medications before consuming a meal to reduce the inevitable postprandial glucose spike. This requirement also aims to reinforce a routine with those taking antidiabetic medication, as research from Vervloet and colleagues (2013) demonstrated that disruptions to a regular routine can be detrimental to overall treatment efficacy.¹⁹⁰ As diabetes management is heavily focused on meeting glycaemic targets and reducing glycaemic variability, metabolic state is an important factor when deciding when to take medications and when to self-measure BGL. To elaborate, measuring BGL in the morning before breakfast is an example of 'fasting

BGL', as the absence of food or medications in the last eight hours results in a glucose baseline.¹⁸⁹ The importance of fasting and postprandial BGL are highlighted in the literature as they are key indicators of health. For example, fasting BGL is useful for diagnosing diabetes and monitoring its progression, and postprandial BGL is predictive of cardiovascular complications in diabetes.¹⁹¹⁻¹⁹³ The ability to measure BGL in different metabolic states may to relevant to the development of future measures of BGL, and thus is relevant to this thesis. This will also be relevant to the methodology chapters, where it will be discussed how BGL can be measured in fasting and postprandial states.

1.6.4 Homeostasis

Consider that with any physiological system, there is significant minute-to-minute variation due to ongoing homeostatic processes. In the case of glucose homeostasis, BGL fluctuates in response to metabolic demands, such as the change between exercising and being at rest, as well as in response to the digestion of energy sources after consuming a meal.¹⁹⁴ These are just two of the main pressures which cause BGL to fluctuate across the day. In the absence of these a state which is referred to as 'fasting' – BGL remains very stable in a person without diabetes: between 4.0 and 5.5 mmol/L¹⁸⁹ Following the consumption of a meal, an individual would expect their BGL to fluctuate between 5.6 and 7.7 mmol/L. Therefore, individuals who present with a fasting BGL greater than 5.5 mmol/L or a postprandial BGL greater than 7.7 mmol/L are classified as having impaired glycaemia and may have diabetes or pre-diabetes (refer to Table 1.1). Clinicians may also order a 'random' blood glucose assessment, where BGL is assessed at an undefined point where a person is not necessarily fasting nor postprandial. The objective of this is the determine whether a person's BGL is above 11.1 mmol/L, because BGL above this threshold at any point is indicative of diabetes and warrants further investigation with a glucose tolerance test. Current literature indicates that random BGL is difficult to interpret, as it may be affected by confounders such as recent food intake and medications,¹⁹⁴ and so for the purposes of this thesis the focus will be on BGL assessed in fasting or postprandial states.

1.6.5 Metabolic State

Though BGL is most often assessed in a fasting state in a clinical setting, there is a strong consensus in the literature that postprandial BGL is a better predictor of diabetes complications. Cavalot and colleagues (2006) contend that postprandial BGL, but not fasting BGL, predicts cardiovascular events in T2D, with stronger prediction in women compared to men.¹⁹⁵ Some

research suggests that fasting BGL is independently associated with cardiovascular disease risk, however this only holds true for groups in the 'high-optimal' range of BGL, between 5.3 and 5.6 mmol/L.¹⁹³ A more recent study conducted by Jingjing and colleagues (2017) found that postprandial BGL performed better in screening for coronary heart disease compared to fasting BGL and HbA_{1c}.¹⁹² Clinical research has also shown that specifically targeting and lowering postprandial BGL in T2D, compared to fasting BGL, leads to better glycaemic management overall and better HbA_{1c}.¹⁹¹ Clearly, postprandial BGL offers key insights into diabetes prognosis, and a convenient method of measuring postprandial BGL may provide meaningful information to people currently living with diabetes. To reiterate, the specifics of how fasting and postprandial BGL are measured will be described in the methodology chapters. As previous sections have explored in detail how diabetes can be treated by current standards, the next section will explore how diabetes – and blood glucose – can be monitored, as this is critical to making clinical decisions and improving treatment efficacy.

1.7 Invasive Glucose Monitoring: The Current Standard

Stringent monitoring of blood glucose is strongly advocated for people living with diabetes for several reasons. The first reason has been discussed throughout previous sections: stringent BGL monitoring is a requirement for the continued health and wellbeing of certain people living with diabetes (Section 1.5). This is most relevant for those with T1D or end-stage T2D on insulin therapy, as the reliance on external insulin is associated with increased risk for severe hypoglycaemic events. In these groups of people, glycaemic excursions occur often in the high range of BGL (hyperglycaemia) and in the low range (hypoglycaemia), and current guidelines recommend BGL be assessed 4-10 times per day.¹⁹⁶ Vigilant monitoring of BGL is therefore critical for detecting these glycaemic excursions early and intervening.¹⁹⁷ The literature advocates strongly that glucose monitoring be a priority even for those with early T2D and other non-severe forms of diabetes.¹⁹⁸ It is important that individuals take responsibility for monitoring their glycaemia, as diabetes is a chronic condition that affects everyone differently. Current literature advocates for frequent monitoring of blood glucose and proactive intervention to achieve target glycaemia and improve quality of life in both the short-term and long-term.¹³⁶⁻¹³⁸

Technologies used to monitor glucose levels can be categorised by the frequency in which they assess glucose levels. Intermittent methods are characterised by their ability to only provide 'snapshot' data, such as a blood glucose assessment which directly measures BGL at a specific point in time, and continuous methods offer constant monitoring of BGL, which may be useful

for vigilant monitoring of potential hypoglycaemia.¹⁹⁹ Both technologies fill different roles and have their own uses in T1D and T2D, and each have different advantages and disadvantages. As a rule, there is a trade-off between accuracy and frequency when it comes to current methods of monitoring of BGL. For example, technologies that are more accurate are impractical to use frequently,²⁰⁰ and technologies that measure glucose levels frequently, such as continuous systems, are less accurate, as they measure interstitial glucose instead of blood glucose.²⁰¹ However, continuous monitoring systems have the advantage of being more useful for detecting hypoglycaemic events, as they may occur unexpectantly and require constant monitoring.

Intermittent measures of BGL are praised in the literature for their accuracy but criticised for their invasive nature.²⁰² They are defined as 'intermittent' because they provide only a snapshot measure of BGL and cannot provide information on BGL over long periods of time. Though it is possible to take many intermittent measurements in quick succession, allowing for changes in BGL data to be tracked similarly to continuous measures, this is impractical due to the invasive nature of these measures. This will become clearer in the following sections, as the nature and role of these intermittent technologies is discussed. There are a range of devices capable of measuring or estimating BGL. The following sections will focus on those which are most used, and which have the most significant clinical significance at present.

1.7.1 Blood Glucose Assessment

A core feature of many chronic conditions that do not require constant hospitalisation or care is that there is a heavy reliance on the individual to self-monitor their condition. In the case of T1D and T2D, this involves self-administering prescribed medication, adopting lifestyle recommendations set by their primary care physician, and self-monitoring of blood glucose (SMBG). However, the ability of a person to monitor their own BGL is based on the strengths and limitations of the blood glucose monitoring tool, as well as how well they understand the tool itself. Therefore, tools that are more convenient or easier to can be more effective compared with tools which require more education for effective use.

According to WHO guidelines, SMBG is synonymous with the blood glucose assessment – the most commonly-used intermittent measure of BGL.²⁰³ SMBG describes the routine of performing blood glucose assessments at specific intervals in one's own home, such as before or after meals, and using the information to adjust their medication or lifestyle in conjunction with advice from their primary care physician. The blood glucose assessment requires a glucose meter, or glucometer, to assess a small sample of blood typically drawn from the finger. As such, it is often referred to as the 'fingerspot' check. Capillary BGL measured at the fingertip

correlates strongly with BGL measured in the arteries.¹⁹⁹ BGL is recorded by the glucometer in millimoles per litre, and the optimal ranges are dependent on the metabolic state of the individual when they perform the blood glucose assessment. For example, fasting BGL ranges between 4.0 and 5.5 mmol/L in people without diabetes, and is above 7.0 mmol/L in people with diabetes before taking medication.⁴⁷

In n=30 non-insulin-treated T2D subjects, SMBG led to significant improvements in HbA_{1c} compared to the control group, and was an effective tool for determining antidiabetic drug dosage.²⁰⁴ Importantly, the authors of this study confirmed that detailed training was not required for subjects with suboptimal glycaemic to interpret the results of SMBG and adjust their drug treatment. Per the 2011 Consensus Report from the Coalition for Clinical Research – SMBG Scientific Board, SMBG is most effective when subjects can use it as a tool to guide their own treatment, and both patients and health care providers should be educated on how to interpret and respond to information provided from SMBG.²⁰⁵ Amongst the board members, there was also a consensus that the non-insulin-treated T2D trials support the efficacy of SMBG, and that further studies are required to assess the cost-benefits of SMBG across different end points, for example not limited to HbA_{1c}.

1.7.1.1 Strengths

The blood glucose assessment may be performed in a clinical setting by physicians as part of a routine check-up, similar to blood pressure monitoring, though it is more commonly used in SMBG as the primary home-monitoring tool.²⁰² Studies from Halldorsdottir and colleagues (2013) and Tack and colleagues (2012) indicate that glucometers are performing accurately in the context of diabetes management,^{206, 207} and there are a wide range of affordable and cost-effective glucometers convenient for at-home SMBG.²⁰⁸ Though there are minor variations in SMBG approaches in the literature, SMBG almost always involves a glucometer and blood glucose assessment, and there are no significant clinical differences between the different SMBG approaches in T2D (n=453) according to a randomised trial conducted by Farmer and colleagues (2007).²⁰⁹

For glucose monitoring, the complexity of the diabetes is related to how often BGL needs to be monitored. For example, in cases of insulin-dependent diabetes, which includes most cases of T1D as well as cases of end-stage T2D, the individual's health and wellbeing depends on their ability to supply themselves with insulin in the form of medication.⁴⁷ Dieting and exercise are insufficient as the primary means of lowering BGL to optimal levels. Allemann and colleagues (2009) advocate for stringent BGL monitoring in insulin-dependent diabetes, including blood

glucose assessments before and after meals, as well as continuous monitoring of BGL.²¹⁰ These tools empower insulin-dependent individuals with the means to self-monitor spikes in BGL after meals, as well as respond to hypoglycaemic events instigated by their medication to prevent hospitalisation. Nagelkerk, Reick, and Meengs (2006) claim that empowering people living with diabetes with the knowledge of specific diet strategies and plan of care are important for improving SMBG and overall glycaemia.²¹¹ This may be related to an individual's tendency to "take what I think works for me", as it has been observed that a person's understanding of their diabetes treatment, including their knowledge of the benefits and risks, is positively correlated with how closely they follow that treatment.²¹² Improving the accessibility and ease of use of current SMBG technologies may therefore also improve people's knowledge of their condition, including diabetes progression, glycaemic management, and treatment effectiveness.

1.7.1.2 Limitations

Though the blood glucose assessment is an accurate estimate of BGL, glucose concentration is only recorded at the one specific point in time.^{28, 29} As a consequence, this recording can be significantly affected by recent food ingestion or recent exercise, as well as acute illness.²¹³ It is important to control for these confounding variables when assessing BGL using the blood glucose assessment, because the glucometer will provide only a single measurement, and it is impractical to perform additional assessments in a short period of time.²¹⁴ A common method of controlling for these confounders is for subjects to refrain from any food or drink, except water, in the eight hours leading up to their appointment. This allows for the blood glucose assessment to record a fasting BGL. Clinical research has shown that SMBG is an effective tool on its own and can improve glycaemia in diabetes without intensive drug treatment.²¹⁵ Specifically, SMBG leads to better glycaemic management in the short term, but over time that management shows diminishing returns. To elaborate, a systematic review and meta-analysis found significant reductions in long-term glycaemia in people with T2D after 12 weeks and after 24 weeks of SMBG, but changes in long-term glycaemia were non-significant after 52 weeks.²¹⁶

There is considerable variability in the guidelines of different international diabetes organizations regarding optimal SMBG and what it involves.²¹⁷ The lack of consistent guidelines may lead to confusion about the most optimal frequency and timing of SMBG. Additionally, some studies indicate that the cost-benefit analysis of SMBG is unfavourable. The results from a parallel group randomised controlled trial (n=453) suggest that SMBG after 12 months leads to increased healthcare costs and is unlikely to be cost-effective when used routinely.²¹⁸ This is ratified by a systematic review of the literature by Clar and colleagues (2010) which found that

SMBG is not cost-effective and only offers limited improvements in glycaemic influence in noninsulin dependent diabetes, including T2D treated with oral antidiabetic medication or diet alone.²¹⁹ However, a meta-analysis of seven randomised controlled trials found a significant decrease in overall glycaemia in response to SMBG, providing large-scale results which suggest that people who utilise SMBG have better glycaemia compared to those who do not.²²⁰

Clearly, there is value to SMBG. It is suggested that SMBG may lead to more cost-effective glycaemic management when it is cheaper, more convenient, and when it is supplemented with healthcare education.²¹⁹ Cost effectiveness is generally evaluated in the context of an individual's current glycemia. SMBG is important for those with highly-fluctuating glycaemia, such as people with T2D who take oral medications which cause hypoglycaemia, but has limited benefit when glycaemia does not fluctuate significantly or where HbA1c is <7.5%.²²¹ Where insulin treatment is involved, and an individual is more reliant on exogenous insulin, SMBG is cost effective and essential to managing the dosing of insulin.

1.7.2 Continuous Glucose Monitoring

Continuous glucose monitoring (CGM) describes any technology capable of assessing BGL in a constant manner without pause. This allows for changes in BGL to be tracked in real-time, as data is continuously transmitted from a sensor, which is directly inserted into the tissue of interest, to a receiver which can display data as a glucose level.¹⁹⁹ The major advantage of CGM is that life-threatening increases or decreases in BGL can be detected as long as the device is being worn, which can be all hours of the day if necessary. CGM systems with better alarm features, such as the Dexcom G6, are associated with superior avoidance and intervention of glycaemic events.²²² As discussed, CGM is strongly recommended in insulin-dependent diabetes, as the reliance on externally-delivered insulin increases the individual's risk of hypoglycaemia.²⁸ CGM systems measure glucose levels in real-time and can identify severely low blood glucose, which can occur unexpectantly or even when asleep, which is known as nocturnal hypoglycaemia.²²³

CGM systems – devices which incorporate CGM technology – allow people to monitor their BGL vigilantly for glycaemic events and intervene in a timely manner. The use of a CGM system is associated with more optimal glycaemia.²²⁴ Most commercially-available CGM systems function by measuring interstitial glucose concentrations (Figure 1.8), which is the concentration of glucose surrounding cells.²²⁵ In a steady state, levels of glucose in the interstitial space are strongly correlated with capillary and arterial glucose levels, however they lag behind arterial concentrations only when BGL is changing rapidly, such as after the consumption of a meal.²²⁶

Unlike the blood glucose assessment, which directly measures glucose concentrations from a fingerspot blood sample which is strongly correlated with arterial BGL, a CGM sensor relies on minimally-invasive means to estimate arterial BGL. Based on data from Medtronic MiniMed CGM system, one of the most common systems in use, interstitial glucose concentration estimates can lag behind BGLs by 4-10 minutes.²²⁵





Figure 1.8 highlights the components involved in the continuous monitoring of interstitial glucose concentration, a close approximation of blood glucose concentration. The glucose sensor is inserted beneath the skin where it can contact the interstitial fluid. It then relays glucose data back to the continuous glucose monitoring system, which can process the data and report values digitally through an interface. Adapted from Medtronic MiniMed, Inc (2017).²²⁷

1.7.2.1 Strengths

A major strength of CGM is that is has been consistently shown to increase glycaemic management and quality of life in T1D²²⁸ and T2D²²⁹ with better outcomes seen in those who use CGM for at least 6 days per week.^{230, 231} This is not specific to insulin-dependent diabetes, as researchers also advocate the use of CGM in early T2D groups that have not progressed to insulin therapy.²³² At present, blood glucose is monitored in T2D on a more intermittent basis because the threat posed by hypoglycaemia is much lower²³³ and lifestyle-interventions take longer to become effective in achieving target levels of glycaemia.⁶⁷ Some justify the use of CGM in T2D because it is beneficial in reducing glycaemic variability.¹⁹⁸ Studies have consistently shown that glycaemic variability is associated with dangerous microvascular complications, and this variability occurs in early T2D as well as end-stage.²³⁴ Glucose variability it also associated with nocturnal hypoglycaemia episodes.²³⁵ CGM can detect large variations in blood glucose and

empower individuals with the information to intervene.²³⁶ This is not possible by measuring HbA_{1c} levels, as this only provides meaningful information about long-term trends in BGL, and the retrospective analysis does not allow for immediate intervention. Though convenient in some aspects, many SMBG tools used in T2D lack the ability to track daily glucose fluctuations, which may be more relevant in ideal management of diabetes.²³⁷

Many CGM systems are programmed to alert the wearer with an alarm when they detect glycaemic excursions. This allows for the wearer to intervene in a timely manner, and is especially useful in people with hypoglycaemia unawareness.²³⁸ Few alternatives to CGM exist for insulin-dependent diabetes, as hypoglycaemia onset can occur quickly, and can cause confusion, drowsiness, and can lead to coma.²³⁹ In T1D, CGM is often paired with SMBG, such as using blood glucose assessments to calibrate their CGM systems and to provide more accurate reference values when BGL is fluctuating rapidly after a meal. Some modern CGM systems in use, such as the Dexcom G5 and G6, are approved for non-adjunctive use, meaning they do not require calibration with routine blood glucose assessments.^{240, 241} This removes one of the core problems with CGM and increases its viability for use in both T1D and T2D.

To summarise, a review of recent studies has shown that there has been steady improvement in the reliability of CGM systems, in that devices require less calibration and more accurately detect glycaemic excursions.²⁴² Moreover, many recent studies are attempting to address specific concerns with CGM. For example, the PRECISE Study²⁴³ and the PRECISE II²⁴⁴ trial demonstrated the use of a CGM system that is intended to be used with sensors that last for 90 days, to reduce the discomfort and pain associated with weekly sensor replacements. To summarise, CGM is ideal in both T1D and T2D, in people with and without insulin-dependence. This is because hypoglycaemia, which is more prevalent in insulin dependent diabetes, is not the only clinically significant complication that can be reduced with CGM. Glycaemic variability, caused by sudden spikes in BGL, can also be reduced with CGM, and this is responsible for significant risk of complications in all diabetes cases.¹⁴⁴ Clearly, the ability to constantly track BGL changes in real-time is a major strength.²³⁸ However, for reasons that will be presented in the following section, modern CGM systems face many problems that make it difficult for people with diabetes and their primary care physicians to justify their use.

1.7.2.2 Limitations

CGM systems provide valuable data – irreplaceable in many circumstances – which forms the basis of many clinical decisions for people living with diabetes. However, concerns about the accuracy and reliability of CGM systems have been prominent since their introduction.²⁴⁵

Generally, performance of CGM systems is poor in the very low BGL range,^{246, 247} raising concerns that hypoglycaemia may not be detected in time or at all in some cases. Two widely-used CGM systems show low accuracy in the hypoglycaemic range.²⁴⁸ For these reasons, people with diabetes often cannot rely on readings from CGM systems alone when making treatment decisions, such as insulin dosage and timing, and may need to rely on glucometers (SMBG) because of their higher accuracy, even at very low ranges of glycaemia.^{245, 249} Though some specific CGM systems do not require calibration with a blood glucose assessment, many people with T1D continue to utilise SMBG, or the intermittent blood glucose assessment method, routinely because of its superior accuracy.²³⁸ For example, it can be used if a person is experiencing autonomic symptoms of early hypoglycaemia, such as sweating and heart palpitations, in circumstances where their CGM system has not yet detected the glycaemic excursion due to lag-time.¹³⁶ The concept of this 'lag-time' is illustrated in Figure 1.9. Severe hypoglycaemia represents a significant threat to those with insulin-dependent diabetes, and timely intervention of these glycaemic events is key to improving quality of life and preventing subsequent hypoglycaemic events.

Though there have been major improvements in the reliability and accuracy of CGM systems in recent years,^{250, 251} the core problems of affordability and convenience have yet to be solved by any current CGM system. The issue of affordability is prevalent across most commercial CGM systems. CGM systems have a high upfront cost and maintenance costs, and the question of costeffectiveness is difficult to answer when the benefit of using a CGM system is potentially lifesaving, depending on the severity of diabetes. Generally, though, the literature is in consensus that CGM is cost effective relative to SMBG and other health interventions.²⁵² Mean total lifetime costs of running a Dexcom G6, for example, are an estimated 18% higher compared to SMBG, and the mean quality-adjusted life years are 16% higher compared to SMBG, so there is an approximately equal value to the higher cost of using a CGM system.²⁵³ This cost analysis is largely dependent on the severity of the insulin-dependence, and differs greatly between individuals. The results of the DIAMOND randomized trial show that CGM is costeffective below a certain threshold, depending on the price an individual is willing or able to pay for their healthcare, with increased glycaemic management and reduced hypoglycaemic events.²⁵⁴ This is ratified by other studies which have concluded that CGM systems are cost effective in people living with T1D,²⁵⁵ and also people with T2D who do not use insulin,²⁵⁶ but people must be able to pay the initial out-of-pocket expenses to achieve the better quality of life associated with long-term CGM use. However, many individuals, particularly those in developing nations, may simply be unable to afford the high initial cost of the CGM system, as

well as the maintenance costs such as replacing sensors, even if it does provide superior quality of life.



Figure 1.9 Line graph comparing the glucose levels recorded by a continuous glucose monitoring system, a glucometer, and a laboratory glucose analyzer.

Figure 1.9 shows the readings observed by three different glucose monitoring technologies. The red line indicates data recorded by a CGM system, the black stars are data points measured from multiple blood glucose assessments (SMBG) performed in quick succession, and the blue circles represent data from a YSI 2300 START Plus laboratory glucose analyzer. The CGM system used in this example does not necessarily represent the accuracy of all CGM systems – it is simply an example showcasing a common problem with many devices. As seen, the readings from the CGM system (red line) lag behind the other two measures when glucose levels are changing rapidly, such as between 08:00 and 09:00am when the subject consumed a meal. The red line also lags behind the other two measures around the 12:00pm mark, when glucose levels drop into the low range. Adapted from Schrangl and colleagues (2018).²⁵⁷

1.7.3 Additional Measures

At present, there are a range of measures of blood glucose that are used in many different clinical settings. Two additional measures of glycaemia that have yet to be discussed in this section are the HbA_{1c} assessment and the glucose tolerance test. These do not fit neatly under

the definitions of 'intermittent' or 'continuous' as they are assessed in laboratory settings and provide unique measures of glycaemia. Compared to CGM and SMBG, these additional measures are less relevant to this thesis as they cannot provide information that can be used in real-time to make clinical decisions – they are outside the scope of this PhD candidature. However, they still provide meaningful information and are used routinely in diabetes diagnosis and monitoring. As such, they will be discussed only briefly to provide additional context for glucose monitoring.

1.7.3.1 Glucose Tolerance Test

The glucose tolerance test is the gold-standard for diagnosing diabetes, prediabetes, and gestational diabetes, as well as and impaired glucose tolerance.²⁵⁸ It involves a fasting BGL assessment, followed by oral ingestion of a standardized glucose load (75 gram glucose solution diluted in water).²⁵⁹ Blood glucose is assessed at standard times at 0, 1 hour, and 2 hours post-consumption. The results of the 2-hour oral glucose tolerance test may indicate normal, impaired, or diabetes levels of blood glucose.²⁵⁹ Additional measurements may be taken during the two hour period, for example BGL measured at the 30-minute and 60-minute mark can predict risk of diabetes and all-cause mortality.²⁶⁰ Some authors contend that BGL assessed at the 60-minute mark is a stronger predictor of lifetime risk of diabetes and diabetes complications compared to the 2-hour mark.²⁶¹ The glucose tolerance test is a strong clinical marker of diabetes and the gold-standard for diagnosis.

1.7.3.2 HbA_{1c} Assessment

One of the key outcomes used to assess diabetes progression is HbA_{1c}, or the percentage saturation of haemoglobin by glucose molecules carried in the blood.²⁶² This is a biomarker found in the blood which represents glycated haemoglobin levels and is routinely assessed in a laboratory setting from a person's blood sample.²⁶³ HbA_{1c} is often viewed as the long-term average of an individual's BGL over the past 120 days, which is the lifespan of the average red blood cell.²⁶⁴ As such, it can provide longitudinal averages of BGL, and this can be used to determine the effectiveness of the person's treatment and lifestyle interventions in maintaining the glycaemic target. HbA_{1c} levels are strongly correlated with fasting BGL,²⁶⁵ and they are often interpreted to represent 'mean blood glucose', however this is a distinguishable metric which can be measured using a glucose monitoring system. An analysis of 387 participants across three randomized trials demonstrated that HbA_{1c} may sometimes underestimate or overestimate mean glucose,²⁶⁶ and therefore they are distinguishable metrics with specific

clinical uses. Many authors agree that an assessment of HbA_{1c} is sufficient as an estimate of long-term glycaemia, as supported by the results of the Diabetes Control and Complications Trial,²⁶⁷ and this assessment is usually conducted every six months for people with T2D. However, an open cohort study conducted by Ohde and colleagues (2018) concluded that HbA_{1c} monitoring is sufficient once every 12 months for people with stable glycaemia, and this should become the clinical standard to reduce over-testing.²⁶⁸ Though there are minor discrepancies here in the literature, there is a clear consensus that long-term measures of BGL are useful in tailoring specific treatments to individuals to achieve glycaemic targets, supporting their use in clinical settings.

The HbA_{1c} assessment is a useful diagnostic tool for diabetes,²⁶⁹ and is also the gold standard for monitoring long-term glycaemic management in diabetes.²⁶⁴ The available evidence for the prevention of diabetes complications is to maintain HbA1c below 7%, and there is significant literature to support this, such as the Diabetes Control and Complications Trial in T1D²⁷⁰ and the United Kingdom Prospective Diabetes Study for T2D.²⁷¹ However, some authors contend that HbA_{1c} levels are insufficient in predicting the risk for diabetes complications. Frequent glycaemic excursions, which occur mostly around meals and when antidiabetic medication is taken incorrectly, also contribute to glycaemic variability, but do not necessarily contribute to HbA_{1c}.²³² As discussed, glycaemic variability is associated with the progression of complications such as retinopathy, and this is speculated to be related to hyperglycaemia-induced oxidative stress.²⁷² Though this physiological model is theoretical, new evidence suggests that reducing glycaemic variability may aid with maintaining target HbA_{1c} in reducing risk of complications.¹⁴⁴ To add to this, neither the HbA_{1c} assessment nor the blood glucose assessment are able to accurately measure short-term fluctuations in BGL.²⁷³ This is important because significant, acute variations in BGL contribute to glycaemic variability, and are associated with diabetes complications.^{274, 275} The literature advocates for the prevention of glycaemic spikes that arise commonly around mealtimes, as a means of reducing glycaemic variability.¹⁴⁴ Systems which track BGL continuously and allow real-time analysis of that data may therefore have another advantage over the static, snapshot readings provided by intermittent measures of BGL.²⁷⁶ It is important to consider, however, that the relationship between glycaemic variability and diabetes complications is new and has only a few studies to support it. Further research is needed in this area.

The glycaemic target for diabetes, which has been referenced throughout this chapter, is generally defined by $HbA_{1c} < 7.0\%$.²⁶² For some people with diabetes, such as those also living with known cardiovascular disease, this target may be stricter, such as < 6.5%.²⁷⁷ Prevention of long-term complications of diabetes is an important goal for individuals and their physicians to

work toward, and can be achieved through tracking of HbA_{1c} with routine assessments.²⁷⁸ T2D is a progressive condition, and complications are often viewed as an inevitability as opposed to a risk. Treatment with oral medication that lowers BGL is a frontline treatment, and consequently most patients with T2D will eventually require treatment with multiple drugs, known as poly-pharmacy, and also in combination with insulin.²⁷⁹ The literature stresses the importance of meeting HbA_{1c} targets at all stages of diabetes, though this becomes more complicated as diabetes progresses, due to the increased likelihood of comorbidities and concomitant medications with age and diabetes progression.²⁸⁰ The results from the GUIDANCE study (n=7,597) indicate there are many barriers to effectively meeting HbA_{1c} targets.²⁸¹ As with similar studies, only approximately half of their total sample were able to meet the optimal glycaemic target of HbA_{1c} < 7.0%. There were findings in this study that the authors claimed were counterintuitive, for example, subjects using diabetes medication and antihypertensives were less likely to meet targets for optimal glycaemia and blood pressure, respectively, suggesting that treatments are being allocated correctly, but on their own are not effective in terms of ensuring clinical endpoints. The authors Stone and colleagues (2013) also considered it surprising that the presence of one or more macrovascular complications was a predictor for optimal glycaemia, and suggested that this was because these people were more likely to have frequent appointments with their health care providers and more intensive care compared to subjects without any macrovascular complications.

In terms of early detection and screening for diabetes, HbA_{1c} is also important. A reading of HbA_{1c} that is greater than or equal to 6.5% is indicative of diabetes. These percentages roughly equate with long-term averages in BGL and can be converted to millimoles per litre. For example, an HbA_{1c} value of 6.5% equates to an average of 7.7 mmol/L BGL over the past 120 days.²⁸² An individual with < 7.0% HbA_{1c} may be described as an unmanaged diabetes case, but they will not be considered in remission from diabetes unless they are able to maintain this glycaemic threshold in the absence of medication.²⁶³ As discussed, sustained remission from diabetes is rare without surgery, but possible. A HbA_{1c} level above 7.5% indicates a person's diabetes is unmanaged, as they are at risk of long-term complications, and a person with HbA_{1c} between 7.0% and 7.5% is not necessarily unmanaged, but their glycaemic is not optimal and they should aim to reduce their glycaemia to the target level of 7.0%.²⁸³ However, a large proportion of people living with diabetes do not consistently reach the target of $HbA_{1c} < 7.0\%$.¹⁷⁵ Many factors contribute to this, however the literature has identified that a key factor in the inability of many people living with diabetes to maintain target glycaemia is that current glucose monitoring tools are limited. The inability of people with diabetes to meet glycaemic targets is not entirely caused by limitations of glucose monitoring tools – lifestyle interventions

are often difficult to meet due to the social and commercial pressures of the Western lifestyle, as discussed in the first section of this chapter, and treatments should also be improved. However, this thesis focuses only on glucose monitoring tools. The following sections will provide further evidence for why current standards in glucose monitoring are insufficient given the scope of diabetes.

1.7.4 Achieving Glycaemic Targets

A core concept of this chapter is that glucose monitoring, when applied properly, leads to significant improvements in glycaemic management and allows people to achieve target glycaemia for longer periods. This ultimately leads to a better quality of life as well as a reduced risk for acute and chronic diabetes complications. As diabetes is a rising global health issue and a leading cause of premature death and illness, extra significance is awarded to technologies which aim to reduce the burden of diabetes, including those which aim to improve glucose management.²³ However, despite the overwhelming consensus that stringent glycaemic management is the most ideal course of diabetes management, many people living with diabetes are unable to achieve consistent glycaemic management.²⁸⁴ This is determined by the ability of a person to maintain their HbA_{1c}, or long-term blood glucose, below 7% (Section 1.7.3.2). A largescale study concluded that almost 50% of the Australian diabetes population were unable to maintain their HbA_{1c} levels at the recommended level of <7.0%.¹⁰⁴ Other studies suggest this is a global trend, and confirm that about half of all people living with diabetes are unable to reach this glycaemic target.²⁸⁴ Furthermore, this trend is more apparent in groups that are insulin dependent. Between 2016 and 2018 in the United States, only 21% of people with T1D receiving specialized care achieved the glycaemic target of HbA_{1c} < 7.0%.²⁸⁵ According to Rickels (2020), advancements in glucose monitoring and insulin delivering technology in recent years have not led to corresponding improvements in glycaemic management for insulin-dependent adults or youths with diabetes.⁹⁸ Grenier and colleagues (2016) agree that there is an important opportunity to improve on the current standards in diabetes care and treatment outcomes.²⁸⁶

To reiterate, the main concern is that 7% HbA_{1c} represents a major clinical threshold in diabetes, and that the inability to maintain HbA_{1c} below this threshold is associated with increased risk for diabetes complications. The DIALECT-1 study found that 64% of a T2D group (n=450) were unable to achieve target glycaemia, and the authors attributed this to increased resistance to insulin and antidiabetic medication, which occur over time as diabetes progresses.¹⁷⁵ Despite this, it has been discussed that the inability to achieve target glycaemia is not necessarily an inevitability of diabetes progression, but is something that can be amended

by improving medication and glucose monitoring.²⁸⁷ There is growing recognition that the major problems of CGM systems and SMBG are key drivers of not meeting optimal glycaemic targets.²⁸⁸

1.7.5 Barriers

Barriers to diabetes management generally refer to factors which reduce the effectiveness of management strategies and overall treatment effectiveness. However, of particular importance to the literature are barriers specific to SMBG. Since so much of diabetes management relies on day-to-day self-medication and self-monitoring performed by the individual, effective SMBG is of critical importance to proper diabetes management.^{289, 290} However, several systematic reviews have recognised that there are many barriers to SMBG which reduce its effectiveness, and research in this area is severely lacking given the scope of diabetes.²⁹¹⁻²⁹³ The review conducted by Saha (2019) labelled current evidence of barriers to SMBG as inadequate and weak.²⁹² Even among primary care physicians, attitudes toward the adequacy of SMBG lack a consensus.²⁹⁴ However, there is an association between education and efficacy in regards to SMBG. The use of smartphone technology, integrated with healthcare systems, was also shown to be associated with better use of SMBG and increased effectiveness of diabetes management.²⁹² This is promising because it indicates that convenience is important to people who rely on themselves to monitor their own chronic condition, such as SMBG in diabetes. Improving convenience of SMBG technologies may also improve use, and thus improve treatment outcomes.²⁹⁵ This is reinforced by Ng and colleagues (2020) who list inconvenience as a major barrier to SMBG. Other barriers listed in this 2020 paper include fear of pain and injection, fear of side effects of medication, and concern that resorting to insulin indicated the individual had reached end stage diabetes.²⁹¹ Though SMBG by blood glucose assessment is generally cheaper than CGM, many people with T1D or T2D still report high cost as a barrier to SMBG.²⁹⁶ Consequently, SMBG is low in people with diabetes, and it is even lower in groups with less severe symptoms, such as those who are at low risk of hypoglycaemia or acute hyperglycaemic.297

Another barrier to maintaining target glycaemia is the fact that many people with diabetes are undiagnosed or otherwise unaware of their condition. Approximately 50% of all people with diabetes are unaware of their condition, and though rates of undiagnosed diabetes are lower in developed nations (25-33%),²⁹⁸ they are estimated to be as high as 75% in some developing nations.¹⁰³ As discussed, this is likely related to the rising prevalence of diabetes worldwide and the fact that symptoms of hyperglycaemia can be quite mild and have a slow onset.⁸² The

probability that people who are unaware of their diabetes are actively managing their condition are near zero. As such, the prevalence of undiagnosed diabetes and current limitations of SMBG and CGM are key factors that are implicated in the inability of half of all people living with diabetes to achieve target glycaemia. The development of a new glucose monitoring technology which addresses the concerns facing current SMBG and CGM could lead to significant improvements in glycaemic management in those with diabetes. If implemented as a routine health assessment conducted by general practitioners and other primary care physicians, this technology could also reduce the prevalence of diabetes unawareness.

Engaging with recommended treatment and lifestyle changes is an important consideration for any chronic condition, but it is especially relevant in diabetes. This is because lifestyle and diet interventions can be difficult to maintain over long periods of time,²⁹⁹ and medications such as glucose-lowering drugs can lose their efficacy over time as diabetes progresses.³⁰⁰ This can lead to increases in medication dosage or addition of other medications, such as insulin, to influence BGL. Consequently, even those who do engage strongly with their treatment plan may not be able to prevent their diabetes from progressing or worsening.³⁰¹ The feeling of frustration associated with this is one of the most commonly reported barriers to stringent diabetes management – another is frustration from lack of glycaemic management.²¹¹ The effect of these barriers may be alleviated by improving SMBG techniques, such as improving their ease of use, as SMBG in general is associated with better glycaemic management.²¹⁰ The results of a large international study (n=5,104) indicated that use of SMBG was 44% in T1D adults and 24% in T2D adults.³⁰² Other studies have reported similar trends in T1D³⁰³ and in T2D.³⁰⁴ The need to overcome barriers to diabetes management is a core component of this PhD candidature, as it can improve treatment outcomes and lead to a significantly reduced burden from diabetes. To summarise the previous sections: complications of diabetes are highly prevalent and represent a significant burden to those living with diabetes, and SMBG is an important tool for both T1D and T2D to improve glycaemia and reduce these complications.

1.7.5.1 Invasive Technologies

A common element to all the measures of glycaemia discussed so far is that they are all invasive. This refers to their nature as tools which can only function by breaking the skin and accessing the tissues underneath. This is an unfortunate component of current diabetes management and glucose monitoring. The most accurate measures of BGL currently available require direct access to the blood, which is invasive and painful, and tools that estimate BGL by accessing alternative tissues, such as skin, mucous, and saliva, are less accurate.²⁰⁰ These will be discussed

further in the following sections. CGM represents a strong middle-ground, where a sensor requires subcutaneous access for continuous monitoring, which is only minimally invasive, and the trade off in accuracy is acceptable.²⁰¹ However, even with minimally invasive technologies, there is still a requirement that the skin layer is penetrated. The pain associated with these procedures represents a psychological barrier to diligent BGL monitoring,³⁰² and is a common reason for suboptimal CGM.³⁰⁵ This problem is not exclusive to CGM however, as all commercially-successful and clinically-relevant glucose monitoring tools are invasive, including the blood glucose assessment used in SMBG.³⁰⁶ Research by Tanaka and colleagues (2018) demonstrates that people who experience pain from SMBG place less value on SMBG, and pain is associated with lower health-related quality of life, higher mental distress, and higher HbA_{1c} regardless of their number of daily blood glucose assessments.³⁰⁷ According to one study, people who experience fewer barriers to SMBG, such as pain, inconvenience, and cost, are more likely to use SMBG (n=933).³⁰⁸ The literature agrees that the invasive nature of all commercially-available glucose monitors is cause for inefficiency in diabetes management.²⁸⁸ A non-invasive alternative to current SMBG standards can improve routine glucose monitoring as it removes a key barrier, which is pain and fear of invasive procedures.²⁸⁸ However, such a technology would need to match or improve on the accuracy and reliability of current glucose monitors, whilst meeting certain other criteria of ideal monitoring.

1.7.6 Ideal Monitoring

Firstly, glucose monitoring should ideally be continuous in nature. As discussed, a clear strength of CGM systems is they can track glucose levels in real-time and provide meaningful information to people with diabetes and their physicians for optimising treatment. This is true for both T1D and T2D. CGM provides better glycaemic management than non-continuous measures^{309, 310} – it is simply a matter of weighing the costs against the benefits, as current CGM systems are expensive. The ability to monitor BGL constantly cannot be understated in its importance. It allows for individuals to intervene when their glucose levels elevate or drop outside the target range. This can occur frequently and unexpectantly in response to meals, exercise, and medication, and can occur even when a person is asleep. It is not sufficient to simply maintain BGL at an optimal range for long periods of time – these spikes in BGL, even if they are very short in duration, should also be avoided to reduce the risk of diabetes complications. Combined with the fact that hypoglycaemia is prevalent in insulin-dependent diabetes, and may present at unpredictable times, it should be clear that glucose monitoring is more effective when it is continuous in nature.³¹¹

Secondly, glucose monitoring should provide data in real-time. According to a meta-analysis, CGM is more effective and leads to better glycaemic management compared to SMBG when it provides glucose data in real-time, but not when it is retrospective.³¹² This means that a continuous measure of blood glucose should ideally provide information immediately to the individual, and many studies have indicated that integrating real-time CGM with smartphone applications may significantly improve convenience and thus efficacy of stringent glucose monitoring.^{295, 313} Systems which alert people with diabetes in real-time to the onset of hypoglycaemia provide a significant decrease in the rate of severe hypoglycaemia, compared to systems with no alert.³¹⁴ Additionally, a major concern with current glucose monitoring tools is that they are all invasive. An ideal measure of BGL would combine elements of SMBG and CGM and be non-invasive. Therefore, an ideal glucose monitoring system should be continuous, noninvasive, accurate, cost-effective, and provide data in real-time with little to no lag time.³¹⁵ The ability to predict glucose levels in the near future may also be beneficial. Of significance to the present research is a technology that is both continuous and non-invasive, as no commerciallyavailable technology has managed to achieve this to date, and this would represent a significant improvement for those currently utilising SMBG, as well as some who utilise CGM.³¹¹

1.8 Non-Invasive Glucose Monitoring: The Future Standard

Emerging technologies seek to develop an ideal method for monitoring BGL, which would involve a non-invasive, accurate measure of BGL that provides information continuously and in real-time. SMBG and CGM are the most used methods of glucose monitoring in T1D and T2D and are both invasive. Non-invasive technologies are those which analyse physiological variables, including glucose-containing fluids such as saliva and urine, without drawing fluid out of the skin.³¹⁶. They aim to avoid the pain and irritation associated with invasive and minimally invasive devices, though it is important to note that some of these fluid-based technologies, such as those which rely on urine samples, can only provide intermittent measures of glycaemia due to limited supply.³¹⁷ Figure 1.10 summarises the invasive technologies discussed so far (left side) as well as the emerging non-invasive technologies (right side), some of which will be discussed in the next section. At present, none of these are sufficiently accurate and at best only estimate BGL.³¹⁸

Figure 1.10 Current methods for invasive and non-invasive measurements of blood glucose.



Figure 1.10 broadly outlines the different types of invasive and non-invasive technologies used to measure blood glucose. Invasive systems involve inserting a sensor through the skin to achieve direct access to the blood or interstitial fluid. Non-invasive systems use sensors that do not require the skin to be broken to access the target tissues or fluids. The HbA_{1c} assessment (not shown here) is also an invasive technology as it requires a blood sample. Adapted from do Amaral & Wolf (2008).³¹⁹

1.8.1 Absorption Spectroscopy

Many efforts to develop a non-invasive measure of BGL employ some form of absorption spectroscopy, including near-infrared or mid-infrared.³²⁰ Changes in the absorption of light at different wavelengths reflect characteristics of glucose molecules in the blood.³²¹ However, the optimal range of the electromagnetic spectrum to use in these investigations remains a controversial topic, and there is no standardised method. The use of wavelengths in the mid-infrared region, mostly between 8382 and 9708 nm, produces more distinct glucose peaks, compared to spectra produced at lower wavelengths, but there is limited light penetration.³¹⁹ Studies conducted in the therapeutic range between 600 and 2500 nm, such as the visible and near-infrared spectrum, are favourable because they allow for deep tissue analysis.³¹⁹ However, molecules other than glucose absorb in the near-infrared range as well, such as water and fat. In general, the precision of this technology is too poor for clinical use,²²⁶ and requires significant work and modification to be viable.

1.8.2 Fluorescence

The use of fluorescent measurements in the development of a non-invasive measure of BGL is another emerging technology. It relies on the principle that glucose levels in tears reflect similar concentrations to those in blood.³¹⁹ Some of the major benefits of fluorescence include its extreme sensitivity to glucose and its resistance to fluctuations in ambient light intensity.³²² Additionally, this technology is promising due to its convenience, as the sensor can be incorporated into clear contact lenses that can be worn by the individual.³²³ However, since glucose levels are tracked with a 30-minute lag time, this may be a concern as current CGM systems have much shorter lag times.³¹⁹ Due to the poor accuracy of these emerging technologies, Tronstad and colleagues (2018) suggested a multi-sensor approach may work better, whereby information from multiple non-invasive devices, such as fluorescence sensors, bioimpedance, and skin temperature measurements, is combined to improve accuracy in detecting glycaemic events, making them fit for clinical use.³²⁴ Though interest in this area is still gathering, more larger-scale studies are required to address some of these concerns.

1.8.3 Reverse Iontophoresis

Another technology that was developed in the pursuit of a non-invasive measure of BGL is reverse iontophoresis. This relies on a transdermal method, whereby small amounts of interstitial fluid are extracted through the skin with a device.³²⁵ Devices can produce glucose estimations at 10 minute intervals, and this is comparable to current CGM systems. Traditionally, iontophoresis is used to electrically-charge drug molecules to transport them across the skin and into the blood, and thus this technology works in reverse by drawing fluid out of the initial space onto the surface of the skin, where it can be analysed.³²⁶ The Food and Drug Administration approved a wrist-watch device capable estimating BGL by reverse iontophoresis, though the use of this technology in this example is technically minimally invasive, as fluid is being drawn through the skin.³²⁷ Some of the problems with reverse iontophoresis include long calibration periods, skin irritation, and poor accuracy resulting from movement, exercising, and sweating.³¹⁹

1.8.4 Summary

In conclusion, though much interest has been gathering in these areas, the development of a non-invasive, continuous marker of BGL has, to date, been unsuccessful. A recent systematic review and meta-analysis conducted in 2021 concluded that there were no minimally or non-invasive glucose monitoring systems fit for routine monitoring of hypoglycaemia.²⁸⁸ In this paper, the authors Lindner, Kuwabara, and Holt state that current systems are not sufficiently accurate for detecting hypoglycaemia in routine use.²⁸⁸ Lin and colleagues (2017) have also highlighted the lack of a commercially successful non-invasive glucose monitor, despite new technologies emerging in the past 10 years.³¹⁸ As such, there is an important opportunity to develop a novel non-invasive measure of BGL. If sufficiently accurate, this technology could

significantly decrease medical costs and deaths related to diabetes, as the convenience of noninvasive monitoring may improve glycaemic management and therefore glycaemic control.³²⁸

1.9 Glucose Prediction: Novel Applications

With the advent of machine learning, artificial neural networks, and deep learning algorithms, new computer-based approaches to glucose monitoring are emerging. Models are trained on large amounts of data, and when used in a clinical setting these models can provide an estimate of an outcome variable (such as BGL) based on several predictor variables, which could be related to medical history, demographics, or other physiological variables. There is a consensus that some of the problems with current CGM systems, including lag time and low accuracy in the hypoglycaemic range, can be improved by integrating machine learning algorithms and other predictive models into CGM systems.^{329, 330} These models aim to improve the accuracy and reliability of current glucose monitors, such as CGM systems, to deliver more personalised care to individuals with diabetes³³¹ and allow for more timely detection and intervention of hypoglycaemia and hyperglycaemia (the outcome variables).³³² However, a core problem of these models is that they rely on data recorded by glucose monitors that are already in use, including CGM systems. Although machine learning models have led to more accurate prediction of dysglycaemia in combination with glucose monitors, glucose monitors are still invasive in nature and are not used routinely outside of T1D. Most people living with diabetes do not use CGM systems and will not benefit from algorithms which increase their accuracy. Therefore, this thesis is interested in predictive models that use information from non-invasive tools, as it is the invasive nature of current glucose monitoring systems that is cause for concern.

1.9.1 Based on Alternative Measures

Alternative means of predicting BGL – such as those which do not involve a CGM system or SMBG – represents a novel but growing area of interest in the literature. One example is the use of diabetes alert dogs, which can be trained to detect the onset of hypoglycaemia by smell.³³³ However, Los and colleagues (2016) demonstrated that service dogs tend to raise an alert with a high false-positive rate, and by comparison a CGM system detects hypoglycaemia significantly earlier than a trained dog, with a median difference of 22 minutes.³³⁴ As such, there is a need for a non-invasive marker of diabetes, with good potential for screening diabetes risk as well as the ability to estimate or predict BGL.

Based on work from Jung (2016), machine learning may be able to predict a person's risk of hypoglycaemia using information that can be entered into a smartphone, such as surveys, interviews, and diaries.³³⁵ In a hospital setting, the risk of hypoglycaemia may be predicted using electronic medical records alone. A machine learning model developed by Ruan and colleagues (2020) was trained on a data set from 17,658 subjects with diabetes using their oxygen saturation, medications, type of diabetes, and type of procedures the person was undergoing.³³⁶ Though the work of Jung (2016) and Ruan and colleagues (2020) may have limited applications, they highlight the importance of demographic data – such as age, height, and weight – and medical history in strengthening the value of predictive models. Recent work from Lama and colleagues (2021) indicates that there is still current interest in the capabilities of machine learning to predict glycaemia. Their algorithm was able to predict the risk of developing T2D in people without diagnosed diabetes based on their demographic and physiological data. They concluded that the strongest predictors of diabetes risk were body mass index, waist-hip ratio, age, systolic and diastolic blood pressure, and family history of T2D.³³⁷ Foss-Freitas and colleagues (2019) contend that heart rate may be an important predictor of hypoglycaemic events, as their algorithm learned from data from a continuous heart rate monitor and provided useful predictions.³³⁸ This is significant because heart rate monitors are non-invasive, relying on access to the surface of the skin only. The relevance of cardiac rhythms in predicting BGL is growing in recognition in the literature and is based on the principle that autonomic control of the heart reflects autonomic control of BGL. In theory, autonomic activity may predict BGL.

1.9.1.1 Neural Regulation of Blood Glucose

Both the central and peripheral nervous system are involved in glucose homeostasis. However, neural regulation of the pancreas, liver, and gastrointestinal tract, which are the primary organs involved in actioning glucose homeostasis, is the domain of the autonomic nervous system.³³⁹ The autonomic nervous system is the branch of the peripheral nervous system which regulates the metabolic, visceral, and vascular systems of the body. As such, neurodegeneration of autonomic nerves, as seen in diabetic autonomic neuropathy, affects many parts of the body.¹¹⁵ Combined, the autonomic nervous system and the somatic motor system constitute the entire neural output of the peripheral nervous system.³⁴⁰ The somatic motor system controls skeletal muscles, and the autonomic nervous system controls every other tissue and organ in the body that is innervated, including secretory glands, the heart, blood vessels, organs of the digestive and excretory systems, and many others.³⁴⁰ There are two counter regulatory branches which comprise the autonomic nervous system (Figure 1.11). Activation of the sympathetic nervous

system is associated with increased heart rate, vascular constriction, and dilation of the pupils, and is often characterised as the 'fight or flight' response.³⁴¹ Activation of the parasympathetic nervous system has an opposite effect, lowering heart rate, increasing digestion by dilating blood vessels in the gut, and constricting the pupils. It is thus characterised by the 'rest and digest' state.^{342, 343} At rest, restorative and vegetative functions aim to conserve energy by way of parasympathetic innervation of the heart and other organs.³⁴⁴

As the autonomic nervous system is responsible for restoring glucose homeostasis through innervation of the pancreas, liver, and gut, the ability to measure sympathetic or parasympathetic innervation of these organs may theoretically provide meaningful predictions of glycaemia, or indicate if current BGL is high or low. However, autonomic control of these organs cannot be directly measured by non-invasive means.³⁴⁵ This is due to the complex neural processes involved and the lack of a dedicated technology that can non-invasively measure autonomic input to the organs of interest.³⁴⁶ However, autonomic activity in general is strongly correlated with autonomic control of the heart, which can be non-invasively measured.³⁴⁷ Though there are various means of assessing autonomic function in a clinical setting, such as the Valsalva manoeuvre, there is a clear consensus in the literature that heart rate variability (HRV) is the best non-invasive measure of autonomic activity.³⁴⁸⁻³⁵¹



Figure 1.11 Sympathetic and parasympathetic innervation of the human body.

Figure 1.11 illustrates the two branches of the autonomic nervous system: the sympathetic and parasympathetic divisions. These branches maintain homeostasis of internal systems through opposite, counter regulatory actions. For example, sympathetic innervation of the stomach inhibits digestion, whilst parasympathetic innervation stimulates digestion. Adapted from Bear et al. (2016).³⁴⁰

1.10 Heart Rate Variability: An Autonomic Marker

Understanding how HRV is distinguishable from heart rate is relevant to this thesis. An individual's heart rate is a measure of how many times their heart contracts or beats per minute. Acute increases in heart rate can be related to exercise, as there is increased metabolic demand, or even stressful periods, as the nervous system prepares for a 'fight-or-flight' scenario in response to a perceived threat.³⁵² Heart rate also fluctuates naturally in a resting state, synchronised with the respiratory cycle. When breathing in, heart rate increases momentarily to increase the efficiency of the exchange of oxygen and carbon dioxide between the fresh air and the alveoli in the lungs.³⁵³ Heart rate increases when inhaling, as the return of blood to the heart decreases due to increase in thoracic pressure on blood vessels. This leads to the corresponding increase in heart rate, as per Frank Starling Law. When exhaling, pressure in thoracic cavity decreases, which increases blood flow to the heart, and therefore a decrease in heart rate to maintain blood pressure.³⁵⁴ When exhaling, heart rate decreases momentarily to allow for a rest period. This is known as respiratory sinus arrhythmia, a reflection of vagal or parasympathetic control of the heart.³⁵⁵ The effects of metabolic state, respiration, and the autonomic nervous system on the cardiac cycle are well-known topics in the literature, with new understandings emerging in novel areas.³⁵⁶ In the 1990s, it was determined that the optimal rhythms of a heart are not like a metronome; rather, they are complex and nonlinear.³⁵⁷ This variability in heart rate, or HRV, describes the natural tendency of the cardiac cycle to vary between subsequent cycles, a process which reflects the body's ability to adapt to environmental and psychological challenges. Optimal HRV is associated with good health and cardiac adaptability and resilience.³⁵⁸ Blood pressure, including systolic and diastolic indices, is another example of a physiological variable which fluctuates on a minute-to-minute basis in response to changing demands.359,360

Of significance to this thesis is the notion that HRV is assessed through non-invasive means, usually an electrocardiogram (ECG) or heart rate recording, and that HRV is the best marker of autonomic activity. This is due to a combination of various factors, such as accuracy and reliability, the ability to assess it non-invasively, and the ability to measure specific components of autonomic tone. The latter point is possible because different HRV measures reflect different autonomic influences over cardiac activity, including sympathetic and parasympathetic influences.³⁶¹ These unique HRV measures are not measured in beats per minute. Rather, they are calculated by applying complex mathematical equations (see Section 2.1.5.1 and 2.1.5.2 in Methods for more details) to an ECG waveform or heart rate recording.³⁶² Figure 1.12 illustrates the main features of a typical ECG waveform, from which computer programs can extrapolate HRV measures. Specifically, the calculation of HRV measures requires a recording of many R-R

intervals, which can also be achieved with a simple heart rate monitor. A heart rate monitor measures the pulse wave, and from this the pulse rate is determined. An R-wave on an ECG waveform denotes the electrical activity of the heart during its main ventricular contraction, or heartbeat, and an R-R interval denotes the time in milliseconds between two successive heartbeats. Although the main rhythm of the heart is determined by the sinus node – the natural pacemaker of the heart – it is the autonomic nervous system which determines the final heart rhythms through the innervation of sympathetic and parasympathetic nerve fibres.³⁶³ Parasympathetic innervation of the heart is achieved via the vagus nerve, though this nerve shows dysfunction in the early stages of diabetic autonomic neuropathy.³⁶⁴ It has been proposed that the vagus nerve is an early target of neuropathy because it is the longest nerve of the autonomic nervous system.³⁶⁵ The vagus nerve is also continuously active as it is an integral autonomic nerve, and as such is more susceptible to free radical damage, which is increased with hyperglycaemia.





Figure 1.12 illustrates how heart rate variability can be determined from an ECG waveform. A typical waveform provides five 'landmarks' of cardiac activity, including the 'P-wave', 'T-wave', and the 'QRS complex'. **P** = atrial depolarisation. **Q** = depolarisation of interventricular septum. **R**_{1,2,3} = depolarisation of the main ventricular mass, which is the 'heartbeat'. **S** = ventricular depolarisation. **T** = ventricular repolarisation. The R-wave is the most important component of an ECG waveform in determining heart rate variability. The QRS interval precedes ventricular contraction and measures the electrical activity of the ventricle. Adapted from Ortiz Guzmán et al. (2012).³⁶⁶

1.10.1 Applications in Diabetes

Autonomic neuropathy is a major complication of diabetes, driven by unmanaged hyperglycaemia and duration of diabetes,³⁵⁰ as long-term hyperglycaemia causes neurodegeneration in the peripheral nervous system.³⁶⁷ Neuropathy of the autonomic nerves is also driven by hypoglycaemia, and reduced autonomic function then predisposes individuals to further hypoglycaemic events.³⁶⁸ This can lead to a cycle of impaired awareness of hypoglycaemia, as described in Section 1.5.1.3. Neuropathy is an independent risk factor for myocardial infarction and other major cardiovascular events, and early detection of neuropathy is essential in reducing the risk of morbidity and mortality in people living with diabetes.³⁶⁹ The ADA advocates the importance of monitoring autonomic function in diabetes due to the severe nature and prevalence of neuropathy.³⁷⁰ Though some clinicians prefer a symptomatic approach to monitor the onset or progression of neuropathy, HRV can reliably detect autonomic decline before it becomes symptomatic, providing the opportunity for early intervention and prevention of further autonomic decline.³⁷¹ Recent interest in the applications of computational methods in medical science has indicated that models derived from machine learning may be embedded with clinical information systems to predict a person's risk of developing complications.³⁷² This may assist with specific treatment and improving treatment outcomes.

There is a consensus that lower HRV correlates with impaired autonomic activity, and all measures of HRV are reduced in diabetes.³⁷³⁻³⁷⁶ This represents a large area of research. Significant decreases in HRV related to diabetes occur within the first 5-10 years of diabetes onset.³⁵⁰ HRV is an excellent measure of vagal tone, which represents the function of the primary parasympathetic nerve: the vagus nerve.³⁴⁴ Cardiac vagal tone is a predictor of parasympathetic tone in diabetes, and since it can be measured non-invasively with an ECG or heart rate monitor, it represents a convenient method of monitoring autonomic decline in diabetes.³⁷⁷ Other authors concur that cardiac vagal tone is sensitive to autonomic neuropathy in diabetes, and thus HRV provides an affordable, accessible, and easily applicable screening method for autonomic dysfunction.³⁷⁸ There is significant literature to justify the use of HRV in the prognosis of autonomic neuropathy in diabetes.^{348, 352, 379} Though HRV may be a useful tool for primary care physicians to monitor the progression of a major long-term complication of diabetes, this thesis is only interested in the novel application of HRV measures in monitoring glycaemia.

1.10.2 Applications in Glucose Monitoring

In cross-sectional studies, HRV measures are inversely associated with BGL. In general, people with diabetes have suboptimal glycaemia, and this is associated with increased hypoglycaemia and hyperglycaemia. Glycaemic excursions contribute to autonomic neuropathy, and as demonstrated by Deshmukh and colleagues (2021), this is not reversible with euglycaemia restored by islet cell transplantation in T1D.³⁸⁰ People who live with persistent hyperglycaemia and reoccurring hypoglycaemia, have lower HRV measures, which indicates poorer autonomic function.^{381, 382} This relationship is strongest in cases of unmanaged glycaemia, as well as longterm cases of diabetes. This correlation may be due to a range of independent risk factors such as obesity, hypertension, and increasing age which all contribute to both lower HRV and higher BGL.^{313, 371} Measures of HRV and BGL are also general indicators of health, which may explain why many different cross-sectional studies agree that there is an association between the two variables. As such, it has been proposed that HRV, determined by an ECG or heart rate monitor, may be a useful non-invasive measure of general glycaemic health. Machine learning may be useful in strengthening the ability of HRV measures to predict diabetes or high BGL in an individual. The use of HRV measures as predictors of glycaemia has been proposed previously in the Neuroscience Research Unit at the University of Technology Sydney (UTS) by Rothberg and colleagues (2016).³⁸³ This PhD candidature is a continuation of this research, and aims to further investigate the viability of non-invasive HRV parameters as a measure of BGL in people with or without diabetes.

1.11 Basis for Research

This chapter has so far discussed the global nature of diabetes as well as the specific threats posed by diabetes and their significance. To generalize: diabetes causes significant burden to affected individuals and to world governments, both developed and developing, and due to a combination of various cultural and commercial pressures, the prevalence of diabetes is projected to continue increasing.^{384, 385} There is a question of whether the global response to diabetes is adequate, given the large scope of diabetes, especially T2D.^{384, 386} In this thesis, however, the focus is on whether current standards in diabetes management are adequate. Given the information provided in this chapter, current management strategies are inadequate for those living with diabetes. Current technologies need to be improved, or a new technology needs to be introduced, to address this inadequacy. The aim of this thesis is to address some of the main concerns facing current standards in glucose monitoring by proposing an alternative marker of diabetes and BGL which may have clinical utility. A non-invasive measure of BGL,

relying on predictive models trained on alternative physiological data points, such as cardiac rhythms, is clearly an area of interest in the literature. A novel non-invasive technology may be implemented in clinical settings alongside the glucose tolerance test, blood glucose assessment, or even used in combination with CGM to improve accuracy and sensitivity of these existing tools for managing dysglycaemia.³⁸⁷ Though there are many non-invasive measures of BGL under development, such as those which record glucose levels in the saliva and sweat, none are commercially successful or clinically relevant. In this thesis, a novel predictor of BGL is proposed, and the aim is to justify further research into HRV measures as novel markers of BGL.

The long-term effects of suboptimal BGL on autonomic function are well-established within the literature, ^{352, 367} and HRV is widely considered to be the best marker of autonomic activity.³⁴⁸⁻³⁵¹ Also, HRV is more sensitive to glycaemic fluctuations compared to conventional assessments of autonomic function, including the orthostatic test, the Valsalva manoeuvre, and the controlled and deep breathing test.³⁸⁸ The current literature on associations between HRV and BGL is limited,³⁵⁰ and is summarised in Table 1.2. Though limited, current literature has established that measures of HRV are diminished in T1D and T2D, and that HRV is inversely associated with BGL. A recent systematic review and meta-analysis of 25 studies (n=2,932) conducted by Benichou and colleagues (2018) reinforced this consensus. The main link that requires further investigation is whether HRV measures can predict short-term fluctuations in glucose levels, including hypoglycaemia and hyperglycaemia, as well as non-pathological fluctuations in BGL such as those seen after a meal. During certain glycaemic excursions, such as hypoglycaemia, there is a distinct autonomic response defined by autonomic symptoms, such as heart palpitations and sweating. HRV, measured non-invasively, may be able to identify distinct patterns in autonomic activity preceding hypoglycaemia or responding to declining BGL. In general, the ability for HRV measures to predict glycaemia is a novel area that requires further research. Specifically, it is only well-known that HRV is related to BGL when assessed crosssectionally, and not whether short-term fluctuations in BGL may also correspond with changes in HRV measures. This may lead to the development of an regression model capable of estimating the direction and magnitude of BGL based on non-invasive means, as suggested by previous work.389

On a minute-to-minute basis, this relationship between HRV measures and BGL is complex. Acute changes in autonomic activity precede certain glycaemic events, such as the increase in BGL after a meal (postprandial BGL). The work of D'alessio and colleagues (2001) in rhesus macaques demonstrated parasympathetic innervation of the pancreas increases the secretion of insulin, which lowers BGL in anticipation of a postprandial glucose spike.³⁹⁰ This suggests an association between autonomic activity whilst eating and postprandial BGL. However, no study to date has demonstrated an association between postprandial BGL and HRV measures recorded in a prandial state (whilst eating). Such research could justify the development as HRV measures as predictors of postprandial BGL. Furthermore, in response to hypoglycaemia, the autonomic nervous system increases sympathetic innervation of the pancreas to secrete glucagon and restore euglycaemia.³⁹¹ Therefore, HRV measures observed during specific time periods or metabolic states may provide meaningful information for estimating current or future BGL. Currently, there is a lack of consensus regarding the specific changes in autonomic activity, reflected by HRV measures, during hypoglycaemia and hyperglycaemia. The aims and hypotheses in this thesis are based on these gaps in the literature.

Table 1.2 Summary of key findings from studies assessing frequency-domain heart rate variability parameters and blood glucose levels in groups with diabetes and groups without diabetes. *All correlations are significant. **BGL** = Blood glucose level, **CGM** = Continuous glucose monitoring, **ECG** = Electrocardiogram, **HF** = High frequency, **LF** = Low frequency, **LF:HF** = low to high frequency ratio, **T1D** = Type 1 diabetes, **T2D** = Type 2 diabetes.

Authors and year	Sample group	Sample size (n)	Mean age (years)	HRV method	BGL method	Was BGL inversely related to LF and HF?	Limitations
Singh et al. (2000) ³⁹²	Adults without diabetes	1779	47.4	2-hour ambulatory ECG	Fasting BGL	Yes. Also, BGL inversely correlated with LF:HF.	Data was pooled for all 3 groups. HRV vs. BGL was not analysed in each group separately.
	Impaired fasting glucose	56	54.9				
	Diabetes	84	55.4				
Weissman et al. (2006) ³⁹³	Pregnant adults without diabetes	15	Not shown.	ECG	Oral glucose tolerance test	Mostly yes. Increase in BGL led to decrease in HF and increase in LF:HF. LF was not correlated with BGL.	Autonomic processes are complicated by pregnancy. Small sample size.
Stein et al. (2007) ³⁹⁴	Adults without diabetes	1089	72	24-hour Holter ECG	Fasting BGL	Yes	Sample was mostly elderly (mean age 72).
	Diabetes	178					
Jarczok et al. (2013) ³⁹⁵	Adults without diabetes	2441	41.9	24-hour ambulatory heart rate recording	Fasting BGL	Yes	Did not control for medications or illness. Predominately male (75%).
Tarvainen et al. (2014) ³⁵⁰	Adults without diabetes T2D	189 93	64	20-min ECG	Fasting BGL	Only for LF power. HF power not significantly correlated.	Older sample and mostly female (~60%).
Klimontov, Myakina & Tyan (2016) ³⁹⁶	T2D women	67	65	24-hour Holter ECG	24-hour CGM	Postprandial BGL reflected lower LF, but not HF. Eating caused a decrease in LF only.	Conditions of study did not reflect a regular daily routine.
Lutfi & Elhakeem (2016) ³⁹⁷	Adults without diabetes	42	25.8	5-minute ECG	Fasting BGL	No. BGL positively correlated with HF, and inversely correlated with LF:HF.	Small sample size. ECG recordings were only 5-minutes long.
Rothberg et al. (2016) ³⁸³	Adults without diabetes	31	27.9	10-minute ECG	Fasting and postprandial BGL	Yes, but only in T2D sample. Also, BGL inversely related to total power in T2D.	Small sample size. Did not adjust for kilojoule intake as a covariate in postprandial analysis.
	T1D	21	31.5				
	T2D	11	56.2				
Jarman et al. (2021) ³⁸⁹	Adults without diabetes	25	27	10-minute ECG	Fasting and postprandial BGL	Yes, but only for postprandial BGL.	Small sample size. No diabetes sample

In Table 1.2, note the similarities between the studies by Rothberg and colleagues (2016) and Jarman and colleagues (2021). This is because the latter was a continuation of the former, conducted in the same research group: the Neuroscience Research Unit. In these studies, low frequency (LF) power, which represents sympathetic and parasympathetic nervous system activity,³⁹⁸ and high frequency (HF) power, which reflects parasympathetic nervous system activity, were inversely associated with BGL. The low to high frequency ratio (LF:HF) is also mentioned in Table 1.2, and relates to sympathovagal balance. LF power, HF power, and LF:HF are three common measures of HRV used in the literature and in this thesis. The specific physiological correlates of these HRV measures will be discussed in more detail in later sections. At this point it is relevant only to understand that many different measures of HRV can be extrapolated from an ECG waveform or heart rate recording by mathematical equations, and that these HRV measures reflect autonomic activity.

In their 2016 paper, Rothberg and colleagues observed people with diabetes (n=33) and people without any chronic illness (n=31) before and after a meal and assessed fasting and postprandial HRV and BGL. They identified several associations between BGL and HRV parameters recorded in a fasting and a postprandial state, though these findings were only significant in the diabetes group.³⁸³ The work of Jarman and colleagues (2021) continued this initial exploratory research by only studying people without any chronic illness or regular medications, and making improvements based on recommendations from previous literature. In this 2021 paper, n=25 individuals were observed before and after a meal, and several significant correlations were identified between HRV measures and BGL.³⁸⁹ Kilojoule intake was controlled for in the statistical analysis to account for differences in meal composition, and the multiple linear regression analysis demonstrated that roughly 50% of the variance observed in postprandial BGL could be attributed to measures of HRV. This strongly suggests HRV measures may be clinically relevant in the prediction of BGL. To describe this in a real-world scenario, a simple 10-minute heart rate recording may be used to determine HRV measures, which in turn may provide an estimate or prediction of BGL around the same time. Combined with other routinely measured variables, such as age, body mass index, and blood pressure, HRV measures may reliably predict glycaemia. Future research should continue exploring the predictive quality of HRV measures, as well as the relationship between HRV and BGL.

The aims and hypotheses established in this research were based around a central thesis: that HRV measures can predict BGL, and thus HRV may be a viable non-invasive marker of BGL. This research may address some of the core concerns with current glucose monitoring. When developing these aims, the scope of a PhD candidature was considered as well as the specific needs of the main types of diabetes. Though T1D and T2D are similar in that they are

characterised by persistent hyperglycaemia related to insulin dysfunction, there are meaningful differences in their pathophysiology, epidemiology, treatment, and severity (Section 1.4). Importantly, CGM is more useful in T1D, and SMBG is a staple of T2D management. Considering this, two separate methodologies were designed for studies with distinct aims and hypothesis. The first of these was the UTS T2D Study, conducted on people with T2D. The second of these was conducted at Royal North Shore Hospital (RNSH) on people with T1D and is referred to in this thesis as the RNSH T1D Study. Prior to these two main studies, one initial, exploratory study was also conducted based on previous work published by Rothberg and colleagues (2016) and Jarman and colleagues (2021). This was known as the UTS Pilot Study. As such, this thesis is comprised of three separate studies, and the Methods, Results, and Discussion sections of each of these studies are compiled separately.

The UTS Pilot Study was a small exploratory study which investigated the relationship between HRV and BGL in people after they ate a meal and was conducted to provide preliminary data to assist with the design of larger subsequent studies. As such, it was also in part a feasibility study. The second study addressed a broader aim in people with T2D as well as people without and was known as the UTS T2D Study. This was launched after the UTS Pilot Study concluded, as the findings of this initial study, such as time to peak BGL, were used in the design of the UTS T2D Study. The importance of assessing BGL at specific time points and in specific metabolic states was discussed in Section 1.6.5. Additionally, it is recommended that people living with diabetes obtain glucose readings by SMBG before and/or after breakfast to monitor day-to-day variations in BGL data.³⁹⁹ Due to the threat of glycaemic variability and its prevalence in postprandial states, it is also relevant for a novel technology to predict the change in BGL following a meal, as this may allow for adjustments in medication to intervene.^{400, 401} In recognition of this, the methodology of the UTS T2D Study was designed around obtaining fasting and postprandial BGL, where appropriate. The third study focused on people with T1D. As discussed, people with T1D were studied separately to those with T2D due to differences in treatment and management of these distinct conditions, thus the need to dedicate a separate set of aims and studies for each of the main types of diabetes. This third study, the RNSH T1D Study, did not recruit non-diabetes participants for comparison, and reasons for this are provided in Section 4.1.1. The aims and hypotheses of these three studies are provided below, and this novel research may improve current standards in glucose monitoring and improve quality of care in diabetes.

1.12 Aims

- UTS Pilot Study (Chapter 2) Investigate changes in HRV measures and BGL over a twohour period in a small pilot study on people without any chronic illness or regular medications.
- 2. UTS T2D Study (Chapter 3) Investigate correlations between HRV measures and fasting and postprandial BGL in people with T2D and people without T2D.
- 3. RNSH T1D Study (Chapter 4) Investigate acute changes in HRV measures preceding and during hypoglycaemia and hyperglycaemia events in T1D.

1.13 Hypotheses

- 1. UTS Pilot Study (Chapter 2) HRV measures and BGL will change significantly between the start, middle, and end of a two-hour period.
- UTS T2D Study (Chapter 3) LF power, HF power, and total power will be significantly and negatively correlated with both fasting and postprandial BGL in both groups. LF:HF will be positively correlated with both fasting and postprandial BGL in both groups.
- RNSH T1D Study (Chapter 4) LF power, HF power, total power, LF:HF, normalised LF power, normalised HF power, RMSSD, and SDNN will increase significantly from euglycaemia to hypoglycaemia and from euglycaemia to hyperglycaemia.
2.1 Methods

The aim of the Pilot Study was to investigate changes in HRV measures and BGL over a short period of time. More specifically, the Pilot Study aimed to determine the magnitude of changes in HRV and BGL in response to food intake, an important foundation for subsequent research, and was also in part a feasibility study. The research was based on previous work conducted in an Honours year,³⁸⁹ in which n=25 individuals were assessed over a six-hour period. The Honours research presented novel findings on the acute associations between HRV measures and BGL across various timepoints over a six-hour period. In this, it was concluded that the timing of assessments was important in determining the strength of the association between HRV and BGL. As such, this was considered when developing the study design for the Pilot Study aimed to investigate whether changes in HRV and BGL would be significant across a shorter period of two hours, instead of six hours. This research is not novel, but it was an important precursor study, and the Methods are relevant to the second and third studies also conducted as part of this PhD.

2.1.1 Subjects

Participants were recruited from the Sydney Metropolitan region and were contacted about the study through local advertising, such as posters, social media, and word-of-mouth. Potential subjects were notified by email about the study protocol and exclusion criteria, and eligible participants were invited to attend the research laboratory at the University of Technology Sydney (UTS). The protocol was conducted with approval from the UTS Human Research Ethics Committee (HREC) (2014000110) and was developed to address the aims and hypotheses stated in Section 1.12 and Section 1.13, respectively. The sample consisted of individuals who were uninhibited by any chronic illness or regular medications, as verified by a questionnaire.^{402, 403}

2.1.2 Exclusion Criteria

Prior to study commencement, participants were informed of the exclusion criteria by email, and were excluded from the study if, at present, they: were living with a chronic health condition;⁴⁰³ were taking regular medication, prescribed or otherwise;⁴⁰² had an acute illness;¹⁴⁶ smoked more than 10 cigarettes per day or consumed more than 10 standard alcoholic beverages per day, as required by UTS HREC; or were currently pregnant.⁴⁰² Additional criteria were included to adhere to guidelines set by UTS HREC (see Section 2.1.3). Participants were also required to be between the ages of 18 and 69. Eligible subjects were invited to meet with the researcher at the UTS City Campus after a caloric restriction of eight hours.⁴⁰⁴ This meant that participants were required to abstain from food and drink (only water was permitted), as well as nicotine, medications, and alcohol for at least eight hours prior to study commencement. Participants who were unable to meet this restriction, such as if they needed to take pain killers for a headache, were to inform the researcher so that their appointment could be rescheduled for another day. As a final pre-study requirement, both the researcher and participant signed two copies of a consent form, and each party retained one copy.

2.1.3 Blood Pressure

Participants were screened for high blood pressure (BP) as per ethics protocol. BP was measured using an OMRON HEM-7000 (OMRON, Kyoto, Japan) automated BP monitor (Figure 2.1), after each participant had been allowed to rest in a seated position for five minutes, with a two-minute rest interval between each recording.¹⁴⁶ Due to the dynamic nature of BP, the mean of three left-arm BP recordings were used to provide better accuracy of the BP recording.⁴⁰⁵ If the average of these three BP readings was < 140 mmHg systolic and < 90 mmHg diastolic, the participant was included in the study. If systolic BP was between 140 - 160 mmHg, or diastolic BP was between 90 - 100 mmHg, the subject was included in the study, but they were also advised to inform their general practitioner of their blood pressure recordings, if not already aware. Subjects with a systolic BP > 160 mmHg and/or diastolic BP > 100 mmHg were excluded from the study and offered to be escorted to the nearest medical centre. These were based on HREC requirements.

Figure 2.1 Example of participant with an automated blood pressure monitor.



Figure 2.1 shows a participant with an OMRON HEM-7000 automated blood pressure monitor attached to their left arm by an inflatable cuff. Blood flow to the brachial artery was slowly constricted until arterial pressure was exceeded by the cuff pressure. This allowed the monitor to detect the maximum amount of force exerted by the heart on the brachial artery, which was an estimation of systolic blood pressure. In this case, it was given as 111 milligrams of mercury. The cuff then gradually deflated until the diastole of the cardiac cycle was detected, which is an estimation of diastolic blood pressure, or 70 milligrams of mercury. Heart rate is also provided by the monitor and was 80 beats per minute. Image used with consent from the participant.

2.1.4 Blood Glucose Assessment

Provided they were not excluded for high BP or for any other reason stated in Section 2.1.2, participants underwent an initial BGL assessment. Given that they had been asked to fast overnight or restrict their caloric intake to zero for at least eight hours prior, their metabolic state fulfilled the requirements for 'fasting'.⁴⁰⁶ Fasting BGL was an approximation of baseline level, as it was not confounded by carbohydrate or caloric intake, which may vary considerably between individuals depending on their diet. The fasting assessment was undertaken at roughly 9:00am for each participant, as a way of diminishing the effect of circadian rhythms. These rhythms are responsible for natural fluctuation of physiological functions over the course of the day, including those related to the management of glycaemia.⁴⁰⁷ As such, it was important to control for circadian rhythms by standardising the time at which each participant was assessed.

The ACCU-CHEK Performa II glucometer (Roche Diagnostics GmbH, Mannheim, Germany) was chosen for its accuracy to conduct the blood glucose assessment.⁴⁰⁸ The subject's dominant hand was cleaned first, and then the disposable lancets were used to puncture the skin on the tip of the finger to produce a small sample of blood.²⁰³ The blood glucose assessment is highly accurate, and capillary BGL assessed at the fingertip correlates strongly with arterial BGL.¹⁹⁹ An ACCU-CHEK Performa test strip (Roche Diagnostics GmbH, Manheim, Germany) was then applied to draw up the blood, and this was inserted into the glucometer to provide the measurement (Figure 2.2).

Figure 2.2 Example of a participant undergoing a blood glucose assessment.



Figure 2.2 demonstrates how BGL was measured in this study. **A)** The participant's finger was cleaned and prepared using 70% isopropyl alcohol swabs. **B)** The disposable lancet was used to puncture the skin on the side of the tip of the finger. **C)** The subject's fingertip was squeezed gently to produce a small drop of blood. **D)** The test strip, inserted into the glucometer, measured the sample of blood. Following these four steps, the glucometer gave a reading of BGL in millimoles per litre. Biohazard material and sharps were disposed of appropriately. Image used with consent from the participant.

2.1.5 Heart Rate Variability

This section outlines the multi-step process used to obtain HRV data during this study. There are many reputable methods for calculating HRV, though nearly all of them involve collecting R-R interval data as a common step, followed by some form of computer processing to extrapolate HRV measures from the R-R interval data. For the purposes of the UTS Pilot Study, the philosophy was to utilise gold standards and common techniques where possible. The use of highly recognised techniques may facilitate the integration of this novel technology with clinical systems. ECG is considered the gold standard for collecting R-R interval data for HRV analysis.^{409, 410} Though HRV can be recorded over 24-hours for additional clinical utility, it is important to develop this technology over short-term recordings, as this would be more convenient for diabetes screening and for SMBG. For example, a physician could perform a 10-minute ECG to predict diabetes risk, or a simple heart-rate monitor could be used by a person at home to predict current BGL. For this reason, 10-minute ECG recordings were suitable for the aims and scope of this PhD.

HRV measures may be expressed as units of time or frequency, and there are specific steps involved in extrapolating time-domain HRV measures compared to frequency-domain. Time-domain HRV directly reflects the variation in heart-rate over time, and is a measure of the intervals between successive regular cardiac cycles.⁴¹¹ There are multiple types of time-domain HRV measures, and these are generally calculated from longer recordings, traditionally 24 hours or longer.⁴¹² For these longer periods, more complex statistical time-domain measures can be extrapolated. For one time-domain HRV measure, this 24-hour recording provides the gold standard for stratification of cardiac morbidity and mortality risk.³⁵⁸ As only short-term recordings were used in the UTS Pilot Study, time-domain measures were not utilised. However, frequency-domain HRV measures are a set of highly reputable variables which are favoured in this area of study as they reflect specific types of autonomic modulation.^{383, 397} The specific neuroanatomical correlates of each frequency-domain measure will be outlined in this chapter.

2.1.5.1 Recording R-R Interval Data by Electrocardiogram

Subjects underwent a 10-minute ECG after resting in a seated position for 10 minutes to ensure their cardiac activity was not in an excited state. The ECG device was a three-lead FlexComp Infiniti (SA7550) encoder (Thought Technology Ltd., Montreal, Canada), which sampled electrical activity of the heart at a rate of 2,048 Hz.⁴¹³ Electrodes were placed according to the Einthoven configuration,⁴¹⁴ as shown in Figure 2.3. 10-minute recordings have better reproducibility and provide more accurate representations of autonomic activity in comparison

to other short-term HRV recordings, such as 5-minute or 2-minute recordings.⁴¹⁵ When the ECG was complete, the electrical waveform was downloaded from the device for further processing.

Figure 2.3 Einthoven electrode placement for the 10-minute electrocardiogram recording.



Figure 2.3 illustrates the reference positions for the Einthoven electrode placement. Electrical current flows from the negative electrodes (yellow) to the positive electrodes, as indicated by the arrows. The resulting potential is used to produce an ECG of the heart's electrical activity. Adapted from Combatalade (2010).

To derive the R-R intervals from the waveform, the 10-minute ECG waveform was processed using Kubios HRV Premium (ver. 3.3). This software applied a QRS detection algorithm to the waveform to detect the peaks in electrical activity apparent in each cardiac cycle. Each peak corresponded with a single heartbeat, which represents the main depolarisation of the heart as it contracts. On an ECG waveform, heartbeats are referred to as R-waves, and the distance in time between successive heartbeats is known as an R-R interval.⁴¹⁶ Once the R-R intervals for the 10-minute segment were determined, an artefact-detection algorithm was applied using Kubios HRV Premium to remove abnormal values or outliers. The automatic correction method

was used, which detected artefacts that were missed, extra or misaligned beat detections, as well as ectopic beats. For high-quality recordings, typically less than 1% of the R-waves detected will be flagged as artefacts, and they are usually related to the participant moving during the recording.⁴¹⁷ Finally, the R-R intervals were plotted on a time-series graph,³⁸² as shown in Figure 2.4. To clarify, the 'time-series' graph is not related to 'time-domain' HRV measures. As discussed, this study did not utilise time-domain measures. The time-series graph refers simply to how each R-R interval, measured in seconds, was plotted on a Y-axes with 'time of recording' on the X-axis.





Figure 2.4 shows how the time-series graph was derived. The top graph **(A)** represents the first four seconds from a raw ECG trace. Using beat-detection algorithms in Kubios HRV Premium, R-waves were flagged by red plus symbols. The time, in seconds, between each R-wave was calculated as R-R intervals. The time-series graph presented in **(B)** shows every R-R interval for the first 40 seconds of the same ECG recording. The green box in **(B)** displays the three R-R intervals (RR₁, RR₂, RR₃) that were observed in **(A)**. As pictured, RR₁, RR₂, and RR₃ were roughly the same length of about 0.9 seconds. **+** = R-wave, **ECG** = Electrocardiogram, **mV** = Millivolt, **RR** = R-R interval, **S** = Seconds.

2.1.5.2 Extrapolation of Frequency-Domain Measures

Once the time-series graph was prepared, the data was ready for a non-parametric fast Fourier transformation, a highly-recognised means of calculating frequency-domain HRV measures.^{146,}

^{313, 382} There are two common methods favoured by researchers in this area of study: autoregressive spectral analysis and fast Fourier transform. The measures extrapolated through these methods are not interchangeable, and investigations in people living with diabetes usually determine HRV measures by fast Fourier transform as the preferred method.⁴¹⁸ The time-series graph was converted to a power frequency spectrogram using Welch's method, which is a periodogram-based method to solve the discrete Fourier transform.^{313, 419} An example spectrogram, derived from a participant's ECG waveform that was processed, is shown in Figure 2.5. Within the spectrogram, three frequency bands were categorised: very low frequency (VLF) power, which ranged from 0.00-0.040 Hz on the spectrogram; low frequency (LF) power, from 0.04-0.15 Hz; and high frequency (HF) power, from 0.15-0.40 Hz.^{382, 420} The power of each of these bands, calculated from the area under the curve, is expressed in milliseconds squared (ms²). For example, LF power for a participant at baseline may be 900 ms². This power is correlated with both sympathetic and parasympathetic nervous system activity, and changes in the magnitude of this power reflect changes in autonomic control.



Figure 2.5 Frequency spectrogram derived from fast Fourier transform.

Figure 2.5 represents a frequency spectrogram obtained by fast Fourier transform. The power of each frequency is denoted by the area under the curve, and the units are in milliseconds squared. Low frequency typically has the greatest power, as shown in this example, though this has no noteworthy implications. The power of each measure is usually not compared with different measures, for example low frequency is not compared with high frequency. However, high frequency may be compared at post-intervention versus baseline to determine a change in autonomic activity, or compared between different sample groups to determine if autonomic function is different between them. Hz = Hertz, $s^2 = Seconds$ squared.

LF power is associated with sympathetic and parasympathetic nervous system activity, and individuals with higher levels of sympathetic activity, such as during stressful periods, would expect their LF power to increase above their baseline levels.⁴²¹ HF power correlates with parasympathetic nervous system activity and the activity of the vagus nerve, and thus predominates at rest.⁴²² Vagus nerve activity is sometimes referred to as vagal tone. Individuals would expect their HF power to be higher when at rest, and particularly during sleep.⁴²¹ The VLF band represents mostly background noise for short-term recordings, as 10-minutes of recording is not long enough per the Nyquist frequency, which is defined as half the average sampling frequency. For heart rate analysis, this is the mean heart rate. In addition to VLF, LF, and HF power, there were two other measures of HRV that were used in this study. The first of these was total power, which was calculated by measuring the area under the curve for the entire spectrogram from 0.00-0.40 Hz. As such, it can be thought of as the sum of VLF, LF, and HF, as the resulting power is identical to total power. The second of these was the low to high frequency ratio (LF:HF), which is calculated by dividing LF power by HF power. LF:HF represents sympathovagal balance, and increases in LF:HF over time indicate an increase in sympathetic activity relative to parasympathetic activity.⁴²¹ Due to short-term VLF power being largely composed of meaningless background noise, it was not used as an individual measure of HRV. As such, the four HRV measures were LF power, HF power, LF:HF, and total power. Of these, LF power, HF power, and total power were natural logarithm transformed as their distribution was highly skewed, though this is typical of HRV studies.^{146, 423, 424} For a summary of the steps in Section 2.1.5.1 and 2.1.5.2, see Figure 2.6.



Figure 2.6 Flow chart for extrapolation of heart rate variability from electrocardiogram data.

Figure 2.6 displays the step-by-step process by which HRV measures were obtained. The original ECG waveform was recorded over a 10-minute period. This data was downloaded from the device and then the following actions were performed using Kubios HRV Premium. The software was used to apply a QRS detection algorithm and an artefact removal algorithm to prepare a corrected waveform with each R-wave marked with a red plus symbol. A manual check was conducted to ensure that the algorithms had flagged each R-wave correctly, and then the R-R intervals were plotted on a time-series graph. Finally, the frequency spectrogram was produced by fast Fourier transform. The area under the curve for each frequency was measured, and these corresponded with each HRV measure, for example low frequency power and high frequency power. ***QRS** refers to the QRS complex, the central peak that can be observed on any regular ECG, and which correlates with ventricular depolarisation. **ECG** = Electrocardiogram, **HRV** = Heart rate variability.

2.1.6 Follow-up

As described previously, participants underwent a fasting BGL assessment and ECG after a caloric restriction of at least eight hours. This assessment can be referred to as the first or the 9:00am assessment. To satisfy the aims of this study, which intended to record changes in HRV

and BGL across a two-hour period, participants were asked to return to the laboratory for additional assessments. Given that the first assessment occurred at roughly 9:00am, the second assessment was scheduled for 10:00am and the third assessment was at 11:00am. These three times were agreeable with all participants. The selection of these times was arbitrary – the priority was to ensure that participants were assessed at roughly similar times of the day to remove the confounding effect of circadian rhythms. This methodology was expanded through future research.

2.1.6.1 Meal and Additional Assessments

As discussed in Section 1.6.4, BGL remains highly stable in the absence of caloric intake or certain drugs. Without some form of intervention, the BGL of participants in this study would have fluctuated very little from fasting levels, and there would have been few noteworthy observations over a two-hour period. As such, once fasting BGL was recorded in participants at 9:00am, they were asked to leave the laboratory and enjoy a meal of their choice. This allowed for the second assessment at 10:00am to capture an important change in BGL – that which naturally occurs after consuming a meal. Changes in HRV measures during this time were also observed. Though participants were expected to skip breakfast until after the study commenced, which may be considered an irregular meal schedule, research has shown that day-to-day changes in BGL are not significantly affected by consuming meals at irregular times.⁴²⁵

For the 10:00am assessment, participants underwent a second blood glucose assessment and a second 10-minute ECG recording. They were then instructed to leave the laboratory and return at 11:00am for the third blood glucose assessment and third ECG, and they were not permitted to eat during this break. For each of these follow-up assessments, BGL and HRV were assessed as described in Section 2.1.4 and 2.1.5, which is to say that the methodology remained constant regardless of what time of day the assessment related to. The main significant difference between the 9:00am, 10:00am, and 11:00am assessments was the metabolic state of participants, as this changed over the course of the day and with the introduction of caloric intake. However, all participants were required to undergo all three assessments at the specified times, as well as follow the instructions related to the meal.

2.1.6.2 Food Diary

To reduce participation burden, subjects were free to consume a regular meal between the 9:00am and 10:00am assessment, rather than undergo a standardised glucose load such as the

oral glucose tolerance test, which may cause nausea.⁴²⁶ Research suggests that glucose peaks observed after a 2-hour glucose tolerance test are similar to glucose peaks that occur after a regular, standardized meal.⁴²⁷ However, standardized meals do not reflect realistic daily routines, and this study aimed to observe fluctuations in HRV and BGL under loosely-routine circumstances. Therefore, participants were free to eat a meal of their own choosing. To account for differences in the composition of subject's chosen meals, which may have confounded their glucose profiles,⁴⁰² participants were instructed to report on all food and drink consumed between the 9:00am assessment and the 10:00am assessment. This quantitative information was converted by the researcher into kilojoule data, such that kilojoule intake could be used as a covariate in the statistical analysis.

The MyFitnessPal[™] calorie counter search tool was used to estimate nutritional values for food and drink products consumed by participants in this study. Commonly, subjects consumed products from popular brands and chain restaurants, and accurate kilojoule data for these products was available from the MyFitnessPal[™] website. For all other food items, including homemade products, kilojoule data was estimated using values uploaded by users of the MyFitnessPal[™] website. Certain types of food are known to confound postprandial BGL; for example, some food increases BGL disproportionate to the serving of kilojoules in the food. Participants were asked to avoid any of the following common foods known to exaggerate or diminish glycaemia: cinnamon,⁴²⁸ vinegar,⁴²⁹ green tea,⁴³⁰ and black tea.⁴³¹ Otherwise, subjects were not restricted in what or how much they could eat or drink, only that they had to return to the laboratory for the follow-up assessment at 10:00am (Figure 2.7). Figure 2.7 Protocol summary for the UTS Pilot Study.

Pre-study criteria	Contact potential participantsEligible participants invited to laboratory			
8:45am	 Sign consent forms Lifestyle Appraisal Questionnaire Three blood pressure measurements 			
9:00am	 Blood glucose assessment 10-minute electrocardiogram 			
Break (to eat)				
10:00am	 Complete Food Diary Blood glucose assessment 10-minute electrocardiogram 			
Break (no eating)				
11:00am	 Blood glucose assessment 10-minute electrocardiogram 			
Post-study criteria	• Three blood pressure measurements			

Figure 2.7 outlines the step-by-step process for each participant during the Pilot Study. Times given are approximate and were only used as rough guidelines for each participant. When subjects were contacted about the study, they were also informed of the study protocol and the exclusion criteria. Before the study could commence, participants had to refrain from eating or drinking (water was allowed) for at least eight hours prior. Drugs, including medications, nicotine, and alcohol, were also prohibited for this period. For the blood pressure recordings and ECG, participants were allowed to rest in a seated position for at least 10 minutes, to allow them to be in a resting state.

2.1.7 Statistical Analysis

Statistical analyses were conducted using SPSS version 22.0 (IBM SPSS Statistics, USA). Pairedsample t-tests (two-tailed) were used to identify whether BGL or HRV measures were significantly different between the 9:00am, 10:00am, and 11:00am assessments. These HRV measures were LF power, HF power, LF:HF, and total power. Statistical significance was set at p < 0.05. "A Power Primer" by J. Cohens was consulted to determine a sample size (n) which would provide sufficient power for the study. For a comparison of means between 3 timepoints in a single group and a significance set to 0.05 and a large effect size of 0.5, a sample size of n=30 would provide a power level of 0.8, or an 80% probability of rejecting a false null hypothesis at the 5% significance level. As such, the recruitment goal was at least 30 people.

2.2 Results

The goal of the UTS Pilot Study was to recruit n=30 subjects and investigate changes in HRV measures and BGL between the start, middle, and end of a two-hour period. Concerns were raised during recruitment that the ECG data collected from participants was unfit for HRV analysis due to faulty equipment. Efforts were made to resolve the issue and to continue recruitment, however the study was eventually terminated early with n=14 participants assessed, which did not meet the sample power requirements. The total data set was 42 data points (n=14 participants with three assessments each), however, 17 of the 42 ECG recordings were unusable due to considerable levels of noise (Figure 2.8), and some of those that were usable had to be cut and shortened to reduce the effects of noise and artefacts. In terms of demographics, the age range of the sample was 19-55 years with a mean age of 28 ± 11 years. The mean body mass index was 24 ± 4 kg/m² with a sex breakdown of 50% male and 50% female. The mean kilojoule intake after the fasting assessment was 3700 ± 1700 kilojoules.



Figure 2.8 Effect of electrocardiogram noise on extrapolating heart rate variability measures.

Figure 2.8 illustrates how poor ECG data can affect the determination of HRV measures. **(A)** A high-quality ECG trace recorded from a subject, with the R-waves (+) correctly identified by the software. **(B)** A low-quality ECG recorded from a subject that was age and sex-matched with the subject with the high-quality ECG. Note how, due to the 'noisy' ECG, it was difficult for the beat-detection software to identify the R-waves (+) which corresponded with each heartbeat. In this case, the noise was created from faulty equipment. **(C)** The full 10-minute recording of the high-quality ECG after it was processed into a power frequency spectrogram. **(D)** The low-quality ECG resulted in exaggerated high frequency power (green) and reduced low frequency power (pink) after processing. **ECG** = Electrocardiogram, **Hz** = Hertz, **mV** = Millivolt, **PSD** = Power spectral density, **RR** = R-R interval, **s**² = Second squared.

2.2.1 Glucose Level and Blood Pressure

Mean values for BGL and blood pressure recorded at each assessment are presented in Table 2.1. Paired sample t-tests (two-tailed) were used to determine whether these physiological variables changed significantly between each time point. BGL recorded at 9:00am was relatively uniform as participants were in a fasting state, and this was reflected in the relatively small standard deviation compared to BGL recorded in later assessments (Table 2.1). The increase in BGL from 9:00am to 10:00am was significant (p<0.01), as was the overall change in BGL from 9:00am to 11:00am (p<0.01) Though BGL appeared to decrease from 10:00am to 11:00am, the change was not significant. Other physiological variables assessed included heart rate, and systolic and diastolic blood pressure. There was no significant change in any of these three variables between 9:00am and 11:00am, the two points at which these were assessed. Figure

2.9 compares the means for each of these four variables at the three time points assessed and includes a visual representation of standard deviations.

Table 2.1 Blood glucose and blood pressure means and standard deviations recorded in the UTS Pilot Study sample (n=14). Blood glucose measurements were taken at the start, middle, and end of a two-hour period, at 9:00am, 10:00am, and 11:00am, respectively. Blood pressure and heart rate were recorded only at the start and conclusion of the study as per ethics requirement, and so data was provided for the 9:00am and 11:00am assessments only. ****** = p<0.01, **BGL** = Blood glucose level, **BP** = Blood pressure, **bpm** = Beats per minute, **mmHg** = Millimetres of mercury, **mmol/L** = Millimoles per litre.

					р	
Variable	9:00am	10:00am	11:00am	9:00am vs. 10:00am	10:00am vs. 11:00am	9:00am vs. 11:00am
BGL (mmol/L)	4.8 ± 0.3	6.1 ± 0.6	5.6 ± 0.8	<0.01**	0.32	<0.01**
Heart rate (bpm)	71 ± 13	-	70 ± 13	-	-	0.35
Systolic BP (mmHg)	114 ± 11	-	112 ± 9	-	-	0.42
Diastolic BP (mmHg)	73 ± 11	-	73 ± 8	-	-	0.72



Figure 2.9 Blood glucose, blood pressure, and heart rate data for the three assessments (n=14).

Figure 2.9 displays the means for the physiological variables that were assessed at each time point. Error bars represent the standard deviation for means of each variable. **(A)** The low to high frequency ratio showed large standard deviations across the 9:00am and 11:00am assessments and demonstrated no significant change. **(B)** Low frequency power was significantly different from the 9:00am assessment compared to 11:00am. **(C)** High frequency power had moderate standard deviations across all time points, and at 11:00am was significantly lower compared to 9:00am. **(D)** Total power decreased in the 11:00am assessment compared to 9:00am. ** = p<0.01, **mmHg** = Millimetres of mercury, **mmol/L** = Millimoles per litre, **ms**² = Milliseconds squared.

2.2.2 Heart Rate Variability Measures

ECG recordings were obtained in 10-minute epochs at each of the three assessments. In turn, these were transformed into HRV measures, reflecting autonomic activity at each of the three assessments. LF power, HF power, LF:HF, and total power were the HRV measures of interest in this study, and mean values are provided in Table 2.2. Note that the values provided here are natural log-transformed – raw HRV values are rarely presented in the literature as they are highly skewed. The only significant changes in these HRV measures were between 9:00am and 11:00am, as LF power (p<0.001), HF power (p<0.05), and total power (p<0.01) were all significantly lower in the 11:00am assessment compared to the 9:00am assessment. For a visual reference and comparison of these means and standard deviations, see Figure 2.10.

Table 2.2 Heart rate variability measures means and standard deviations recorded in the UTS Pilot Study sample (n=14). LF:HF is a ratio calculated by dividing LF power by HF power, and thus has no units. * = p<0.05, ** = p<0.01, *** = p<0.001, HF = High frequency, LF = Low frequency, LF:HF = low to high frequency ratio, ms^2 = Milliseconds squared.

					p-value	
Variable	9:00am	10:00am	11:00am	9:00am vs. 10:00am	10:00am vs. 11:00am	9:00am vs. 11:00am
LF power (ms ²)	7.0 ± 0.9	7.0 ± 0.7	6.4 ± 0.9	0.80	0.60	<0.01***
HF power (ms ²)	5.8 ± 1.3	6.0 ± 1.1	5.0 ± 1.3	0.49	0.90	<0.05*
Total power (ms²)	7.5 ± 1.0	7.4 ± 0.9	6.8 ± 0.9	0.94	0.69	<0.01**
LF:HF	4.8 ± 4.1	2.8 ± 1.7	5.3 ± 4.1	0.72	0.53	0.94



Figure 2.10 Change in the mean blood glucose level and mean low frequency, high frequency, and total power between the three time points (n=14).

Figure 2.10 displays the means for the physiological variables that were assessed at each time point. Error bars represent the standard deviation for each mean. LF power, HF power, and total power were natural log transformed. **(A)** The low to high frequency ratio showed large standard deviations across the 9:00am and 11:00am assessments and demonstrated no significant change. **(B)** Low frequency power was significantly different from the 9:00am assessment compared to 11:00am. **(C)** High frequency power had moderate standard deviations across all time points, and at 11:00am was significantly lower compared to 9:00am. **(D)** Total power decreased in the 11:00am assessment compared to 9:00am. **ms**² = Milliseconds squared.

2.2.3 Summary of Results

Though the recruitment target was not met in this pilot study due to technical issues, statistical analysis was conducted on the data set to provide information for subsequent research. In this sample, BGL – as assessed by invasive blood glucose assessment – rose significantly (p<0.01) after the ingestion of a meal by an average of 1.3 mmol/L. BGL was also slow to return to baseline or fasting levels, as the difference between BGL from 9:00am to 11:00am was also significant (p<0.01). There was no change in heart rate or blood pressure between the start and end of the study, though certain HRV measures changed over the course of the two hours. LF power, HF power, and total power were all significantly lower at the end of the study (11:00am) compared to the start of the study (9:00am).

2.3 Discussion

The ability to estimate BGL in certain metabolic states, such as before and after meals, without using invasive measures represents an important milestone in diabetes research. A non-invasive marker of fasting or postprandial BGL may address the limitations of current glucose monitoring technologies. In this study, frequency-domain HRV measures and blood glucose concentrations were monitored in a small sample to determine how they fluctuate over a short period of time. Specifically, the aim was to determine the magnitude of changes in HRV and BGL in response to acute food intake. This study was conducted to obtain pilot data and to address a broader set of aims which relate to how HRV measures may be capable of predicting BGL in adults. Though the quality of data collected in this study was poor, there was meaningful information gathered which was implemented in other components of this PhD candidature. For example, this study investigated the dynamics of HRV measures and BGL from baseline to postmeal, and this information was considered when designing future studies. These will be expanded on in later chapters. The following sections will discuss results of the present study and present them in the context of the wider literature.

2.3.1 Dynamics of Heart Rate Variability

The values of the HRV measures observed in this sample of 14 participants were within expected ranges for subjects with no chronic illness or regular medications.⁴³² However, as with many physiological variables, there was considerable variation in the data set as indicated by the standard deviation values (Figure 2.10). This was expected, though it is important that the reasons for this variability be explored. Variability in HRV data may be attributed to several factors, such as sex, age, and general health of individuals.⁴³³ Research by Yamasaki and colleagues (1996) showed that LF power peaks between 8:00am and 12:00pm in males and 12:00pm and 0:00am in females. To clarify, the average male would expect their LF power to be highest in the morning and the average female would expect their LF power to be highest any time between midday and midnight. Therefore, variability in HRV measures between males and females may be related to the timing of assessments. In this study, all participants were assessed at 9:00am, 10:00am, and 11:00am to attempt to control for circadian effects on HRV measures. However, circadian rhythms may be difficult to control for stringently as the literature suggests daily fluctuations in HRV measures are different between males and females. Future studies may consider normalizing the data by subtracting each follow-up assessment from the baseline. This may also control for the differences in baseline HRV between males and females. Additionally, though the mean age of the sample was relatively young (28 ± 11 years), the presence of some older participants – as old as 55 years of age – may have contributed to the variability of the HRV data observed in the sample. This is because all measures of HRV decline with age.⁴³⁴

Interpretation of the findings of the present study requires an understanding of the different measures of HRV and their neurophysiological correlates. To reiterate, LF power is correlated with both sympathetic and parasympathetic nervous system activity, HF power is correlated with parasympathetic nervous system, total power represents total autonomic activity, and LF:HF is an index of sympathovagal balance, whereby increases in LF:HF indicate a shift toward sympathetic predominance. There is no specific way of determining sympathetic activity from HRV. In this study, mean HRV measures, including LF power, HF power, and total power, were all significantly lower at the end of the study (11:00am) compared to baseline levels (9:00am). When discussing these findings, it also important to consider that LF:HF did not change significantly, which indicates that there was no meaningful shift toward sympathetic or parasympathetic dominance. With this information, we can conclude that mean sympathetic and parasympathetic tone were reduced at 11:00am compared to 9:00am, as both LF power and HF power were significantly lower at 11:00am. This explanation is consistent with the findings that total power was also significantly lower at 11:00am, and that LF:HF did not change significantly. Whether this decline in LF power and HF power was related to the consumption of the meal or the change in time of day – which would likely be related to circadian fluctuations in HRV – is not a conclusion that can be made based on the data from this study. By chance, there were relatively more ECG recordings from the 10:00am assessments which were unusable compared to the 9:00am and 11:00am, and thus fewer data points were included in the statistical analysis. This contributed to a smaller standard deviation for LF:HF at the 10:00am assessment, and may have also contributed to LF:HF being misaligned from the values of LF power and HF power at 10:00am.

With reference to the literature, Klimontov, Myakina & Tyan (2016) measured morning fasting HRV as well as afternoon postprandial HRV in n=67 women with T2D and found LF power was significantly lower in the afternoon compared to the morning, and HF power was not significantly different.³⁹⁶ In their paper, the authors argued that the reduction in LF power was due to either the consumption of the meal or the associated rise in BGL, but could not specify which of these was more important based on their analyses. Differences in the findings between Klimontov, Myakina & Tyan (2016) and the UTS Pilot Study may be due to differences in the sample groups, as women with T2D are physiologically distinct to people without diabetes in terms of autonomic activity and glucose management. In a study comparing fasting measures with postprandial measures Lu and colleagues (1999) showed that LF power does not change

after a meal, and HF power increases significantly.⁴³⁵ In a sample of n=30 young subjects (mean age 21 ± 4 years), Soni and colleagues (2017) showed there was an increase in sympathetic activity and a decrease in parasympathetic activity two hours after a meal.⁴³⁶ In this study, participants did not eat at exactly 9:00am, so though the study period was two-hours long, it was not technically two-hours since meal, so a comparison with the work of Soni and colleagues (2017) is not ideal. More recent literature agrees that sympathetic tone, as measured partly by LF power, increases after a meal, however blocking of these sympathetic pathways does not affect postprandial effects.⁴³⁷ This may be because sympathetic activity during absorptive phase of digestion may not be related to postprandial BGL.

Three studies in this area of study on groups of people without diabetes provide conflicting results. In their study of n=68 adults (66% male; mean age 26 ± 4 years), Choi, Choi & Kim (2008) found that HF power and total power were significantly lower in the afternoon (1:30pm to 4:00pm) compared to the morning (8:30am to 11:00am), whilst LF power and LF:HF were unchanged over this time period.⁴³⁸ However, Armstrong and colleagues (2011) reported no change in HF power from the morning to the afternoon in n=12 young adults aged 19-25.⁴³⁹ Lipsitz and colleagues (1993) observed an increase in mean LF power after n=11 young adults (mean age 26 ± 5 years) consumed a mixed meal. Due to the lack of change in plasma norepinephrine, they concluded that this increase in LF power represented cardiac sympathoexcitation. Discrepancies in these findings may be due to differences in the age of samples between the studies, as well as sample sizes.

For LF:HF, two studies on people without diabetes or any other chronic illness demonstrated consistent results. Paolisso and colleagues (1997) showed that ingestion of glucose in subjects was associated with an increase in LF:HF, indicating an increase in sympathetic nervous system activity.⁴⁴⁰ Tentolouris and colleagues (2003) demonstrated that LF:HF increased significantly in lean women (n=15, mean age 35 ± 11 years) in response to carbohydrate ingestion.⁴⁴¹ However, this study did not assess other traditional frequency-domain measures, including LF power, HF power, or total power. It is important to provide context for changes in sympathovagal balance, as indicated by LF:HF. As demonstrated in the present study, changes in LF power and HF power were shown to be consistent, as demonstrated by the lack of change in LF:HF and the decrease in total power. Though this is not a conclusive understanding of the HRV dynamics involved, as there are other factors which may have been involved, this study successfully demonstrated acute fluctuations in HRV measures in response to food intake. As outlined in Chapter 1, this is a novel area of research. It was not the aim of the UTS Pilot Study to conduct rigorous research and consolidate the current literature in this area, but rather to obtain real-world pilot data to use in the development of more rigorous studies.

2.3.2 Dynamics of Postprandial Blood Glucose

In this pilot study, mean BGL of the sample increased significantly (p<0.01) from 9:00am to 10:00am by an average of 1.3 mmol/L after the consumption of a regular meal. Though an increase in BGL was expected, the magnitude of this increase was difficult to anticipate. In a postprandial state, shortly after an individual has consumed food, glucose levels in the blood rise rapidly as nutrients are broken down in the gastrointestinal tract and absorbed into the blood. There is also an associated rise in circulating insulin, which aims to reduce the glycaemic spike following the meal,⁴⁴² though insulin levels were not assessed in this study. Most participants in this study finished their meal at 9:15am, and therefore the blood glucose assessment was performed roughly 45 minutes after the meal. Generally, the timing of this assessment should have captured the peak BGL of participants post-meal. However, the time required for glucose levels to peak after a meal, as well as the magnitude of that peak, are affected by a range of variables, including the composition of the meal and the health of the individual. These variables will be discussed in this section to determine if the results of the UTS Pilot Study are consistent with previous literature.

For meals composed entirely of carbohydrates, BGL increases rapidly for the first 30 minutes after consumption until it peaks, and then steadily declines until BGL returns to baseline levels 120-180 minutes after consumption.⁴⁴³ However, according to Brand-Miller and colleagues (2009), this is not just true for carbohydrates. Analysis of a database of over 1000 foods showed that peak BGL occurs after 30 minutes for most meals consisting solely of one type of food, including milk, yoghurt, legumes, vegetables, soft drinks, or fruit.444 This is generally irrespective of the glycaemic index of the food, for example white wheat bread has a higher index compared to rye bran bread, but both foods produce peak BGL 30 minutes after consumption.⁴⁴³ The glycaemic index of a food is more accurately used to estimate the total change in BGL from fasting to postprandial that can be expected from eating the food, rather than the time taken to achieve that peak BGL.⁴⁴⁵ This change is known as the incremental glucose peak, which can be obtained by subtracting the peak BGL after a meal from the baseline, or the fasting BGL. Esposito and colleagues (2008) contend that incremental glucose peak is a better marker of certain cardiovascular complications of diabetes compared to traditional markers, such as HbA_{1c} and fasting BGL.⁴⁴⁶ This is because incremental glucose peak is a general measure of glucose variability, and higher levels of variability are associated with long-term cardiovascular disease. Incremental glucose peak is higher in homogenous meals, for example a meal like yoghurt which is composed entirely of dairy products, which means BGL will increase rapidly to a peak within 30 minutes and then decline relatively quickly.

For meals consisting of more than one type of food product, which represents most meals and thus are referred to as 'regular meals', peak BGL is achieved after 40-45 minutes.⁴⁴⁷ Other studies, such as one conducted by Elizondo-Montemayor and colleagues (2015), show that BGL peaks between 30-45 minutes post-meal in subjects without diabetes (n=38),⁴⁴⁸ though differences in methodology may contribute to these slight differences in timing. Based on this evidence, meals consisting of multiple food types take longer to produce peak BGL. This is because of the 'food matrix' phenomenon, which effectively slows the rate at which nutrients are absorbed across the gastrointestinal tract, such that glucose levels in the blood do not rise as quickly. This results in a 'flattening' of the glucose curve. The matrix effect is greater in mixed meals, for example mashed potatoes eaten alongside high-protein foods such as chicken breast or salad produce a lower but more sustained glycaemic response compared to mashed potatoes eaten on their own.⁴⁴⁹ Adding different food types to a meal reduces the glycaemic response, which is a desirable outcome when considering that larger glucose fluctuations over time are associated with complications of cardiovascular disease, such as aortic stiffness.⁴⁵⁰ Additionally, there are short-term benefits to glucose regulation. A study conducted by Nilsson and colleagues (2012) highlighted that low, but sustained BGL profiles produced by low glycaemic index carbohydrates enhance cognitive function.⁴⁵¹

In summary, regular meals cause low, sustained increases in BGL. This low increase in BGL was observed in the UTS Pilot Study, recorded as a mean 1.3 mmol/L increase from 9:00am to 10:00am. Of note, there was no significant change observed in sympathetic or parasympathetic tone between 9:00am and 10:00am. The results of this study do not indicate a real-time relationship between HRV and BGL, as BGL increased without a measurable response in HRV measures. The change in HRV was delayed, indicated a delayed relationship. The lack of findings here may be due to the poor quality of ECG data. A more rigorous study design and statistical analysis with high quality data should follow-up this study to confirm. Also in this study, kilojoule intake was identified as a useful measure in this area of research, as it predicts the total glucose response, such as the magnitude of the BGL peak and the time for which postprandial BGL is sustained.⁴⁴⁵ The glycaemic index of the meal and kilojoule value determines the height of the glycaemic response when it reaches its peak. This information, including the timing of assessments and the magnitude of the changes in BGL observed between assessments, was used to develop future studies that were also conducted as part of this PhD candidate. These will be described in later chapters.

2.3.3 Strengths and Limitations

The main purpose of this study was to provide early pilot data to assist with the design of future studies. To that end, a strength of this research was that it condensed and focused previous research, which was conducted in an Honours year, whilst allowing for preliminary analyses to be conducted on HRV measures and BGL. The six-hour methodology utilised in the Honours research was identified to be excessive, and the UTS Pilot Study successfully implemented a shorter study design of two hours whilst maintaining the same number of assessments, at 9:00am, 10:00am, and 11:00am, instead of 9:00am, 12:00pm, and 3:00pm. The 'trial-and-error' approach that was taken with this pilot study, as a prelude to larger, more robust experiments, allowed for significant limitations to be identified so they could be accounted for in the future studies. Though the poor quality of the data reduced the credibility of the results, there were several meaningful limitations identified in this study which were improved in larger, future studies.

Firstly, the use of a stationary ECG instead of a portable ECG was not ideal for this specific area of research. The ECG did not allow for long-term monitoring of HRV, as it required participants to be seated, and movement artefacts were common. This also made it difficult to observe HRV in participants whilst they ate – as participants would have been required to be seated for the entire two-hours of the study. Recording autonomic data during mealtime may have yielded noteworthy data. Future studies should consider using a Holter monitor, which is a smaller, lightweight ECG, or even a simple heart rate monitor to continuously record HRV in participants for longer periods of time. This would allow participants to leave the laboratory and eat a regular meal of their choosing whilst being continuously monitored.

Secondly, the timing of blood glucose assessments was not an accurate reflection of specific metabolic states in all cases. For the 9:00am assessment, the blood glucose assessment was an accurate representation of fasting or baseline BGL. This is because participants were required to abstain from caloric intake for eight hours prior to the 9:00am assessment. However, the second assessment at 10:00am did not necessarily equate with postprandial BGL. True postprandial BGL reflects the peak glucose level in a participant after a meal, which occurs 40-50 minutes after most regular meals. The second assessment was conducted at 10:00am for all participants, and this standardisation was to control for circadian rhythms, however this may have not been a stringent method of control given that HRV circadian rhythms are different between males and females. Due to differences in how long it took participants to leave the laboratory and finish their meal, this 10:00am assessment was not a true reflection of postprandial BGL. As discussed in Section 1.6.5, there is a strong consensus in the literature that postprandial BGL is a

better predictor of diabetes complications such as cardiovascular disease.^{191, 192, 195} Therefore, future studies of a longitudinal design should aim to assess both fasting and postprandial BGL, as the ability to predict BGL in their metabolic states may provide meaningful clinical data.

Another limitation of this study, as well as the following experiments conducted as part of this PhD, is that the exclusion criteria for daily alcoholic beverage consumption of no more than 10 standard drinks per day was substantially higher than the regular daily intake of 2 standard drinks per day. This exclusion criteria was based on requirements set by the UTS HREC, rather than based on examples set in the literature. This is a limitation since excessive alcohol consumption has been associated with autonomic neuropathy. Additionally, participants were free to leave the UTS laboratory between recordings, however this may have increased metabolic demand and reduced glycaemia, as well as caused changes in HRV. Future research should consider restricting the activity of participants whilst they are being monitored.

There were several other limitations of the UTS Pilot Study which were improved on in the later studies conducted as part of this PhD. As an exploratory study, the small sample size was a major weakness of the research, and the lack of a clinical sample, such as a diabetes sample, limited the clinical relevance of the findings. As the overall aim of this PhD candidature was to justify future research into the clinical applications of HRV measures in predicting BGL, it was important that future studies improved on the sample size and included clinical samples. Future studies should base their power calculations on the expected correlation coefficient (r-value) between the two variables of interest, which in this area of study would be BGL (e.g. fasting) and each HRV measure (e.g. total power). Improving on the sample size, as well as the overall data set, would allow for predictive models to be developed, which may better demonstrate the capabilities of HRV measures in estimating BGL.

3.1 Methods

Due to technological issues with data collection in the UTS Pilot Study, the UTS T2D Study attempted to expand on the aims and hypotheses developed in the Pilot Study through a more comprehensive study. In short, the Pilot Study investigated changes in HRV measures and BGL in n=14 participants over a two-hour period. The T2D Study aimed to explore the applications of HRV measures as non-invasive markers of BGL in people with T2D, who may benefit greatly from a non-invasive alternative to current SMBG methods. This was a core component of the thesis. Primarily, the aim was to achieve this by correlation analysis and by multiple regression analysis. It was hypothesised that the physiological association between HRV measures and BGL was strong enough to justify the clinical applications of HRV measures as a non-invasive estimate of glycaemia. It was also important to investigate whether HRV may have applications for people with optimal levels of blood glucose. Thus, the UTS T2D Study recruited a sample of people with T2D and a sample of people without any chronic illness. Associations between HRV measures and BGL were determined in different metabolic states, including fasting and postprandial, as the importance of these has been discussed.

Though the literature agrees that CGM is superior to intermittent glucose monitoring for glycaemic management in T2D, CGM systems are not indicated for T2D. There is insufficient evidence to suggest that the introduction of a reliable non-invasive method of CGM would replace intermittent glucose monitoring in T2D, as efficiency of current SMBG is already low. There is, however, strong evidence to support that introducing a non-invasive method of intermittent glucose monitoring would improve efficacy of SMBG, (Chapter 1). As such, the UTS T2D Study investigated whether HRV measures could be used to predict BGL measured by intermittent forms of SMBG, such as the 'finger spot' check most-commonly used in T2D. HRV measures may offer a non-invasive alternative to current SMBG techniques, which are universally invasive, and thus improve SMBG effectiveness.

3.1.1 Subjects

As with the Pilot Study, subjects were uninhibited by any chronic illness or regular medications, as verified by a questionnaire. Participants were deemed to be appropriate for the T2D group if they were currently living with T2D at the time of commencing the study and had been

diagnosed with the condition at least six months prior by a qualified medical practitioner. Subjects were recruited by word-of-mouth and social media, and eligible participants were invited to the researcher's laboratory at UTS to complete the protocol as approved by UTS HREC (approval number: 2014000110, then updated to ETH19-3676). In addition to standard recruitment procedures, this study was endorsed by Diabetes Australia, and an advertisement was featured on the "Take Part: Current research opportunities" section of their website.

3.1.2 Pre-study Requirements

For a detailed description of the pre-study requirements for subjects, please refer to Section 2.1.2, as they are the same as the UTS Pilot Study. In this study, requirements for participants in the T2D group were adjusted for the sake of practicality; for example, given the high comorbidity of T2D with other lifestyle-related diseases, particularly cardiovascular diseases such as hypertension, it was not practical to exclude participants in the T2D group if they had any other chronic illnesses or if they were taking any medications. However, participants were excluded if they had any history of cardiac autonomic neuropathy or if they were taking medications which affected cardiac function, such as β-blockers or tricyclic drugs.^{452, 453} As with the subjects without T2D, people in the T2D sample group were required to be aged between 18-69 years and were ineligible to participate if they were currently pregnant, smoked more than 10 cigarettes per day, consumed more than 10 standard alcoholic beverages per day, or if their blood pressure (BP) was too high during the pre or post-study BP recordings (Section 2.1.3). This information was collected in the Lifestyle Appraisal Questionnaire (LAQ),⁴⁵⁴ which provided a score of a person's lifestyle risk factors as well as a score of their perceived stress over the previous eight weeks. Subjects in the T2D group were asked to fast for at least eight hours, as with the group without T2D, but they were to continue their medications as best they could and prioritise any medical advice given to them by their doctor or diabetes educator.

3.1.3 Expansion of Previous Methodology

As with the Pilot Study, participants attended the UTS laboratory after an overnight fast or a caloric restriction of at least eight hours. They were required to sign consent forms and undergo pre-study blood pressure measurements, as described in Section 2.1.3. Provided they were not excluded by this point, they were then subject to a blood glucose assessment as described in Section 2.1.4. In the Pilot Study, issues with data collection stemmed from faulty ECG

equipment, and so this study expanded upon this original methodology by using alternative means of recording HRV.

3.1.4 Recording R-R Interval Data by Heart Rate Monitor

Though ECG is the gold-standard for determining HRV, it is not the only reputable method. ECG devices are generally expensive, bulky, and require many electrodes to be attached to the host. This may not represent an appealing alternative to current SMBG methods, even though it is non-invasive. The T2D Study investigated the potential of HRV as a non-invasive alternative to invasive procedures, and thus priority was given to lightweight, portable devices when selecting a device capable of determining HRV. This led to the selection of the BodyGuard 2 (Firstbeat Technologies, Finland) as illustrated in Figure 3.1. HRV measures obtained from portable devices have a small level of error compared to measures obtained from an ECG, however two recent reviews of the literature have concluded that this small error is acceptable given the superior convenience and efficiency of the portable devices.^{455, 456}

Figure 3.1 Application of BodyGuard 2 heart rate monitor.



Figure 3.1 shows the device used to capture R-R interval data in this study. **(A)** The Firstbeat BodyGuard 2 device without electrodes attached. The large 'head' of the device could be removed, exposing a USB-A port which could connect to a computer for data download. **(B)** The head of the monitor attached below the collarbone on the right side of the body by an adhesive electrode. The tail-end of the device also attached to the surface of the skin by an electrode and was positioned on the left side of the body on the rib cage. Image taken and adapted from Firstbeat Technologies (2021).

Due to its portability, the BodyGuard 2 remained attached to participants for the entire duration of the study. It was attached to participants at two sites on their torso (Figure 3.1) after these

sites had been cleaned using isopropyl alcohol wipes. The device recorded continuously, even when outside of the laboratory, and this allowed for subject's HRV to be calculated at any point. After the device was returned from each participant at the conclusion of the study, the R-R interval data was downloaded onto a computer and processed using Kubios HRV Premium. Extrapolation of HRV measures from this point onward followed the exact same methodology as described in Section 2.1.5.2. The main difference between the Pilot Study and the T2D Study in regard to HRV determination was the method by which R-R interval data was obtained.

3.1.4.1 Selection of HRV Intervals to be Analysed

The BodyGuard 2 recorded R-R interval data continuously, and thus there were many options when it came to selecting which periods to analyse. It was also possible to analyse R-R data in periods longer than 10 minutes. However, it was important that all periods to be compared were of the same duration. This is because longer recordings, by their nature, show increased HRV, and thus statistical comparison between a 10-minute and a 20-minute epoch would be meaningless.³⁵⁸ Consequently, the focus of the UTS T2D Study was to capture 10-minute intervals of HRV data that corresponded with noteworthy states during the study. The first of these states was 'fasting HRV' which was calculated from the 10 minutes of R-R data that immediately preceded the fasting BGL assessment. This state may also represent 'resting HRV', as participants were allowed to rest for 10 minutes prior to recording this 10-minute state. Fasting HRV was recorded to set a baseline, as it was an estimation of autonomic activity without the effects of caloric intake.

The second of these noteworthy states was 'prandial HRV', which corresponded with autonomic activity whilst participants were eating their meal. It was calculated by selecting 10 minutes of R-R interval data that overlapped with the eating period of each participant, then processing that into HRV measures. As with the Pilot Study, subjects in the UTS T2D Study were asked to consume a regular meal after their initial fasting assessment (see Section 2.1.6.1). However, instead of returning at a set time, such as 10:00am, participants were required to return 30 minutes after finishing their meal. They were also asked to report on the contents of their meal (see Section 2.1.6.2 for more details), as well as the times in which they started and finished their meal. In the Pilot Study, it was recognized that circadian rhythms were responsible for natural but significant variation in HRV and BGL (Section 2.1.4). This is true not just for people without diabetes, but also people with diabetes.⁴⁰⁷ Therefore, all participants in this study were restricted to being assessed only in the morning, between the hours of 7:00am and 12:00pm.

3.1.5 Postprandial Assessment

There are two key stages of metabolism during which snapshot data of BGL is considered most useful. The first of these is fasting, when BGL is lowest, and the second is postprandial, when BGL peaks after a meal. One of the most desirable outcomes of new and upcoming measures of BGL is the ability to accurately estimate BGL during either of these two states (Section 1.11). A device capable of non-invasively estimating glycaemia in an individual in a fasting state as well as a postprandial state would represent an appealing alternative to current SMBG. Other authors have reinforced the importance of assessing both fasting and postprandial BGL when investigating the relationship of BGL with HRV measures.⁴⁵⁷ As with the Pilot Study, fasting BGL in the UTS T2D Study was captured using a blood glucose assessment after participants undertook an overnight fast. When attempting to capture postprandial BGL, there were several factors that needed to be considered. True postprandial BGL is difficult to measure, as the blood glucose assessment is an intermittent measure of BGL and thus cannot track glycaemia continuously. Even readings from CGM systems lag behind true postprandial BGL as it changes rapidly after a meal. Another factor to consider in this study was that different participants chose meals of different compositions, as expected, however this affected the time required for BGL to reach its peak.

Based on evidence provided in the Discussion of the Pilot Study (Section 2.3), as well as evidence from the literature, it was expected that peak BGL would be achieved 45 minutes on average after participants ate a regular meal.⁴⁴² There are various factors which affect the overall shape of the glucose curve for different individuals, however, the main confounder in the UTS T2D Study was the nutritional content of the meal. Meals high in carbohydrates require less time to produce a peak glycaemic response. To accommodate for the fact that some participants in this study consumed meals higher or lower in carbohydrates than average, which would cause their BGL to peak earlier or later, respectively, all participants underwent a BGL assessment twice. The first was 40 minutes after eating, and the second was 50 minutes after eating, for a total of two blood glucose assessments 10 minutes apart (Figure 3.2). Both BGL values were recorded, but only the higher value was used in the analysis as it was a close approximation of peak postprandial BGL.

Figure 3.2 Timing of assessments in the UTS T2D Study.



Figure 3.2 summarizes the timing of the three glucose assessments relative to the meal and the recording of heart rate. In short, recording of HRV began prior to the fasting BGL assessment, and ended only once the final postprandial assessment was measured. Following the fasting assessment, participants were asked to leave the laboratory and consume a meal of their choice nearby, as well as note down the time at which they finished the meal. Participants returned 40 minutes after this point for a second blood glucose assessment. A third assessment was conducted 10 minutes after the second, and the higher of these two was used as the postprandial BGL. **BGL** = Blood glucose level, **HRV** = Heart rate variability, **Min** = Minutes.

3.1.6 HbA_{1c} Assessment

HbA_{1c} level is considered an essential tool for monitoring diabetes, and people living with T2D routinely undergo assessments of HbA_{1c} as part of how they manage their condition. To properly determine whether HRV measures are predictive of glycaemia, HbA_{1c} levels were assessed in people with T2D to encourage further research into the value of HRV as a predictor of glycaemia. In the present study, HbA_{1c} levels were obtained from each T2D participant using the A1cNow+ Portable HbA_{1c} Test Kit (Point of Care Diagnostics Pty Ltd, North Rocks, Australia). These disposable devices measured HbA_{1c} levels from a blood sample, providing results within five minutes. Like the blood glucose assessment (Section 2.1.4), the HbA_{1c} assessment required only a small sample of blood from a fingerprick. This assessment was carried out at the same time as the fasting BGL assessment, because HbA_{1c} levels reflect only long-term glycaemia, they are not confounded by short-term metabolic states such as fasting.⁴⁵⁸ HbA_{1c} levels of subjects without diabetes were not recorded because it was anticipated there would be little to no variation based on previous literature, and thus analysis would reveal little of importance.

3.1.7 Statistical Analysis

The statistical analysis investigated HRV and BGL in people with TD2 and people without T2D separately. Independent t-tests were performed in SPSS version 22.0 (IBM SPSS Statistics, USA)

to identify any significant (p<0.05) differences between the T2D group and the group without T2D in various categories. These included demographic features such as age, as well as physiological features such as blood pressure, BGL, and HRV measures. Independent t-tests were also used to compare these same sets of factors between men and women in each group. Primarily, this was to determine whether there was an appreciable effect of sex on the variables of interest, as this would require that sex be controlled for in the statistical analysis. For comparison of 'within-subjects' changes, such as comparing the change in BGL from fasting to postprandial levels in just people without diabetes, paired-sample t-tests were used.

Furthermore, the aim of the analysis was to explore the potential for HRV measures to predict glycaemia in both groups. The predictor variables were largely measures of HRV, including LF power, HF power, total power, and LF:HF. For a detailed description of how these measures were obtained, see Section 2.1.5.2. Additional predictor measures were age, BMI, and kilojoule intake. The dependent variables that the present study sought to predict were fasting BGL, postprandial BGL, and, in the group with diabetes, also HbA_{1c} . Correlation analysis was conducted on BGL data and HRV measures to explore potential relationships between the variables. Partial Pearson's correlations were used in all instances, controlling for age and BMI. Additionally, kilojoule intake was applied as a covariate for postprandial BGL, which may have been confounded by the variable meals of participants, but not for fasting BGL correlations, as participants had not eaten yet for this assessment. In instances where multiple HRV measures were correlated with a single dependent variable, such as fasting BGL or postprandial BGL, a multiple linear regression was carried out to identify the strongest predictor variables and to establish a regression model demonstrating the proportion of the outcome variable (BGL) which can be explained by the predictor variables (HRV measures). Regression models, a basic form of machine learning, perform reasonably well compared to advanced machine learning models in terms of predicting glycaemic events in diabetes.³³⁶ However, it is important that the model is implemented in real-time, as this provides clinical significance.

To ensure that the assumptions of the ANOVA were not violated, the data set was checked for outliers, normal distribution, linear relationships, homoscedasticity, and multicollinearity. Normal Q-Q plots of the studentized residuals were used to check that the data points were linearly organised, which confirmed normal distribution. Linear relationships between the dependent variable (BGL) and independent variables (HRV measures) were confirmed by scatter plots and are shown in the Results. Homoscedasticity was assumed by checking that scatterplots of the residuals were equally distributed and did not follow any pattern or skewed distribution. To ensure the independent variables were not highly correlated with each other, which may affect the interpretation of the multiple regression model, the variables were checked for multicollinearity in SPSS. Variance inflation factor values were checked to ensure they were between 1-10, indicating no multicollinearity.

To determine a sample size (n) which would provide sufficient power for the study, J. Cohen's "A Power Primer" was consulted. For a correlation analysis between two independent groups, which included a T2D group and a group of subjects without T2D, with a significance (α) set to 0.05 and a large effect size (r) of 0.5, a sample size of n=30 in each group would provide a power level of 0.8, or an 80% probability of rejecting a false null hypothesis at the 5% significance level. As such, the recruitment goal was at least 30 people with and 30 people without T2D.

3.2 Results

The UTS T2D Study investigated the relationship between frequency-domain HRV measures and glycaemia in a sample of people with T2D as well as a sample of people without any chronic illness. Recruitment was facilitated through social media advertising as well as with assistance from Diabetes Australia. For this study, recruitment of n=30 participants in each group was not possible due to COVID-19 restrictions implemented by the Australian Federal Government and NSW State Government after 21st March 2020, with repeated restrictions imposed between 2020 and 2021. As such, the UTS T2D study was concluded early after n= 27 people with T2D and n=29 subjects without T2D were recruited.

3.2.1 Demographic Differences: Non-Diabetes vs. Diabetes

Notable demographic data, including mean age and BMI of each sample, is presented in Table 3.1. Additionally, mean scores are provided for the LAQ Part 1 (scales higher with lifestyle risk factors, such as poor diet) and LAQ Part 2 (assesses the effect of stress on a person's life). Independent t-tests established whether there were any significant differences (p<0.05) in these variables between the two samples. Additional information was collected from the T2D sample, for example the mean HbA_{1c} level was $6.9 \pm 1.4\%$ and the median level was 6.6%. The mean duration of diabetes was 10 ± 8 years. 52% of the T2D sample reported that they knew of at least one first-degree relative with T2D, and 26% knew of at least one first-degree and one second-degree relative with T2D.

Table 3.1 Means and standard deviations for demographic differences between the group without diabetes (n=29) and the group with diabetes (n=27). Data is presented as mean \pm standard deviation. BMI was calculated as weight (in kilograms) divided by height (in metres) squared. The LAQ Part 1 was scored separately from Part 2, and mean scores are provided in this table. *** = p<0.001, **BMI** = Body mass index, **kg/m**² = Kilogram per square metre, **LAQ** = Lifestyle Appraisal Questionnaire.

Variable	Non-diabetes group	Type 2 diabetes	p-value
n	29	27	-
Age (years)	33 ± 13	51 ± 10	<0.001***
BMI (kg/m ²)	24 ± 3	33 ± 6	<0.001***
LAQ Part 1	10 ± 8	20 ± 6	<0.001***
LAQ Part 2	18 ± 13	22 ± 10	0.24

From Table 3.1, it can be inferred that individuals in the T2D sample were approximately 18 years older on average (p<0.001) and had higher BMI (p<0.001) compared to individuals in the group without diabetes. As such, correlation analyses may need to control for these demographic differences. People with T2D also scored higher in the LAQ Part 1 (p<0.001), and though mean LAQ Part 2 scores were also higher in the T2D sample, the difference compared to the group without diabetes was not significant.

3.2.2 Physiological Differences: Non-Diabetes vs. Diabetes

Mean values for physiological data, including blood pressure, BGL, and HRV measures, are presented in Table 3.2. Blood pressure was averaged from pre-study and post-study values. Kilojoule intake was determined using a food diary which participants completed after eating a regular meal. Independent t-tests were conducted to determine whether there were any significant differences in these physiological measures between the two sample groups.
Table 3.2 Means and standard deviations for physiological differences between the group without diabetes (n=29) and group with diabetes (n=27). Data is presented as mean ± standard deviation. Fasting BGL was recorded after a fast of at least eight hours and postprandial BGL was recorded 40-50 minutes after a meal. Fasting HRV was determined using the first 10-minutes of the recording. LF power, HF power, and total power are natural log transformed. * = p<0.05, ** = p<0.01, *** = p<0.001, **BGL** = Blood glucose level, **BP** = Blood pressure, **HF** = High frequency, **HRV** = Heart rate variability, **kJ** = Kilojoule, **LF** = Low frequency, **LF:HF** = low to high frequency ratio, **mmHg** = Millimetres of mercury, **mmol/L** = Millimoles per litre, **ms**² = Milliseconds squared.

Variable	Non-diabetes group	Type 2 diabetes	p-value
BP (mmHg)			
Systolic	115 ± 10	124 ± 12	<0.01**
Diastolic	74 ± 7	83 ± 8	< 0.001***
BGL (mmol/L)			
Fasting	4.9 ± 0.5	8.1 ± 2.2	<0.001***
Postprandial	6.4 ± 1.1	11.2 ± 3.7	<0.001***
Change in BGL	1.4 ± 1.0	3.1 ± 2.4	<0.01**
Kilojoule intake (kJ)	2280 ± 1537	1600 ± 878	0.05*
Fasting HRV (ms ²)			
LF power	6.9 ± 0.7	5.7 ± 1.2	<0.001***
HF power	5.8 ± 0.8	4.3 ± 1.4	<0.001***
Total power	7.6 ± 0.6	6.3 ± 1.1	<0.001***
LF:HF	3.9 ± 2.6	5.9 ± 4.1	0.04*

Included in Table 3.2 were a considerable number of significant differences between the two different groups. Systolic (p<0.01) and diastolic (p<0.001) blood pressure were both 9 mmHg higher on average in the T2D sample compared to the sample without diabetes. This difference is highlighted in Figure 3.3

Figure 3.3 Comparison of systolic and diastolic blood pressure between participants without diabetes (n=29) and diabetes participants (n=27).





Figure 3.3 provides blood pressure means with error bars as standard deviations. Blood pressure was averaged from the three pre-study and three post-study recordings, all of which were conducted after five minutes at rest. ** = p<0.01, *** = p<0.001, **BP** = Blood pressure, **mmHg** = Millimetres of mercury, **T2D** = Type 2 diabetes.

Table 3.2 also presented BGL data. Compared to the group without T2D, BGL was higher in the T2D group in both the fasting (p<0.001) and postprandial (p<0.001) state. In addition to this, paired t-tests determined that BGL increased significantly from fasting to postprandial in the group without diabetes (p<0.001), and in the T2D group (p<0.001). These results are visualized in Figure 3.4. There was also greater variability in BGL for the T2D sample, as indicated by the larger standard deviations compared to the sample without diabetes. There was a significantly higher change in BGL across the T2D group (p<0.01). During the meal, T2D participants consumed almost 700 kilojoules less on average than their counterparts without diabetes.



Figure 3.4 Comparison of blood glucose level and kilojoule intake between the group without diabetes (n=29) and the diabetes group (n=27).

Figure 3.4 compares the means of fasting and postprandial BGL with error bars as standard deviations. BGL was compared at fasting vs. postprandial, as well as non-diabetes vs. diabetes. Paired t-tests compared fasting BGL with postprandial BGL in the non-diabetes group and compared fasting BGL with postprandial BGL in the T2D group. Independent t-tests compared fasting BGL between the two samples and then compared postprandial BGL between the two samples. * = p<0.05, ** = p<0.01, *** = p<0.001, BGL = Blood glucose level, kJ = Kilojoule, mmol/L = Millimoles per litre, T2D = Type 2 diabetes.

Finally, Table 3.2 also presented HRV data from the fasting state, which represents baseline HRV. Though not shown, prandial HRV showed the same differences between non-T2D and T2D samples. For simplicity, only fasting HRV was used to compare baseline HRV between the groups. It is important to remember that, though it is easy to compare different HRV measures because some share the same units (ms²), this is not an appropriate analysis. Generally, LF power is high than HF power, just as systolic blood pressure is usually higher than diastolic blood pressure. This is not a meaningful comparison. Independent t-tests were only conducted to determine if HRV measures were significantly different between the group with diabetes and the group without. All measures were significantly different between the two groups. People with T2D had lower LF power (p<0.001), HF power (p<0.001), and total power (p<0.001) but higher LF:HF (p<0.05). These differences are highlighted in Figure 3.5.



Figure 3.5 Comparison of heart rate variability measures between non-diabetes (n=29) and diabetes groups (n=27).

Figure 3.5 presents the means of the different HRV measures with error bars as standard deviations. HRV measures were from the fasting state and were determined using the first 10-minutes of the recording. * = p<0.05, *** = p<0.001, **HF** = High frequency, **LF** = Low frequency, **ms**² = Milliseconds squared, **T2D** = Type 2 diabetes.

3.2.2.1 Males vs. Females

The sample size was sufficiently large to compare differences in males (n=15) and females without diabetes (n=14), as well as T2D males (n=13) and T2D females (n=14). The primary goal was to establish whether there were significant differences between men and women in the samples, as this would have indicated that sex had an appreciable confounding effect on the data and should be controlled for in the statistical analysis. Mean age and BMI, as well as kilojoule intake, are compared in Table 3.3 for the sample without diabetes as well as the T2D sample. Mean LAQ scores were also compared between males and females, but no significant differences were observed. No significant differences were observed between age of males and age of females, or BMI of males and BMI of females, in either sample. Within the group without diabetes, males consumed over 1000 kilojoules on average than their female counterparts (p = 0.04).

Table 3.3 Demographic differences between males (n=15) and females (n=14) without diabetes, as well as males (n=13) and females with diabetes (n=14). Data is presented as mean \pm standard deviation. BMI was calculated as weight (in kilograms) divided by height (in metres) squared. Kilojoule intake was determined using a food diary which participants completed after eating a regular meal. * = p<0.05, **BMI** = Body mass index, **kg/m**² = Kilogram per square metre, **kJ** = Kilojoule, **T2D** = Type 2 diabetes.

	Non-diabetes T2D			D		
Variable	Males	Females	p-value	Males	Females	p-value
n	15	14	-	13	14	-
Age (years)	32 ± 14	33 ± 11	0.95	49 ± 12	51 ± 10	0.70
BMI (kg/m ²)	25 ± 4	24 ± 3	0.43	31 ± 6	34 ± 7	0.27
kJ intake	2900 ± 1900	1700 ± 800	0.04*	1800 ± 1100	1500 ± 600	0.37

Physiological data from each sample was also separated into males vs. females, including systolic and diastolic blood pressure, fasting and postprandial BGL, and HRV measures (Table 3.4). The four main HRV measures that were investigated in this study were LF power, HF power, total power, and LF:HF. Only HRV measures from the fasting state are presented in Table 3.4. There were no significant differences in systolic or diastolic blood pressure between males and females for either sample. To demonstrate that differences in physiological variables between diabetes and non-diabetes groups were consistent for both males and females, additional independent t-tests were used to compare physiological data in males between the two samples as well as females between the two samples. The results from these analyses are presented in Figure 3.6. It was determined that males in the T2D sample had higher diastolic blood pressure than their non-diabetes counterparts (p<0.05). Compared to females in the non-diabetes sample, T2D females had both higher systolic (p<0.05) and diastolic (p<0.01) blood pressure. Statistical analysis shown in Table 3.4 only compares men and women from the same group, either T2D or non-diabetes. Statistical comparison of men without diabetes compared with men with diabetes, for example, was conducted but is only presented in figures.

Table 3.4 Physiological differences between males (n=15) and females (n=14) without diabetes, as well as males (n=13) and females with diabetes (n=14). Data is presented as mean \pm standard deviation. Systolic and diastolic blood pressure values were averaged from pre-study and post-study values. Change in BGL = 'postprandial' subtracted by 'fasting' BGL. Fasting HRV was determined using the first 10 minutes of the recording. **BGL** = Blood glucose level, **BP** = Blood pressure, **HF** = High frequency, **HRV** = Heart rate variability, **LF** = Low frequency, **LF:HF** = low to high frequency ratio, **mmHg** = Millimetres of mercury, **mmol/L** = Millimoles per litre, **ms**² = Milliseconds squared.

	Non-diabetes			TZ		
Variable	Males	Females	p-value	Males	Females	p-value
n	15	14	-	13	14	-
BP * (mmHg)						
Systolic	116 ±11	113 ± 11	0.50	125 ± 13	124 ± 12	0.74
Diastolic	74 ± 8	74 ± 7	0.91	81 ± 8	85 ± 8	0.32
BGL (mmol/L)						
Fasting	5.1 ± 0.5	4.7 ± 0.4	0.05	8.0 ± 2.0	8.2 ± 2.4	0.83
Postprandial	6.5 ± 1.1	6.2 ± 1.1	0.60	10.7 ± 3.0	11.6 ± 4.4	0.58
Change	1.4 ± 1.1	1.5 ± 1.0	0.74	2.7 ± 2.4	3.4 ± 2.5	0.51
Fasting HRV (ms ²)						
LF power	6.7 ± 0.7	7.1 ± 0.7	0.16	5.9 ± 1.2	5.6 ± 1.2	0.53
HF power	5.7 ± 0.7	5.8 ± 0.9	0.78	4.3 ± 1.6	4.2 ± 1.3	0.84
Total power	7.5 ± 0.5	7.8 ± 0.6	0.11	6.2 ± 1.2	6.3 ± 1.1	0.96
LF:HF	3.6 ± 2.6	4.3 ± 2.2	0.48	5.9 ± 4.3	5.8 ± 4.1	0.92

Figure 3.6 Differences in blood pressure between males (n=15) and females (n=14) without diabetes as well as males (n=13) and females with diabetes (n=14).



Figure 3.6 presents the means of blood pressure with error bars as standard deviations. Systolic and diastolic blood pressure values were averaged from pre-study and post-study values. Systolic and diastolic values are reported at different levels on the same scale, as diastolic blood pressure is generally lower than systolic. * = p<0.05, ** = p<0.01, **BP** = Blood pressure, **mmHg** = Millimetres of mercury, **T2D** = Type 2 diabetes.

Returning to Table 3.4, fasting BGL was not significantly different in males compared to females without diabetes, or between males and females with T2D. Neither postprandial BGL nor the change in BGL from fasting to postprandial was significantly different between the sexes in either sample. Though not indicated in the table, independent t-tests confirmed that fasting (p<0.001) and postprandial (p<0.001) BGL were both higher in T2D males (p<0.001) and T2D females (p<0.001) compared to their non-diabetes counterparts. These results are shown in Figure 3.7

Figure 3.7 Differences in blood glucose level between males (n=15) and females (n=14) without diabetes, as well as males (n=13) and females with diabetes (n=14).



Figure 3.7 visualises the means of fasting and postprandial BGL with error bars as standard deviations. Fasting BGL was recorded after a caloric restriction of at least eight hours and postprandial BGL was recorded 40-50 minutes after eating a meal. * = p<0.05, *** = p<0.001, **BGL** = Blood glucose level, **mmol/L** = Millimoles per litre, **T2D** = Type 2 diabetes.

Statistical analysis was used to explore differences in each measure between males and females. Though different HRV measures can appear similar in units, it is not appropriate to compare them, as this would be analogous to comparing differences between systolic and diastolic blood pressure. Differences in their values are due to the intrinsic nature of each measure. In Table 3.4 it was shown that there were no significant differences in any HRV measure between males and females in the sample without diabetes, and none in the T2D sample. These analyses are visualised in Figure 3.8 for the sample without diabetes and Figure 3.9 for the T2D sample.

Figure 3.8 Differences in heart rate variability measures between males (n=15) and females (n=14) without diabetes.



Figure 3.8 presents differences in HRV measures between males and females without diabetes with error bars as standard deviations. HRV measures are all from the fasting state, which was determined using the first 10 minutes of the recording. There are significant differences in power level, but this is implicit in their nature. **BGL** = Blood glucose level, **HF** = High frequency, **LF** = Low frequency, **ms**² = Milliseconds squared.

Figure 3.9 Differences in heart rate variability measures between males with diabetes (n=13) and females with diabetes (n=14).



Figure 3.9 presents differences in HRV measures between T2D males and females with error bars as standard deviations. HRV measures are all from the fasting state, which was determined using the first 10 minutes of the recording. **BGL** = Blood glucose level, **HF** = High frequency, **LF** = Low frequency, **ms**² = Milliseconds squared, **T2D** = Type 2 diabetes.

3.2.3 Correlation Analysis

The primary aim of the correlation analysis was to investigate associations between HRV measures and BGL in each sample as a basis for developing regression models, which are a basic form of machine learning. Before this analysis could be performed, it was important to determine if any confounding variables existed in the data set, as partial Pearson's correlation allows for multiple variables to be included as covariates to account for their confounding effect. Based on the literature and the scope of this PhD candidature, the main confounders of concern for this data set were sex, kilojoule intake, age, and BMI. In Section 3.2.2.1, it was demonstrated that there were no significant differences between men and women for either sample, with the exception that kilojoule intake was lower in women compared to men. As such, sex was not controlled for in the partial correlation analysis. Kilojoule intake was identified as a confounder and was included as a covariate for any correlation analysis involving postprandial BGL, for otherwise there would be no manner of controlling for the different meal composition of participants. As subjects had not yet eaten for the fasting assessment, kilojoule intake was not included as a covariate for any analysis involving BGL.

Several methods were used to determine the confounding effect of age and BMI on the data. These two were correlated against the variables of interest, which were fasting BGL, postprandial BGL, and each of the four HRV measures from the fasting state as well as the prandial state. For the group without diabetes, participant age was correlated with fasting BGL (r = 0.46; p = 0.02) but not postprandial BGL. Age was also associated with prandial LF power (r = -0.47; p = 0.02), prandial HF power (r = -0.46; p = 0.02), and prandial total power (r = -0.43; p = 0.03), but not prandial LF:HF. Age was also not related to any of the four HRV measures from the fasting state. Participant BMI was significantly correlated with fasting BGL (r = 0.47; p =0.01) and postprandial BGL (r = 0.41; p = 0.03). As with age, BMI was significantly correlated with prandial LF power (r = -0.55; p < 0.01), prandial HF power (r = -0.40; p = 0.04), and prandial total power (r = -0.49; p = 0.01), but not prandial LF:HF or any of the four HRV measures from the fasting state. As for the T2D group, age and BMI weren't associated with any of the variables of interest, which included fasting and postprandial BGL, and each of the four HRV measures from the fasting state as well as the prandial state. Though these findings indicated that age and BMI were confounders in only the sample without diabetes, age and BMI were included as covariates in the analysis of both the sample without diabetes and the T2D sample for consistency. A review of the literature also indicated that controlling for age and BMI in the analysis was appropriate, as most studies agree they are strongly related and can confound the statistical analysis.^{459, 460} It is important that the analysis of the present study investigates whether HRV measures predict BGL even when accounting for age and BMI, as well as kilojoule intake.

3.2.3.1 Partial Correlations

Partial Pearson's correlations were used to investigate associations between HRV and BGL in the two groups separately, as correlations may be stronger or weaker in T2D compared to people without diabetes. HRV measures from the fasting state, including LF power, HF power, total power, and LF:HF, were correlated against both fasting and postprandial BGL for the group without diabetes. This analysis was repeated using data from the T2D group (Table 3.5). Then, the same four HRV measures were extrapolated from the prandial state – when participants were eating – and correlated against postprandial BGL, but not fasting BGL. This was because the prandial state comes after fasting, and there is little value in predicting BGL from a previous metabolic state using HRV. As with the fasting HRV measures, this analysis was conducted for both the group with and the group without T2D separately (Table 3.5). Also, kilojoule intake was not included as a covariate for HbA_{1c}, as recent food intake has no effect on HbA_{1c} levels. HbA_{1c} was not significantly associated with any HRV measure when controlling for age and BMI, though HbA_{1c} was significantly associated with fasting BGL (r = 0.63; p < 0.01) and postprandial BGL (r = 0.60; p < 0.01) in the diabetes group. As HbA_{1c} was not recorded in participants without diabetes, there was no correlation analysis to perform in this group. Table 3.5 Partial Pearson's correlations between heart rate variability measures and blood glucose level, accounting for age, body mass index, and kilojoule intake as covariates in the group without diabetes (n=29) and the diabetes group (n=27). Data is presented as mean \pm standard deviation. The missing r and p values for fasting blood glucose level are because these analyses do not fit the scope of this PhD. It was only relevant to investigate whether heart rate variability measures could predict current of future blood glucose. * = p<0.05, ** = p<0.01, **BGL** = Blood glucose level, **BMI** = Body mass index, **HF** = High frequency, **HRV** = Heart rate variability, **LF** = Low frequency, **LF:HF** = low to high frequency ratio, **p** = p-value, **r** = correlation coefficient.

Variable	Non-	diabetes	Type 2 diabetes		
variable —	Fasting BGL	Postprandial BGL	Fasting BGL	Postprandial BGL	
Fasting HRV					
LF power					
r	0.10	-0.13	-0.63	-0.46	
р	0.64	0.55	<0.01**	<0.05*	
HF power					
r	0.04	-0.31	-0.57	-0.33	
р	0.84	0.15	<0.01**	0.17	
Total power					
r	-0.09	-0.43	-0.66	-0.51	
р	0.69	0.04*	<0.01**	0.03*	
LF:HF					
r	-0.03	0.16	0.35	0.08	
р	0.89	0.47	0.14	0.75	
Prandial HRV					
LF power					
r	-	-0.33	-	-0.45	
р	-	0.13	-	0.05	
HF power					
r	-	-0.39	-	-0.43	
р	-	0.07	-	0.07	
Total power					
r	-	-0.61	-	-0.48	
р	-	<0.01**	-	0.04*	
LF:HF					
r	-	0.47	-	0.24	
р	-	0.02*	-	0.31	

From Table 3.5 it can be inferred that fasting BGL was not significantly associated with any of the fasting HRV measures in the group without diabetes. However, postprandial BGL was correlated with fasting total power (r = -0.43; p = 0.04), prandial total power (r = -0.61; p < -0.61

0.01), and prandial LF:HF (r = 0.47; p = 0.02) in the group without diabetes. The scatter plots for these significant correlations are presented in Figure 3.10.



Figure 3.10 Scatter plots for postprandial blood glucose level and significantly correlated heart rate variability measures in the sample without diabetes (n=29).

Figure 3.10 describes the relationship between each HRV measure and the dependant variable, postprandial BGL, using scatter plots. **(A)** Fasting total power was negatively associated with postprandial BGL, with a correlation coefficient (r) of -0.43 after controlling for covariates. **(B)** Prandial total power was also negatively correlated with BGL, with a correlation coefficient (r) of -0.61. **(C)** Prandial LF:HF had a positive relationship with BGL, with an r-value of 0.47. **BGL** = Blood glucose level, **LF:HF** = low to high frequency ratio, **mmol/L** = Millimoles per litre, **ms**² = Milliseconds squared.

Also in Table 3.5, several significant correlations were also identified in the T2D sample. Fasting BGL was negatively associated with fasting LF power (r = -0.63; p < 0.01), HF power (-0.57; p = 0.02), and total power (r = -0.66; p < 0.01). The relationship between each of these is described further by scatter plots in Figure 3.11. For postprandial BGL in the T2D sample, the HRV

measures which were significantly correlated were fasting LF power (r = -0.46; p < 0.05), fasting total power (r = -0.51; p = 0.03) and prandial total power (r = -0.48; p = 0.04). Scatter plots for these correlations are shown in Figure 3.12.





Figure 3.11 displays the relationship between each HRV measure and fasting BGL using scatter plots. **(A)** Higher fasting BGL was associated with lower fasting LF power. R was equal to -0.63 after controlling for covariates. **(B)** People with higher fasting BGL tended to have lower HF power, with an r-value of -0.57. **(C)** Fasting total power was also negatively correlated with fasting BGL. This was the strongest relationship (r = -0.66). **BGL** = Blood glucose level, **mmol/L** = Millimoles per litre, **ms**² = Milliseconds squared.





Figure 3.12 shows the relationship between the HRV measures observed in fasting and eating diabetes participants and postprandial levels of blood glucose. **(A)** Lower levels of low frequency power tended to be associated with higher levels of blood glucose (r = -0.46). **(B)** Total power measured in a fasting state was associated with levels of blood glucose after a meal (r = -0.51). **(C)** Postprandial BGL was also associated with total power measured in a prandial state (r = -0.48). **BGL** = Blood glucose level, **mmol/L** = Millimoles per litre, **ms**² = Milliseconds squared.

3.2.4 Multiple Linear Regression

Given that there were three HRV measures that were significantly correlated with postprandial BGL in the sample without diabetes, a multiple linear regression was performed (Table 3.6). Fasting total power, prandial total power, and prandial LF:HF were entered into the model. All predictors were retained by the model, but none were statistically significant.

Table 3.6 Regression analysis for postprandial blood glucose and the significantly correlated heart rate variability measures in the group without diabetes (n=29). 'B' and 'Standard Error' are unstandardized coefficients as the variables remain on their respective scales, for example total power scale is represented in units of milliseconds squared and blood glucose level is represented in units of millimoles per liter. 'Beta' are standardized coefficients as the variables have been adjusted to fit on the same scale. **BGL** = Blood glucose level, **LF:HF** = low to high frequency ratio.

Regression summary for dependent variable: postprandial BGL R = 0.66; R ² = 0.44; Adjusted R ² = 0.36; F(3,22) = 5.67 p = 0.005, Standard Error of the Estimate = 0.90								
Variable B Standard Beta t p Error								
(Constant)	14.36	3.11		4.61	0.00			
Fasting total power	-0.39	0.36	-0.22	-1.10	0.28			
Prandial total power	-0.70	0.46	-0.38	-1.53	0.14			
Prandial LF:HF 0.07 0.08 0.19 0.90 0.38								

A second regression was conducted in the diabetes sample for fasting BGL and the three significantly correlated HRV measures. Fasting LF, fasting HF, and fasting total power were entered into the model. All predictors were retained by the model, but none were statistically significant (Table 3.7).

Table 3.7 Regression analysis for fasting blood glucose and the significantly correlated heart rate variability measures in the diabetes group (n=27). 'B' and 'Standard Error' are unstandardized coefficients as the variables remain on their respective scales, for example total power scale is represented in units of milliseconds squared and blood glucose level is represented in units of millimoles per liter. 'Beta' are standardized coefficients as the variables have been adjusted to fit on the same scale. **BGL** = Blood glucose level, **HF** = High Frequency, **LF** = Low frequency.

Regression summary for dependent variable: fasting BGL R = 0.74; R ² = 0.55; Adjusted R ² = 0.49; F(3,22) = 9.09 p < 0.001, Standard Error of the Estimate = 1.61							
Variable B Standard Beta t I							
(Constant)	17.41	1.92		9.09	0.00		
Fasting LF	-0.82	0.71	-0.42	-1.16	0.26		
Fasting HF	0.21	0.38	0.13	0.54	0.59		
Fasting total power -0.88 0.69 -0.44 -1.27 0.22							

For postprandial BGL in the diabetes sample, there were significant associations with three of the eight HRV measures that were investigated. Therefore, a linear regression was performed using fasting LF power, fasting total power, and prandial total power as the predictor inputs, and postprandial BGL was entered as the predicted outcome Table 3.8. The three predictor variables were retained by the model, but none were statistically significant within the model.

Table 3.8 Regression analysis for postprandial blood glucose and the significantly correlated heart rate variability measures in the diabetes group (n=27).'B' and 'Standard Error' are unstandardized coefficients as the variables remain on their respective scales, for example total power scale is represented in units of milliseconds squared and blood glucose level is represented in units of millimoles per liter. 'Beta' are standardized coefficients as the variables have been adjusted to fit on the same scale. **BGL** = Blood glucose level, **LF** = Low frequency.

Regression summary for dependent variable: postprandial BGL $R = 0.52; R^2 = 0.27; Adjusted R^2 = 0.16; F(3,21) = 2.550$ n = 0.083 Standard Error of the Estimate = 3.22							
Variable B Standard Beta t p							
(Constant)	21.61	3.91		5.53	0.00		
Fasting LF	-0.62	1.40	-0.19	-0.44	0.66		
Fasting total power	-0.54	1.65	-0.17	-0.39	0.75		
Prandial total power	-0.62	0.89	-0.20	-0.69	0.50		

3.2.5 Summary of Results

In this study, n= 27 people with T2D and n=29 people without diabetes were recruited and assessed. Individuals in the T2D sample were approximately 18 years older on average (p<0.001) and had higher BMI (p<0.001) compared to individuals in the sample without diabetes. People with T2D also tended to score higher on the LAQ Part 1 (p<0.001), which is associated with lifestyle risk factors such as poor diet and low physical activity. There were many significant differences between the two groups in terms of blood pressure, BGL, and measures of HRV. Comparisons were also made between the sexes in each sample, and the statistical analysis revealed there were no meaningful differences between men and women in terms of demographics or physiology, though women without diabetes did consume an average of 1200 kilojoules less than their male counterparts in this study.

Multiple correlation analyses were performed, using age and BMI as covariates as well as kilojoule intake where appropriate. Scatter plots and correlation coefficients determined that there was a general inverse correlation between BGL and HRV measures, for both samples. Where two or more measures of HRV were significantly correlated with BGL, a multiple linear regression was performed. From the multiple linear regression analysis in Table 3.6, it was determined that HRV measures, including fasting total power, prandial total power, and prandial LF:HF, accounted for 44% of the variance in postprandial BGL observed in the sample without diabetes (R²=0.44). In Table 3.7, the regression analysis revealed that HRV measures obtained from a fasting state alone, including LF power, HF power, and total power, accounted for 55% of the variation in the dependent variable: fasting BGL in T2D (R²=0.55). The third multiple linear regression analysis revealed that fasting LF and total power and prandial total power accounted for 27% of the variance in postprandial BGL in the diabetes group (Table 3.8). This concludes the statistical analysis and presentation of results. The following chapter will interpret these results and discuss them in the context of the literature.

3.3 Discussion

The UTS T2D Study aimed to add to the growing body of literature which has suggested HRV measures may be valuable in glucose prediction. The results of this research may invite further interest in the applications of HRV in diabetes management and may justify further research into this topic. In this section, the results of the present study will be discussed in the context of the current literature, and the core strengths and weaknesses of the study will be described, with suggestions for future research. In summary, there was an inverse correlation between BGL and HRV measures in general, for both the non-diabetes sample and the T2D sample. The multiple linear regressions indicated that HRV measures, including fasting total power, prandial total power, and prandial LF:HF, accounted for 44% of the variance in postprandial BGL observed in the sample without diabetes. Further, HRV measures obtained from a fasting state alone, including LF power, HF power, and total power, accounted for 55% of the variation in the dependent variable: fasting BGL in T2D. Finally, fasting LF and total power and prandial total power accounted for 27% of the variance in postprandial BGL in the diabetes group.

3.3.1 Sample Comparison

A roughly equal number of people without diabetes (n=29) and people with T2D (n=27) were recruited for this study. Due to the fact the people with T2D tend to be older and have a higher BMI compared to people who live without any chronic illness, it was expected that there would be meaningful differences between the samples in terms of demographics and physiology. Individuals with T2D were 18 years older on average and had higher BMI compared to their non-diabetes counterparts. A higher BMI in the T2D group was anticipated, as obesity is a major risk factor of T2D and is highly comorbid.⁴³ This was reinforced by the scores of the lifestyle questionnaire. The T2D group scored higher on the LAQ Part 1, indicating their lifestyles were associated with more risk factors, including modifiable risk factors such as poor diet and low physical activity, as well as non-modifiable risk factors such as family history of chronic illness. The higher age in the T2D group was also anticipated. However, there was a discrepancy between the mean age of the T2D group in this study compared to samples published in the literature. Though the median age of T2D onset is decreasing, with childhood obesity becoming more prevalent worldwide, T2D at present predominately affects older individuals, and the risk of developing T2D increases with age. A large-scale epidemiological study of Australians with T2D (n=743,709) conducted by Huo and colleagues (2018) indicated that the mean age was 65 years old.⁴⁶¹ Nanayakkara and colleagues (2018) reported a similar mean of 63 years of age in their sample of n=3,419 Australians with T2D,462 as did Knowles and colleagues (2020) who recruited n=59 Australian participants with T2D with an average age of 61 years old.⁴⁶³ In this study, the mean age of T2D participants was 51 \pm 10 years of age, which is 14 years younger than the large cohort of Huo and colleagues (2018). Evidently, the characteristics of the participants recruited in the UTS T2D Study do not accurately represent those of the wider Australian T2D population, as they were much younger on average. However, a lower HbA_{1c} and younger age of the T2D group is a desirable trait when investigating diagnostic capabilities of HRV, as the influence of autonomic neuropathy on BGL and HRV measures is lower in these groups.

In terms of physiology, there were major differences between the T2D group and the group without diabetes that were related to HRV measures, blood pressure, and BGL. LF power, HF power, and total power were all significantly lower on average in the T2D sample, and LF:HF – which is calculated differently to the other three measures, as it is a ratio of LF to HF power was significantly higher in the T2D sample. This result was expected, as there is substantial literature showing that autonomic activity, as measured by HRV, declines with age and with diabetes duration (see Section 1.10.1). BGL was also significantly higher in the T2D group, though this was also expected as this is the defining characteristic of this clinical sample, and the higher blood pressure was also anticipated as blood pressure increases with age and hypertension is highly comorbid with T2D.⁶⁶ These physiological differences between the samples are consistent with findings of previous literature, and analysis of these differences was conducted only to provide evidence that the diabetes participants and participants without diabetes recruited in this study were representative of their respective populations. In addition to this, the values for HRV measures, blood pressure, and BGL were found to be within the expected ranges for the participants without diabetes,^{432, 464} and this was also true for the T2D participants.^{375, 465} However, HbA_{1c} was assessed in T2D participants as an added measure of their glycaemic status. The mean HbA_{1c} for the sample was $6.9 \pm 1.4\%$, and 16 out of the total 27 participants (59%) presented with an HbA_{1c} level below 7.0%. Based on large-scale epidemiological data, only 50% of Australians with T2D maintain their HbA1c levels below the recommended level of 7.0%.¹⁰⁴ Therefore, there is a discrepancy of about 9% between this sample and the wider Australian diabetes population in terms of glycaemic management. This may be related to this sample being relatively young compared to the wider Australian T2D population, and glycaemic management tends to be better in younger people.⁴⁶⁶

3.3.1.1 Sex Comparison

Statistical analysis showed no significant difference between men and women for any demographic variable or for any of the physiological variables of interest, including HRV measures and BGL. A meta-analysis of 172 studies (n=63,612, 50% male) has demonstrated that LF power, total power, and LF:HF are lower, and HF power is greater in women compared to men.⁴⁶⁷ This suggests that autonomic control of the female heart is characterized by higher vagal, or parasympathetic activity, and the male heart is characterized by relatively higher sympathetic dominance. These sex differences are also generally greater with increasing age.⁴⁶⁷ This may explain why sex differences were unclear in the present study, as the sample did not include a wide range of age groups, and was mostly people 25-40 years old. The small sample size may also have been unable to detect the effect of sex on HRV measures in the statistical analysis, due to effect size.

In terms of sex differences in glycaemia, Kautzky-Willer and colleagues (2012) contend that fasting and postprandial BGL tend to be lower in women (n=611) with optimal glycaemia compared to men (n=361).⁴⁶⁸ In their study, the female sample was younger, and this may have contributed to their statistical findings. However, a broader view of the literature suggests there is a consensus that women without diabetes have lower fasting BGL, even though evidence for sex differences in postprandial BGL may be less conclusive. For example, research conducted by Faerch and colleagues (2010) in a sample of n=6,006 people without chronic illness showed that women have lower fasting BGL, but higher postprandial BGL were explained by height and fasting BGL, but differences in fasting BGL were not explained by height. In the UTS T2D Study, glucose levels were comparable between men and women. Mean fasting BGL in women was 4.7 \pm 0.4 compared to 5.1 \pm 0.5 in men. Though this was difference was not significant, it was close to significance (p=0.05) and increasing the sample size may have provided a significant p-value, aligning the results with that of previous literature.

There were also no physiological differences when comparing men and women in each sample, for example no significant differences between T2D males and T2D females. The only exception was women without diabetes consumed an average of 1200 kilojoules less than their male counterparts, and research from various countries suggests this is normal.⁴⁷⁰⁻⁴⁷² Where relevant, kilojoule intake was included as a covariate in the correlation analysis, and this accounted for the differences in meal composition between individuals, including the difference in average kilojoule intake between men and women.

3.3.2 Correlation Analysis and Confounders

Age and BMI were significantly correlated with measures of HRV and BGL. This was only true for the sample without diabetes – age and BMI were not related to any variables of interest in the T2D group. Additionally, kilojoule intake was not correlated with any variable of interest. However, it was decided that kilojoule intake would be included as a covariate for statistical analyses of both groups, to control for differences in meal composition between participants. Partial Pearson's correlation analyses were performed, using age, BMI, and kilojoule intake as covariates. In instances where at least two different HRV measures were significantly correlated with a single BGL measure, a multiple linear regression was conducted. This included postprandial BGL in the sample without diabetes, fasting BGL in the diabetes group, and postprandial BGL in the diabetes group.

3.3.3 Heart Rate Variability Associations with Fasting Blood Glucose Level

In the UTS Pilot Study, the aim was to investigate acute fluctuations in HRV measures and BGL, which involved statistical comparison of these variables between three time points. In the UTS T2D Study, the aim was to investigate associations between HRV measures and BGL to demonstrate the predictive applications of HRV measures. As such, differences between fasting and postprandial HRV measures were not investigated, as this was investigated in the Pilot Study. The main statistical analysis of the present study involved correlation analyses of the four HRV measures and fasting and postprandial BGL. In cases where at least two HRV measures were significantly correlated with a single BGL variable, whether it was fasting or postprandial, a multiple regression analysis was conducted to further explore the predictive value of the HRV measures. As discussed in Section 1.9.1.1, assessments of autonomic activity may provide meaningful predictions of BGL fluctuations, or even indicate if current BGL is high or low. The intention of this section is to discuss how the findings of this study add to these emerging applications of HRV and to compare these results with previous literature. Firstly, this section will explore the observed associations between HRV measures and fasting BGL in people without diabetes.

3.3.3.1 Non-Diabetes Sample

The results of the UTS T2D Study indicate no significant correlation between HRV and fasting BGL in people without diabetes, and this is consistent with similar small-scale studies. However, this does not align with the findings of larger studies in this area. A previous study conducted by

Rothberg and colleagues (2016) observed no significant associations between frequencydomain HRV measures and fasting BGL in people without diabetes.³⁸³ The follow-up to this study, conducted by Jarman and colleagues (2021) in an Honours year, also found no significant relationship between HRV and fasting BGL in people without diabetes.³⁸⁹ These studies were similar in design, and recruited relatively small numbers of participants. In larger studies, such as one conducted by Singh and colleagues (2000) on n=1,179 adults (mean age 47.4 years), fasting BGL was inversely correlated with LF power, HF power, and LF:HF.³⁹² However, in this study, the data from the n=1,179 participants was pooled with n=56 adults with impaired fasting glucose and n=84 people with diabetes for the statistical analysis. Including participants with suboptimal ranges of BGL may have strengthened the correlation analysis. Two other large-scale studies report similar findings but were also different in their approach to the analysis. For example, Jarczok and colleagues (2013) concluded that both LF power and HF power were inversely related to fasting BGL,³⁹⁵ however their sample of n=2,441 adults was predominately male (75%), and there are significant differences in autonomic activity between men and women, as discussed. The dominance of male subjects may have skewed the results. Additionally, Stein and colleagues (2007) found that fasting BGL was negatively related to LF power and HF power in adults (n=1089),³⁹⁴ however their sample was mostly elderly men (mean age 72 years). Though there were limitations to the designs of these studies, as well as differences in how they approached methodology and analysis, it is likely that HRV measures are significantly associated with fasting BGL in people without chronic illness. As stated, the results of the UTS T2D Study did not align with the findings of larger studies in this area. This may be due to the relatively younger sample that was recruited in the UTS T2D Study as well as the decision to not pool the participants without diabetes with the diabetes participants in the statistical analysis.

3.3.3.2 Diabetes Sample

In the UTS T2D Study, fasting BGL was negatively associated with fasting LF power (r = -0.63, p < 0.01), HF power (r = -0.57, p < 0.01), and total power (r = -0.66, p < 0.01). Though Rothberg and colleagues did not report any significant correlations between HRV and fasting BGL in people without chronic illness, they identified numerous correlations in people with diabetes.³⁸³ In their pooled analysis of T1D and T2D participants (n=32), fasting BGL was correlated with HF power (r = -0.46, p = 0.01), LF:HF (r = 0.50, p < 0.01), and total power (r = -0.40, p = 0.02). Fasting BGL was not significantly associated with LF power (r = -0.29, p = 0.12). In an analysis of just their n=11 participants with T2D, fasting BGL was only significantly correlated with HF power (r = -0.64, p = 0.04) and LF:HF (r = 0.78, p = 0.01).³⁸³ Early research on vagal tone in

diabetes, conducted by Liao and colleagues (1995), demonstrated that HF power was significantly lower in people with higher fasting BGL, compared to those with lower fasting BGL.⁴⁷³ There is substantial research in the area of fasting BGL in people with T2D, as it is consistently implemented as a baseline for comparing BGL. This inverse relationship between fasting BGL and HRV measures has been confirmed by numerous studies,^{326, 394} and a meta-analysis published by Benichou and colleagues (2018) indicates this correlation is consistent across many clinical studies.⁴⁷⁴ There are two notable discrepancies between the results of Rothberg and colleagues and the UTS T2D Study. Firstly, Rothberg and colleagues (2016) did not observe any significant association between LF power and fasting BGL in their diabetes sample; and secondly, the UTS T2D Study did not observe any relationship of significance between LF:HF and fasting BGL. These are both likely due to low sample power, as the small sample size of the UTS T2D Study and the study conducted by Rothberg and colleagues (2016) may have restricted certain statistical analyses from indicating significance. It is clear from the literature that LF power and LF:HF should be significantly related to fasting BGL.

As more than one predictor variables were significantly associated with fasting BGL in the diabetes group, a multiple regression analysis was performed. Fasting LF power, HF power, and total power accounted for 55% of the variation in fasting BGL in the T2D group ($R^2 = 0.55$). Rothberg and colleagues (2016) demonstrated that fasting HRV measures, including LF:HF and total power, predicted 31% of the variance in fasting BGL in diabetes subjects ($R^2 = 0.31$).³⁸³ Discrepancies in this R-squared value may be related to sample differences, as the Rothberg study pooled participants with T1D (n=21) and T2D (n=11) into a single diabetes group for the regression analysis, and there are meaningful neurophysiological differences between T1D and T2D.

3.3.4 Heart Rate Variability Associations with Postprandial Blood Glucose Level

Currently, there is less interest in the applications of HRV measures in predicting postprandial BGL compared to fasting BGL. This may be related to methodology constraints, as assessing participants in a postprandial state requires more resources and time organisation compared to assessing them in a fasting state. However, there is evidence that postprandial BGL may be equal in importance to fasting BGL in monitoring complications of diabetes, as well as general monitoring of diabetes progression and health. As discussed in Section 1.11, the ability to predict fasting or postprandial BGL using non-invasive means represents an important goal in diabetes research. Therefore, the UTS T2D Study also investigated associations between HRV measures and postprandial BGL, despite the constraints this added to the methodology.

Additionally, the UTS T2D Study was the first study to date to investigate associations between prandial HRV measures and postprandial BGL. Regarding prandial HRV and its predictive qualities, the findings presented in this thesis are novel, as most studies in this area have investigated associations between fasting HRV and BGL. The notion of predicting postprandial BGL using autonomic markers recorded whilst are eating is novel. This is important to consider for the following sections where these findings are discussed.

3.3.4.1 Non-Diabetes Sample

In the UTS T2D Study, there were fewer predictor variables that were significantly correlated with postprandial BGL, and so these three HRV measures accounted for less variance in postprandial BGL. As discussed, the notion that postprandial BGL may be predicted by HRV measures assessed whilst eating, which reflect preprandial autonomic activity, is novel. This study is the first the date to acknowledge these findings and builds on previous literature which indicated the value of HRV measures. Based on previous work conducted in this research unit and published by Jarman and colleagues (2021), postprandial BGL is negatively associated with LF power (r = -0.62, p < 0.01) and total power (r = 0.57, p < 0.01) in subjects without diabetes (n=25).³⁸⁹ In this paper, HRV measures explained 52% of the variance in postprandial BGL. However, these five predictor HRV measures were recorded in different metabolic states, not just fasting. For example, LF power and total power from the fasting assessment, combined with LF power, HF power, and total from the postprandial assessment, accounted for 52% of the variance in postprandial BGL. This may explain why the UTS T2D Study (n=29), which assessed a similar number of participants, demonstrated that fasting total power, prandial total power, and prandial LF:HF accounted for 44% of the variance in postprandial BGL in the sample without diabetes, compared to 52% in the previous study.

3.3.4.2 Diabetes Sample

In the UTS T2D Study, fasting HRV measures were significantly and negatively associated with postprandial BGL, including LF power (r = -0.46, p < 0.05) and total power (r = -0.51, p = 0.03). Additionally, total power from the prandial assessment was also negatively correlated with postprandial BGL (r = -0.48, p = 0.04). Combined, these three HRV measures accounted for 27% of the variance in postprandial BGL in the diabetes group. Contrary to the findings of Rothberg and colleagues (2016), there was no significant correlation with LF:HF. This may be because

LF:HF was only correlated with BGL in the pooled diabetes group in the Rothberg study, and by comparison, there were no T1D participants in the diabetes group for the UTS T2D Study. Additionally, Rothberg and colleagues recorded HRV measures in a fasting state, but not a prandial state.

There is little research on the associations of postprandial BGL and HRV measures, as most studies focus on fasting BGL or focus on the changes in HRV measures after a meal. Myakina, Klimontov & Safarov (2015) monitored HRV and glucose levels in T2D subjects (n=70) for 24 hours by continuous CGM and ECG. Postprandial LF:HF was inversely related with daytime mean BGL (r = -0.29, p < 0.05) and peak BGL (r = -0.33, p < 0.05),⁴⁷⁵ though these glucose measures are not the same as the measures used in the UTS T2D Study. In their 2016 paper, Rothberg and colleagues demonstrated that postprandial BGL was significantly associated with LF:HF (r = 0.44, p =0.01) in their pooled diabetes sample (n=32), and in their sample of just n=11 T2D participants, postprandial BGL was associated with HF power (r = -0.64, p =0.04).³⁸³ The authors concluded that the physiological differences between T1D and T2D were cause for the differences in these correlation analyses. To reiterate, where prandial HRV measures were demonstrated to be associated with postprandial BGL, these findings are novel.

3.3.5 General Relationship of Blood Glucose Level and Autonomic Activity

The literature has provided substantial evidence that there is an association between HRV and BGL. There are consistent findings in the literature that when these two variables are recorded in a large group of people at a single time-point, they are significantly and inversely correlated. This is true even when HRV and BGL are assessed in different states of metabolism, and there are several factors which contribute to this correlation. Physical inactivity and a sedentary lifestyle are key risk factors for T2D and impaired fasting BGL,^{476, 477} and lack of exercise is associated with lower HRV measures and poorer autonomic tone.⁴⁷⁸ Therefore, in crosssectional studies, people with lower HRV measures tend to have higher BGL, and physical fitness is a key confounder as it predicts both high BGL and low HRV.⁴⁷⁹ Another key factor involved is that chronically high BGL, a landmark of diabetes, leads to neurodegeneration of the nerves of the autonomic nervous system and gradual loss of autonomic tone. As discussed in Section 1.5.2.3, autonomic neuropathy is a long-term complication of diabetes. Consequently, in groups of people with chronically high BGL, there is a tendency for autonomic neuropathy to be more prevalent compared to groups with lower or optimal ranges of BGL, and thus HRV measures are lower in people with higher BGL. However, these factors only explain why HRV and BGL are related when comparing groups of people with significantly different lifestyles and with significantly different levels of blood glucose. This does not explain what was observed in the UTS T2D Study. HRV measures and BGL were still correlated in groups of people without diabetes, where differences in physical fitness are smaller and the effects of autonomic neuropathy are likely non-existent. These major confounders were accounted for in the UTS T2D Study. As such, the next section will explore the physiological and neural foundations of the relationship between HRV measures and BGL, as a basis for understanding why HRV and BGL are related even in specific groups of people.

3.3.5.1 Autonomic Response to Food Intake

The role of the autonomic nervous system in the regulation of glucose and fat metabolism is not well understood,¹²⁹ however it is recognised that autonomic functions are important for the metabolism of food and drink into glucose, and that some autonomic functions may precede changes in BGL. The physiological principle behind this involves insulin. Following the consumption of a meal, the absorption of nutrients into the blood via the gastrointestinal tract is associated with a small increase in plasma insulin, even before there is a rise in blood glucose.³⁹⁰ This is part of the pre-absorptive, or cephalic phase. Activation of the ANS during both the pre-absorptive and absorptive phases of insulin is important in determining postprandial insulin activity.⁴⁸⁰ As such, changes in HRV may precede the release of insulin.⁴⁸¹ This is relevant because insulin enacts a stimulatory effect on the sympathetic nervous system,^{481, 482} which can be measured by LF power. This may explain why various studies, including the UTS T2D Study, show that LF power is inversely proportional to BGL in subjects without diabetes with optimal autonomic function.^{392, 394, 395}

3.3.6 Strengths and Limitations

The results of this study indicate that HRV measures may estimate BGL in people without diabetes or people with diabetes. However, this conclusion does not have broad applications. For example, due to the study design, it cannot be concluded based on this research whether an individual could predict their BGL at any point in the day based on their HRV data. It may only be possible to estimate postprandial BGL, and even then, this may require HRV data to be recorded at a specific time point. Many researchers hope to demonstrate that autonomic activity, as measured by HRV, may predict BGL throughout the day, allowing for a continuous and non-invasive measure of BGL in real-time. However, this has yet to be demonstrated as a superior alternative to current glucose monitoring, and HRV measures continue to lack clinical relevance in this specific area. Regardless, this thesis presents novel findings. These findings

alone do not justify the use of HRV in clinical settings, such as estimating BGL in people with diabetes. However, these findings add to the growing evidence that HRV may have clinical applications in diabetes, as well as present novel findings that had yet to be provided in the literature. The scope of this PhD thesis was only to demonstrate that BGL may be estimated in certain important metabolic states, such as fasting or postprandial, using simple non-invasive technology and gold standard HRV processing techniques. The aims and hypotheses were sufficiently addressed through the methodology of this study, and it is important that the scope of this findings and the limitations of the study be considered.

In Section 1.9, the potential for machine learning to improve current glucose monitoring devices was discussed. In this study, regression analysis was used, which is a basic form of machine learning. The use of linear regression in this PhD research represents a lack of risk prediction. In clinical studies, linear regression often shows no significant results, and future research in this area should consider more rigorous statistical methods instead. However, these is some contention on this in the literature. Some authors contend that, in terms of predicting BGL, classic autoregression techniques perform worse than machine learning models, including machine-learning-based regression models and deep learning models.^{331, 483} However, in their review article, Xie and Wang (2020) observed no significant advantage between the two types of method in predicting BGL.⁴⁸⁴ There may be a bias in how authors promote the value of their predictive models. Rodríguez-Rodríguez and colleagues (2019) suggest that accurate glucose prediction can only be achieved by monitoring glucose levels over a short period of time and using a low sampling frequency.⁴⁸⁵ However, they believe that this makes wearable sensors ideal for glucose prediction as they have limited requirements for hardware. The UTS T2D Study was also more interested in how machine learning can be integrated with SMBG, a topic which has comparatively little attention in the literature, potentially because the nature of SMBG makes it difficult to collect the large number of data points required to train a predictive model.486

A limitation of the UTS T2D Study was that certain confounders were not controlled for due to the scope of the study. As with the first experiment, the exclusion criteria for daily alcoholic beverage consumption of no more than 10 standard drinks per day was substantially higher than the regular daily intake of 2 standard drinks per day. This is a limitation since excessive alcohol consumption has been associated with autonomic neuropathy. Future studies should consider excluding subjects with a duration of diabetes (time since date of diagnosis) over 5 years, since longer duration is correlated with autonomic neuropathy and a more delayed prandial response. Since participants were not assessed specifically for autonomic neuropathy, an exclusion criteria could have been added for the T2D group excluding subjects with any known peripheral neuropathy or previous foot ulcer as signs of existing neuropathy. Autonomic neuropathy causes significant changes in HRV, even when it is not related to diabetes progression. In this study, participants did not undergo a physical examination or screening prior to the study, which may have been useful in identifying autonomic neuropathy that participants were unaware of. Instead, participants were asked to provide details about any current comorbid conditions and any current medications, including dosage. Certain medications were prohibited and were grounds for exclusion from the study if participants were currently taking them (see Section 3.1.2), however most medications were allowed in the study to allow for real-world data to be recorded. Some studies suggests that subjects with T2D should abstain from medications that affect BGL or cardiovascular function prior to study participation, including antihypertensives, as they are confounders.^{146, 419} Since time to peak prandial glucose is delayed in people with suboptimal glycaemia, this study design could have also benefited from excluding T2D subjects with HbA_{1c} over 7.5%. According to a review paper, there are several studies which indicate BP is related to HRV measures, and on a minute-tominute basis this relationship is due to the autonomic nervous system exercising homeostatic control over the cardiorespiratory cycle.487 For example, increased metabolic demand due to walking up a staircase may be associated with both a change in BP and HRV measures. Future studies should attempt to control for BP, such as in the statistical analysis or by restricting the movement of subjects during the study assessments, so that any correlations identified between HRV measures and BGL may be less confounded by BP, which may influence the changes in HRV.

It was not within the scope of this study to control for every major confounder variable, as it was not the aim to demonstrate that changes in BGL are strictly related to autonomic activity. The aim was to investigate whether the relationship between HRV measures and BGL was sufficient for the development of a regression model, which may in future be used to predict BGL through non-invasive means. An algorithm or model that predicts BGL using HRV measures may need to consider the effect of medications on physiological data rather than ignore it or attempt to remove that effect from a sample entirely. As with most clinical samples, there were various comorbid conditions and medications involved which would have affected levels of HRV and BGL observed in the present study. However, this is reflective of a real-world scenario. People living with diagnosed diabetes are likely to be medicated and have multiple comorbid conditions. As such, only basic variables such as age, BMI, kilojoule intake, and sex were controlled for in this study as these can be easily entered into a device for BGL prediction, and a real-world application of HRV would ideally incorporate these variables for added accuracy.

Participants were free to eat whatever they wanted to, as it was the intention of the study design to allow participants to resume their routine as close as possible, however there is only so much that statistics can do to control for these differences in the analysis. In this study, kilojoule intake was included as a covariate in the statistical analysis, and participants were advised to avoid certain foods such as tea and cinnamon which have known anti-glycaemic properties. However, this is an imperfect measure of controlling for significant differences in participant's meals, and a robust study design should ultimately attempt to standardize these meals, if meals are part of the study design. Though it was the intention of this study to observe subjects whilst they consumed meals of their choice, which would be considered 'regular' by their own standards, the differences in meal portions between subjects was not controlled for. Only the kilojoule intake of the meals was accounted for, and differences in the distention of subject's stomachs, due to consuming meals that may have been relatively larger compared to the group average, may have confounded autonomic responses since greater stomach distention would lead to greater sympathetic nerve activity.⁴⁸⁸ Future studies should aim to standardise meal portions. Participants were also free to leave the UTS laboratory between recordings, which would reflect a regular routine as even office workers would expect some levels of movement. However, even low levels of activity, such as walking for a few minutes, increase metabolic demand and may lead to reduced glycaemia as well as changes in HRV. Future research should consider restricting the activity of participants whilst they are being monitored.

One of the strengths of the methodology was the implementation of two BGL assessments for the determination of postprandial BGL. CGM could not be used in this study due to resourcing issues, as CGM is not standard of care for most people with T2D, and the research unit lacked the proper ethics approval and qualified staff to administer a CGM system to either healthy subjects or subjects with T2D. Without CGM, it is difficult to capture the peak level of blood glucose after a meal, and even with CGM, observed recordings may lag behind true glucose levels. In this study, BGL was assessed by invasive blood glucose assessment in participants without diabetes 40 minutes and 50 minutes after eating, and the higher of these two values was used to provide an accurate assessment of postprandial BGL. The timing of the BGL assessments was the same with T2D participants, as peak BGL was expected to be achieved at roughly the same time in both groups. This was important for consistency, as it can be concluded that HRV measures may estimate BGL at roughly the same time in individuals regardless of whether they have diabetes or not. However, according to the literature, the glycaemic response to food may be complicated by the presence of metabolic diseases. In the case of T2D, this effect is difficult to quantify. Kanaley and colleagues (n=42) contend that BGL peaks later in people with T2D compared to people without diabetes, usually at least 60 minutes after food ingestion.⁴¹⁹ However, research conducted by Esposito and colleagues (n=644) showed that BGL peaks 40 minutes after consuming a mixed meal in cases of T2D, which is the same time observed in groups without diabetes.⁴⁴⁶ Given that Esposito and colleagues sampled a much larger group and from a wider demographic, and that the effect of T2D on the postprandial response is difficult to quantify, it may be accepted that BGL peaks after a meal at roughly the same time in people with and people with T2D. Even in cases where the presence of diabetes may have affected metabolic function, it is possible that the difference in time required for BGL to peak was accounted for by the use of two postprandial BGL assessments, one after the other. Future studies may consider the use of CGM to capture true postprandial BGL, as the time to peak postprandial BGL may be delayed in people with T2D and other comorbid conditions. From this study, it can be concluded that HRV measures may predict postprandial BGL in T2D or people without T2D 40-50 minutes after eating.

4.1 Methods

T1D and T2D are more different than they are similar (Section 1.4), and glucose monitoring technology should be tailored to the needs of each group separately. As such, the RNSH T1D Study investigated how HRV measures may be used in T1D as a non-invasive alternative to current glucose monitoring. Because people with T1D are dependent on exogenous insulin to survive, daily glucose levels fluctuate greatly, and this increases the incidence of glucose excursions – events where BGL deviates greatly from baseline levels. Glucose excursions in the high range may cause mild symptoms such as thirst and fatigue in the short-term, and complications arise in the long-term if these excursions go unmanaged. Glucose excursions in the low range are usually referred to as hypoglycaemia, and these may quickly become life-threatening. In particular, the detection and early intervention of hypoglycaemia is a higher priority in cases of T1D because it is more common compared to T2D. As such, the specific aim of this study was to observe changes in HRV during and preceding glycaemic excursions, to justify the use of HRV measures as markers of suboptimal BGL.

4.1.1 Subjects

According to the Global HAT Study (n=27,585), hypoglycaemic events occur more often in people with T1D (83.0%) compared to T2D (46.5%). Rates of any hypoglycaemic event and any severe event in T1D were 73.3 and 4.9 events per patient-year, respectively. For people with T2D, occurrence of hypoglycaemia was 19.3 per year for any event and 2.5 per year for severe events.⁸⁶ There was a concern that this rate of hypoglycaemia in T1D would be difficult to capture without long periods of recording or observation. To improve the number of observed glycaemic excursions in this study, recruitment prioritised those who were at higher risk of experiencing glycaemic excursions. Many of these individuals had undergone recent changes in their medication or were experiencing problems with their CGM systems, resulting in more frequent glucose fluctuations. It was important that individuals experienced glycaemic excursions whilst they were being monitoring so that the relevant data could be collected.

Recruitment of people with T1D started at RNSH in collaboration with the Department of Endocrinology, through endocrinologists and study coordinators working at the hospital. Approval was gained from both UTS HREC (2014000110) as well as the Northern Sydney Local

Health District HREC (LNR/14/HAWKE/174) (Site Specific Assessment reference: LNRSSA/16/HAWKE/8). The researcher attended the hospital Department of Endocrinology to recruit people with T1D. As with the UTS T2D Study, it was difficult to exclude diabetes participants if they were also living with any other lifestyle-related diseases or taking any medications (Section 3.1.2). In particular, people with T1D were not excluded for cardiac autonomic neuropathy, as they were in the previous study, as it does not confound changes in HRV during glycaemic excursions, and these were the main states of interests in this study.489 However, some selection criteria were applied based on recommendations from previous literature. For one, subjects were free of clinically-apparent coronary heart disease or congestive heart failure.⁴⁹⁰ They were also required to be aged between 18 and 69 years old, and were excluded if they were currently pregnant, smoked more than 10 cigarettes per day, consumed more than 10 standard alcoholic beverages per day, or if their BP was too high during the pre or post-study BP recordings (Section 2.1.3). Because glycaemic excursions occur very rarely in people without chronic illness, it was anticipated that little to no meaningful data would be captured by including them in the study. Thus, there was no control group recruited for the RNSH T1D Study. Instead, there was a focus on comparing changes in HRV during glycaemic excursions with individual's baseline HRV measures. An investigation of the relationship between HRV and BGL in individuals without chronic illness was provided in the UTS T2D Study, as appropriate.

4.1.2 Pre-study Requirements

Provided they suited the eligibility criteria, participants were invited to RNSH to discuss the study with the researcher and their endocrinologist. Subjects were not expected to fast or abstain from any medications prior to the study due to concerns this would interfere with their glycaemic health, and this was a greater concern in this study compared to the UTS T2D Study. Participants were asked to continue taking their regular medications and prioritise any medical advice given to them by their doctor or diabetes educator. Participants were advised they could withdraw from the study at any point without reason and were asked to sign two copies of a consent form, one of which they retained. Following this, BP was recorded, and the BP requirements were checked based on UTS HREC guidelines (Section 2.1.3).

4.1.3 Continuous Glucose Monitoring

All subjects who participated in this study were required to use some form of CGM, as this is the gold-standard for collecting data on glycaemic excursions. The devices used in this study were the Medtronic Guardian Connect, which is a CGM system which connects with the insulin pumps such as the MiniMed 640G and 670G (Medtronic Australasia Pty Ltd, Macquarie Park, Australia), and the Dexcom G4 and G5 (Dexcom, Inc., San Diego, Unites States of America). Subjects using the FreeStyle Libre Flash Glucose Monitoring System (Abbott Medical Australia Pty Ltd, Macquarie Park, Australia) were also included in this study, as this device is capable of continuously recording glucose levels, though the data is provided retrospectively and not in real-time, as with the Dexcom and Medtronic systems. All devices were owned by the participants themselves and were operating before the start of the study. Participants were advised to continue operating their devices for the duration of the study, including calibrating the CGM systems with blood glucose assessments where necessary. All subjects knew to calibrate their own systems using their personal lancet devices and glucose test strips and were instructed to continue as per their standard of care. The CGM systems used by participants were made available to the researcher when the participants returned to the hospital so that the glucose data could be downloaded. HbA1c levels were estimated by the CGM systems based on weekly fluctuations in glycaemia.

4.1.4 Recording of R-R Interval Data by Heart Rate Monitor

Due to the unpredictable nature of glycaemic excursions and their infrequent timing, short-term observations of T1D participants were unlikely to capture sufficient samples, even in high-risk cases. Therefore, participants were required to undergo CGM and heart rate recording for a minimum of 24 hours to ensure a reasonable chance of recording HRV during glycaemic excursions. The device used for R-R recording was the FirstBeat BodyGuard 2, and a detailed description of how this was attached to participants can be found in Section 3.1.4. Wearable single-lead devices are now commonly-used for continuous HRV monitoring in outpatient settings.⁴⁹¹ The BodyGuard 2 was attached at the hospital and participants then returned to their regular schedules and work routines whilst being mindful that the device was to remain attached. Instructions were provided on how to remove and reattach the monitor using fresh electrodes provided, so that participants could shower and maintain proper hygiene as usual. Participants were advised that they were to wear the device for as long as comfortable up to a maximum of 72 hours, as additional fresh electrodes would not be provided after this point.

to the researcher. The data was then downloaded and used with Kubios HRV Premium. For details on how HRV measures were extrapolated from the raw R-R interval data, see Section 2.1.5.1 and 2.1.5.2 (UTS T2D Study: Methods).

4.1.5 Meals, Medications, and Physical Activity Diary

Autonomic activity and glycaemia are affected by a range of day-to-day activities and events. To gather important information about participant's regular routines, which they were encouraged to continue,⁴⁹¹ each subject was required to provide details regarding any meals they ate, medications they took, or any physical activity they undertook. Participants were instructed on how to complete the diary and were provided with some examples, such as "1:30pm chicken sandwich with cheese and mayonnaise" and "8:00-8:15am walked to the train station". Mealtimes were useful for identify periods when participant's glucose levels were likely to spike, and physical activity times were useful for identifying periods of HRV data to be excluded for analysis because of the confounding effect of exercise on HRV.

4.1.6 Selection of Glycaemic Excursions and R-R Interval Data

Between 24-72 hours of continuous R-R interval data and glucose data were collected from each participant in this study. Glucose levels are typically measured every five minutes by CGM systems per standardized reporting.⁴⁹² However, it should be noted that the data downloaded by the glucose monitors for this study only provided a glucose reading every 15 minutes, and each glucose data point was timestamped. This provided a reference point for determining the exact time at which glucose excursions occurred. For example, if a participant's BGL dropped below a certain value, it was possible to determine the exact time during the day at which that occurred. Additionally, each R-wave recorded by the heart rate monitors was timestamped. This allowed for HRV epochs to be calculated in the 10 minutes before, during, and after the start every glycaemic excursion, and thus it was possible to observe changes in HRV during a hypoglycaemia or hyperglycaemia event. Noteworthy changes in certain HRV measures may serve as an early indicator of glycaemic excursions.

However, care had to be taken when determining an appropriate threshold for defining these excursions. Although BGL < 4.0 mmol/L and BGL > 7.7 are the most widely used definitions of hypoglycaemia and hyperglycaemia, respectively, these do not represent thresholds for clinically significant glycaemic excursions. In practice, symptoms do not necessarily arise when BGL exceeds either of these thresholds,^{493, 494} even though this does not occur under regular

physiological conditions.⁴⁹⁵ For example, when hypoglycaemia episodes are defined as BGL < 4.0 mmol/L, less than 40% of those episodes are symptomatic. When hypoglycaemia is defined by a more clinically relevant threshold, such as < 3.0 mmol/L, roughly 70% of subjects experience autonomic symptoms, and only 30% do not.⁴⁹⁴ This is relevant to this study because in Section 1.11: Basis for Research, it was established that HRV measures may be predictive of hypoglycaemic or hyperglycaemia due to the distinct autonomic response which occurs during glycaemic excursions. If this response is absent from most of the glycaemic events that are recorded, then any observed changes in HRV are unlikely to be related to the glycaemic excursion. Therefore, it was important to utilise a more stringent and clinically relevant definition of hypoglycaemia when selecting excursions to analyse.

Based on recommendations from multiple clinical trials, hypoglycaemia was defined as BGL < 3.0 mmol/L and hyperglycaemia was defined as BGL $\geq 12.0 \text{ mmol/L}$.⁴⁹⁶⁻⁴⁹⁸ In addition to this, for a hypoglycaemic event to be included for analysis in this study, it could not be followed by BGL returning to 4.5 mmol/L or more within 10 minutes.⁴⁹⁹ Glycaemic excursions were only selected for analysis if the entire event was captured by both the glucose monitor and the heart rate monitor. For example, for some participants there was over two weeks of glucose data to be downloaded from the glucose monitors, but only 48 hours of R-R interval data from the heart rate monitors. Thus, only glucose data that overlapped with the 48 hours of R-R interval data was relevant to be analysed.

4.1.7 Measures of Heart Rate Variability

The measures of HRV used in this study included low frequency (LF) power, high frequency (HF) power, the low to high frequency ratio (LF:HF), total power, and normalised low frequency (LFnu). LFnu is more strongly correlated with sympathetic nervous system activity compared to LF power,⁵⁰⁰ though ultimately both sympathetic and parasympathetic activity are reflected in LF power and LFnu. LFnu is calculated from the formula:

$$LFnu = 100 \times \left(\frac{LF}{LF + HF}\right)$$

Where the derived value is a percentage, or proportion of LF power to HF power. This is calculated differently to LF:HF, which is simply LF power divided by HF power. The high frequency counterpart to LFnu, known as normalised high frequency power (HFnu), is calculated as follows:

$$HFnu = 100 \times \left(\frac{HF}{LF + HF}\right)$$
As such, the sum of LFnu and HFnu will always be 100%, as they are algebraic inverses. Because of this, statistical analysis involving LFnu and HFnu produces identical, but inverse values.⁵⁰¹ For example, correlating BGL against LFnu and then against HFnu will produce the same p-value and correlation coefficients, however the sign of the 'r' coefficient will be reversed, as these variables are inverses. To avoid redundancies, only LFnu and not HFnu will be referred to in the statistical analysis.

Additionally, time-domain measures of HRV were used in this study as long-term recordings were available due to continuous, at-home heart rate recording. These included the root mean square of successive RR interval differences (RMSSD), and the standard deviation of NN intervals (SDNN). RMSSD is closely associated with HF power and reflects parasympathetic control of HRV. SDNN measures sympathetic and parasympathetic activity on HRV, and is closely related to LF power.⁵⁰²

4.1.8 Statistical Analysis

For a preliminary analysis, correlations between BGL and various measures of HRV measures were investigated in only the first participant of this study. These correlation analyses, combined with scatter plots, was performed in SPSS version 22.0 (IBM SPSS Statistics, USA) to investigate a possible trend in HRV measures when BGL was in the low range or in the high range. Given the large number of data points collected over the 24 hours for each participant, there were no concerns about sample power when using Pearson's correlations to identify significant (p<0.05) correlations between glucose level and HRV in a single participant. This statistical analysis aimed to continue previous research and the results are provided only for a comprehensive analysis of the relationship between HRV and BGL. This preliminary analysis was not important for the main aim of this research.

The main analysis in this study aimed to identify changes in HRV measures during a hypoglycaemic event, as stated in the aims and hypotheses. Non-overlapping HRV epochs were calculated from 10-minute samples, and these were sorted into three independent groups. These were 'Pre-hypo, 'Hypo, and 'Post-hypo, which refer to 10-minute epochs recorded 10 minutes prior to the start of the hypoglycaemic event, 10 minutes during, and 10 minutes after, respectively. These three groups represent a longitudinal approach, as the same set of HRV measures were calculated from the same sample at three different time points. Longitudinal studies typically have higher statistical power compared to cross-sectional studies.⁵⁰³ As such, a one-way analysis of variance (ANOVA) with repeated measures was used to determine any significant (p<0.05) differences between the related means: Pre-hypo, Hypo, and Post-hypo HRV

measures. A significant difference (p<0.05) would indicate a change in HRV over the course of the hypoglycaemia excursion. To ensure that the assumptions of the repeated measures ANOVA were not violated, the data set was checked for outliers, normal distribution, and sphericity. Shapiro-Wilk Test, appropriate for small sample sizes (n<50), was used to confirm that variables were normally distributed. Normal Q-Q plots showed that the data points were linearly organised, which also confirmed normal distribution. Mauchly's test of sphericity was used to indicate any violations of the assumption of sphericity in the data and a Greenhouse-Geisser correction was applied in cases where the assumption of sphericity was violated.⁵⁰⁴ These steps were repeated for the hyperglycaemic events that were observed, and the HRV means were labelled as 'Pre-hyper', 'Hyper', and 'Post-hyper'.

With reference to Cohen's "A Power Primer", the minimum sample size (n) required to achieve 0.8 power for this study was determined. For an ANOVA comparing the means of three related groups (Pre-hypo, Hypo, and Post-hypo), with a significance (α) of 0.05 and a large effect size (r=0.5), it was necessary to observe at least n=21 hypoglycaemia and n=21 hyperglycaemia events to provide sufficient power.⁵⁰⁵ It was estimated that it would be necessary to recruit 15 participants with T1D to satisfy these requirements.

4.2 Results

The RNSH T1D Study aimed to investigate the activity of HRV measures during glycaemic excursions, including hypoglycaemia and hyperglycaemia. Both time and frequency-domain measures of HRV were investigated, as data was collected over a 24-hour period. For this study, 15 participants with T1D were recruited, with a sex breakdown of 4 males and 11 females. Participants were aged between 20-61 years. The average duration of the heart rate recordings was 30 hours, with most participants opting to begin recording between 10:00am and 12:30pm. Additional sample information is reported in Table 4.1. HbA_{1c} was measured by the CGM systems using weekly estimations of BGL, and the mean was 6.9%. Duration of diabetes for each subject was not recorded by the researchers in error and due to logistical problems created by the COVID-19 infection control measures implemented in March 2020.

Table 4.1 Means and standard deviations for demographic and physiological features of the sample (n=15). Data is presented as mean ± standard deviation. Body mass index was calculated from weight (in kilograms) divided by height (in metres) squared. Blood pressure was determined using the average of three left-arm recordings. HbA_{1c} was estimated by the CGM systems. **BMI** = Body mass index, **BP** = Blood pressure, **kg/m**² = Kilogram per square metre, **mmHg** = Millimetres of mercury, **mmol/L** = Millimoles per litre.

Variable	Mean ± standard deviation		
Age (years)	36 ± 16		
BMI (kg/m ²)	25 ± 3		
Systolic BP (mmHg)	127 ± 14		
Diastolic BP (mmHg)	77 ± 7		
Estimated HbA1c (%)	6.9 ± 0.9		

4.2.1 Preliminary Correlation Analysis

A preliminary correlation analysis was conducted on the first participant recruited in this study to investigate a potential relationship between HRV measures and BGL as the two variables fluctuated over a 24-hour period. A Pearson's correlation determined there was no significant association between BGL and any HRV measure, including LF power, HF power, total power, LF:HF, LFnu, HFnu, RMSSD, SDNN, or even mean heart rate. Scatter plots of these variables of interest as they fluctuated over the course of the 24-hour monitoring period were equally distributed (Figure 4.1). This preliminary analysis was not important to the aim of this study. However, the results of this analysis are shown here to provide some noteworthy information.



Figure 4.1 Scatter plots of glucose level versus various heart rate variability measures for Participant 01 (n=96).

Figure 4.1 visualises the relationship between HRV measures and BGL as variables fluctuated in a single participant over the course of the study period. Each independent variable (x-axis) was plotted against the dependant variable, BGL, on scatter plots with a linear trendline representing the R² value. HRV measures were calculated from 15-minute epochs and were matched with each glucose data point measured at the same time. Glucose levels were recorded once every 15 minutes. Over 24-hours, this provided n=96 data points. All values were taken from a single participant. Note that the plots of LFnu and HFnu are symmetrical and the R² values are identical, as they are algebraic inverses. **BGL** = Blood glucose level, **bpm** = Beats per minute, **HF** = High frequency, **HFnu** = Normalised high frequency, **LF** = Low frequency, **LF:HF** = low to high frequency ratio, **LFnu** = Normalised low frequency, **mmol/L** = Millimoles per litre, **ms** = Milliseconds **ms**² = Milliseconds squared, **RMSSD** = Root mean square of successive RR interval differences, **SDNN** = Standard deviation of NN intervals.

4.2.2 Hypoglycaemia

A review of the literature was used to establish a threshold for defining hypoglycaemia, which was BGL < 3.0 mmol/L. Based on this criterion, n=22 hypoglycaemia events were observed over the course of 450 hours of subject activity, which satisfied the requirements of the power analysis. For each instance of hypoglycaemia, three 10-minute epochs of R-R interval data were processed into HRV measures, providing information on HRV at the following points: 10 minutes prior to the event ('Pre-hypo'), at the start of the event ('Hypo'), and 10 minutes after the event ('Post-hypo'). Initially, trends in HRV measures were investigated before and after some of these hypoglycaemia events by visual inspection of line graphs. Three of these line graphs are shown in Figure 4.2. These show HRV activity 30 minutes before and after three instances of hypoglycaemia, each occurring in a different participant. As there were no obvious trends seen in this initial analysis, this was followed up with a statistical analysis (n=22).



Figure 4.2 Change in heart rate variability before and after hypoglycaemia in three events.

Figure 4.2 shows heart rate variability before and after three different hypoglycaemia events experienced by three different participants. For each individual figure, n=1. The black dotted line indicates the start of the hypoglycaemia, measured as blood glucose < 3.0 mmol/L, and HRV data is presented 30 minutes before and after the start of the event. As seen in Hypo Event #01, the event started at 3:45pm. **Hypo** = Hypoglycaemia, **ms**² = Milliseconds squared.

For the statistical analysis, the means of seven HRV measures were calculated for three HRV epochs – Pre-hypo, Hypo, and Post-hypo – from the n=22 instances of hypoglycaemia. These HRV measures were LF power, HF power, total power, LF:HF, normalised LF power, SDNN, and RMSSD. A one-way ANOVA with repeated measures determined if there were any significant differences between mean Pre-hypo, mean Hypo, and mean Post-hypo for each of the seven HRV measures (Table 4.2). Shapiro-Wilk test confirmed that all variables analysed were normally distributed (p>0.05) at all levels, and normal Q-Q plots were inspected manually to ensure data points were linearly organised, which also confirmed normal distribution. As with the previous studies, and as is common in HRV studies, LF power, HF power, and total power were natural log transformed to ensure normal distribution, as they are typically skewed without log transformation.^{146, 423, 424}

Table 4.2 Mean heart rate variability measures (n=22) at 10 minutes prior, during, and 10 minutes after hypoglycaemia. All the measures presented in this table are frequency-domain, except for SDNN and RMSSD, which are time-domain. The analysis of variance determined there was no significance difference between any of the repeated measures for all variables. **ANOVA** = Analysis of variance, **HF** = High frequency, **Hypo** = Hypoglycaemia, **LF** = Low frequency, **LF:HF** = low to high frequency ratio, **LFnu** = Normalised low frequency power, **ms** = Milliseconds, **ms**² = Milliseconds squared, **NS** = Non-significant.

Variable	Pre-hypo	Нуро	Post-hypo	ANOVA
n	22	22	22	-
LF power (ms ²)	6.2 ± 1.4	6.2 ± 1.2	6.1 ± 1.2	NS
HF power (ms ²)	4.9 ± 1.5	4.9 ± 1.5	4.9 ± 1.6	NS
Total power (ms ²)	6.6 ± 1.4	6.6 ± 1.2	6.6 ± 1.2	NS
LF:HF	4.4 ± 2.9	4.7 ± 3.4	4.5 ± 3.3	NS
LFnu (%)	76 ± 11	77 ± 12	75 ± 14	NS
SDNN (ms)	35 ± 20	33 ± 16	34 ± 19	NS
RMSSD (ms)	27 ± 23	23 ± 16	25 ± 19	NS

A graphical comparison of the first three frequency-domain HRV measures, including mean LF power, HF power, and total power, is presented in Figure 4.3. Mauchly's test of sphericity showed no violation of the assumption of sphericity for the LF power data set, $\chi^2(2) = 0.649$, p = 0.72, or the HF power data set, $\chi^2(2) = 1.733$, p = 0.42. or the total power data set, $\chi^2(2) = 0.635$, p = 0.73. As such, Greenhouse-Geisser correction was not applied for any of the following one-way ANOVA with repeated measures. The ANOVA revealed that there was no significant

difference between the three time points for LF power (F(2, 42) = 0.023, p = 0.98), HF power (F(2, 42) = 0.091, p = 0.91), or total power (F(2, 42) = 0.001, p = 1.00).



Figure 4.3 Comparison of mean low frequency, high frequency, and total power at three time points during hypoglycaemia (n=22).

Figure 4.3 shows how frequency-domain measures compare before, during, and after hypoglycaemia. Data is presented as means with error bars as standard deviations. Though LF power, HF power, and total power are presented on the same axis, it is not appropriate to compare values between different measures of heart rate variability. Only differences between the three time points for each measure were investigated. **HF** = High frequency, **LF** = Low frequency, **ms**² = Milliseconds squared.

A visual representation of the other two frequency-domain HRV measures, LF:HF and normalised LF power, is shown in Figure 4.4. According to Mauchly's test, there was no violation of sphericity for the LF:HF data set, $\chi^2(2) = 0.306$, p = 0.86, or the normalised LF power data set, $\chi^2(2) = 1.709$, p = 0.43. As such, a Greenhouse-Geisser correction was not applied. A repeated measures ANOVA revealed that mean LF:HF was not significantly different across the three time points (F(2, 42) = 0.148, p = 0.86), and neither was mean normalised LF power (F(2, 42) = 0.316, p = 0.73).

■ Pre-hypo ■ Hypo 🖾 Post-hypo



Figure 4.4 Comparison of mean low to high frequency ratio and normalised low frequency power at three time points during hypoglycaemia (n=22).

Figure 4.4 illustrates the change in LF:HF and normalised LF power before, during, and after hypoglycaemia. Errors bars represent standard deviations, and each bar indicates a mean value. LF:HF is represented without units, as it is a ratio of LF to HF power. Normalised LF power is presented as a percentage as it indicates the power of LF relative to HF. **LF** = Low frequency, **LF:HF** = low to high frequency ratio.

For the time-domain measures, including SDNN and RMSSD, means and standard deviations are compared in Figure 4.5. Mauchly's test of sphericity showed no violation of the assumption of sphericity for SDNN, $\chi^2(2) = 3.252$, p = 0.20, and so a one-way repeated measures ANOVA was conducted without Greenhouse-Geisser correction. The ANOVA determined that mean SDNN was not significantly different between the three time points (F(2, 42) = 0.438, p = 0.65). There was a violation of the assumption of sphericity for RMSSD, $\chi^2(2) = 22.031$, p < 0.05, and so a repeated measures ANOVA was conducted with Greenhouse-Geisser correction. The Greenhouse-Geisser correction affects the degrees of freedom applied in the analysis, but not the partitioning of Sum of Squares of the F-statistic. This statistical analysis demonstrated that mean RMSSD was not significantly different (F(1.199, 25.185) = 0.741, p = 0.42).

Figure 4.5 Comparison of time-domain measures at three time points during hypogylcaemia (n=22).



Figure 4.5 visualises differences in mean SDNN and RMSSD before, during, and after hypoglycaemia. Data is presented as means with error bars as standard deviations. Time-domain measures of HRV such as these are measured in milliseconds. As with frequency-domain measures, differences between mean SDDN and mean RMSSD were not investigated. **RMSSD** = Root mean square of successive RR interval differences, **SDNN** = Standard deviation of NN intervals.

4.2.3 Hyperglycaemia

In this study, hyperglycaemia was defined as BGL \geq 12.0 mmol/L. Over the course of 450 hours of recording participants, n=33 instances of hyperglycaemia were captured, based on this criterion. As with hypoglycaemia, it was investigated whether there were any changes in HRV measures over the course of many hyperglycaemia events. HRV measures were calculated by the same methods, and the three HRV epochs were labelled similarly as Pre-hyper, Hyper, and Post-hyper, referring to HRV epochs that occurred 10 minutes before, during, and 10 minutes after the start of a hyperglycaemia event, respectively. For an initial analysis, line graphs were used on three instances of hyperglycaemia to inspect changes in HRV before and after the event starts (Figure 4.6).



Figure 4.6 Change in heart rate variability before and after hyperglycaemia in three events.

Figure 4.6 highlights the changes in low frequency power and high frequency power during individual hyperglycaemia events. Each of these events occurred in different participants, and for each individual graph, n=1. The black dotted line marks the start of the hyperglycaemia, when blood glucose rose above 12.0 mmol/L, and HRV data is provided for 30 minutes before and after this event start. **Hyper** = Hyperglycaemia, **ms**² = Milliseconds squared.

Following this, a statistical analysis was conducted on the means of the seven HRV measures investigated in this study. As with the hypoglycaemia events, a repeated measures ANOVA determined any significant changes in LF power between Pre-Hyper, Hyper, and Post-hyper means calculated from the n=33 hyperglycaemia events. This was then repeated for the other HRV measures, including HF power, total power, LF:HF, normalised LF power, SDNN, and RMSSD. The means and standard deviations of these HRV measures at each of the 10-minute epochs are provided in Table 4.3.

Table 4.3 Mean heart rate variability measures (n=33) at 10 minutes prior, during, and 10 minutes after hyperglycaemia. All the measures presented in this table are frequency-domain, except for SDNN and RMSSD, which are time-domain. The analysis of variance determined there was no significance difference between any of the repeated measures for all variables. **ANOVA** = Analysis of variance, **HF** = High frequency, **Hyper** = Hyperglycaemia, **LF** = Low frequency, **LF:HF** = low to high frequency ratio, **LFnu** = Normalised low frequency power, **ms** = Milliseconds, **ms**² = Milliseconds squared, **NS** = Non-significant.

Variable	Pre-hyper	Hyper	Post-hyper	ANOVA
n	33	33	33	-
LF power (ms ²)	6.1 ± 1.2	6.2 ± 0.9	6.0 ± 1.2	NS
HF power (ms ²)	4.9 ± 1.6	5.2 ± 1.3	5.0 ± 1.4	NS
Total power (ms ²)	6.6 ± 1.2	6.7 ± 0.9	6.5 ± 1.2	NS
LF:HF	4.2 ± 2.9	3.8 ± 2.7	3.6 ± 2.7	NS
LFnu (%)	74 ± 15	71 ± 17	71 ± 17	NS
SDNN (ms)	34 ± 19	33 ± 15	31 ± 17	NS
RMSSD (ms)	26 ± 21	27 ± 18	25 ± 21	NS

Figure 4.7 visualises the differences in LF power, HF power, and total power. According to Mauchly's test of sphericity, the HF power data set violated the assumption of sphericity, $\chi^2(2) = 8.066$, p = 0.02. As such, a Greenhouse-Geisser correction was used in conjunction with the repeated measures ANOVA. This determined no significant difference in mean HF power between the three time points (F(1.627, 52.071) = 1.285, p = 0.28). The assumption of sphericity was not violated by either the LF power data set, $\chi^2(2) = 2.134$, p = 0.34, or the total power data set, $\chi^2(2) = 2.781$, p = 0.25. As such, the repeated measures ANOVA was conducted without Greenhouse-Geisser correction. There was no significant difference in LF power (F(2, 64) = 1.335, p = 0.27) or total power (F(2, 64) = 1.322, p = 0.27) between the three time points.

Figure 4.7 Comparison of mean low frequency, high frequency, and total power at three time points during hyperglycaemia (n=33).



Figure 4.7 shows the difference in frequency-domain measures of heart rate variability before, during, and after hyperglycaemia occurred. These measures are presented as means with error bars as standard deviations. Though these measures are presented on the same axis, it is inappropriate to compare values between different measures of heart rate variability. **HF** = High frequency, **LF** = Low frequency, **ms**² = Milliseconds squared.

The change in two other frequency-domain measures of HRV, LF:HF and normalised LF power, is visualised in Figure 4.8. Mauchly's test indicated that the assumption of sphericity was violated in both the LF:HF ($\chi^2(2) = 12.008$, p < 0.01) and normalised LF power ($\chi^2(2) = 8.457$, p = 0.02) data sets. Repeated measures ANOVA with Greenhouse-Geisser correction showed that there was no significant difference in mean LF:HF (F(1.483, 43.003) = 1.488, p = 0.24) or normalised LF power (F(1.615, 51.664) = 1.091, p = 0.33) between the three time points.



Figure 4.8 Comparison of mean low to high frequency ratio and normalised low frequency power at three time points during hyperglycaemia (n=33).

Figure 4.8 demonstrates differences in certain frequency-domain measures of heart rate variability during a hypoglycaemic event. Data is presented as means with error bars as standard deviations. LF:HF is a ratio, and has no units, and normalised LF power was calculated from the relative value of LF power compared to HF power. **LF** = Low frequency, **LF:HF** = low to high frequency ratio.

Analysis was also performed on time-domain measures of HRV during hyperglycaemia, and a graphical representation of the means is shown in Figure 4.9. Mauchly's test of sphericity showed the assumption of sphericity was violated for SDNN, $\chi^2(2) = 12.492$, p < 0.01, and so a one-way repeated measures ANOVA was conducted with Greenhouse-Geisser correction. The ANOVA determined that mean SDNN was not significantly different between the three time points (F(1.502, 48.060) = 0.524, p = 0.55). There was also a violation of the assumption of sphericity for RMSSD, $\chi^2(2) = 6.307$, p < 0.05, and so a repeated measures ANOVA was conducted with Greenhouse-Geisser correction. The different (F(1.681, 52.118) = 1.961, p = 0.16)

Figure 4.9 Comparison of time-domain measures at three time points during hyperglycaemia (n=33).



Figure 4.9 represents the mean activity of time-domain measures at three time points during hyperglycaemia. Errors bars represent standard deviations, and each bar indicates a mean value. Time-domain measures of HRV such as these are measured in milliseconds. Differences between mean SDDN and mean RMSSD were not investigated. **RMSSD** = Root mean square of successive RR interval differences, **SDNN** = Standard deviation of NN intervals.

4.2.4 Summary of Results

Preliminary correlation analysis of the first participant in the study revealed that daily fluctuations in BGL and HRV measures were largely unrelated, and scatter plots of the variables of interest were equally distributed. Line graphs showing HRV activity 30 minutes before and after three instances of hypoglycaemia, each occurring in a different participant, were inspected and no obvious trends were identified. This was followed-up with a statistical analysis. The repeated measures ANOVA determined there was no significant difference in any HRV measure in the 10-minutes before, during, or after a hypoglycaemic (n=22) or hyperglycaemic (n=33) event based on means of the 15 participants.

4.3 Discussion

Glycaemic excursions are prevalent in people living with T1D and represent a significant threat to their continued health and wellbeing. Current standards in glucose monitoring are inadequate given the severity of T1D and its complications, and recent interest in improving these standards is justified. A non-invasive and convenient measure of glycaemic excursions may significantly improve effectiveness of stringent glucose monitoring, which will lead to better glycaemic outcomes and quality of life for those with T1D. In the RNSH T1D Study, people with T1D participated in a 24-hour study continuously monitoring their HRV and glucose levels. The aim was to determine whether glycaemic excursions, including hypoglycaemia and hyperglycaemia, were marked by significant changes in measures of HRV. Preliminary analysis of these results focused on whether HRV measures were associated with glucose fluctuations across the day, as well as whether individual glycaemic excursions were associated with identifiable changes in HRV. These early analyses showed no meaningful findings, and machine learning techniques may be required to identify trends in HRV during glycaemic excursions, which would require a larger sample size. Additionally, the follow-up statistical analysis by ANOVA indicated there were no significant changes in any HRV measure during hypoglycaemia or hyperglycaemia. This section will explore these findings in the context of the wider literature and discuss the limitations of this research.

4.3.1 Preliminary Analysis of the Sample

Due to the small number of subjects, it is difficult to judge whether the sample was reflective of the wider Australian T1D population. The sample was of similar age and BMI compared with samples observed in other studies in this specific area of research.^{491, 506} In terms of physiology, the values for systolic and diastolic blood pressure were generally within optimal ranges.⁵⁰⁷ Glucose levels were significantly above the optimal range,⁴⁰⁶ though this was expected as this is a clinical characteristic of T1D. The mean HbA_{1c} for the sample was $6.9 \pm 0.9\%$, and as with the UTS T2D Study, roughly half of the sample presented with an HbA_{1c} level below 7.0%, which is the target glycaemia level. This is similar to the proportion of Australians with diabetes who achieve target glycaemia, based on large-scale epidemiological data.¹⁰⁴ The preliminary correlation analysis revealed no significant relationship between HRV measures and BGL values measured at the same time points. A significant finding here would have indicated that fluctuations in HRV are closely related to fluctuations in BGL, and that HRV may be used to estimate glucose levels at any time. This would have been a novel finding, as this has not been demonstrated anywhere in the literature to date. Furthermore, analysis of individual glycaemic

events revealed no consistent relationship between autonomic activity and the onset of hypoglycaemia and hyperglycaemia. This was more likely to have yielded a significant result, as other studies have shown significant changes in HRV measures at the onset of hypoglycaemia. These will be discussed in the next section.

4.3.2 Changes in Autonomic Activity during Hypoglycaemia

In a study conducted by Klimontov, Myakina & Tyan (2016) in 11 older women with T1D, LF power was lower during hypoglycaemia (n=12), as defined by CGM glucose reading ≤ 3.9 mmol/L, compared to fasting LF power.³⁹⁶ HF power was similar during hypoglycaemia compared to fasting HF power, and LF:HF was lower during hypoglycaemia only in subjects with cardiac autonomic neuropathy. Additionally, LF power, HF power, and LF:HF during hypoglycaemia were all similar to fasting levels when the hypoglycaemia event occurred at night.³⁹⁶ As such, the key finding was that daytime LF power was significantly diminished during hypoglycaemia, a finding which was not shared in the RNSH T1D Study. This is likely related to differences in the sample, as the present study recruited an equal number of men and women and with a much younger mean age compared to the study by Klimontov, Myakina & Tyan. In another study, Myakina, Klimontov & Safarov (2015) recruited n=73 subjects with T2D (aged 48-78 years old) who underwent simultaneous CGM and ECG monitoring. They demonstrated that LF power, HF power, and LF:HF all increased significantly (p<0.01) during hypoglycaemia (BGL < 3.9 mmol/L), for both daytime and nocturnal events.⁴⁷⁵ In subjects with cardiac autonomic neuropathy, LF power decreased during hypoglycaemia. Further, Deshmukh and colleagues (2021) observed a significant change in RMSSD by Holter monitoring of T1D subjects with hypoglycaemia unawareness. This study also demonstrated that this change in RMSSD persists following islet cell transplantation and the restoration of euglycaemia in T1D subjects.380

These are just two examples of studies which demonstrated there are marked changes in certain HRV measures during hypoglycaemic events. Due to publication bias, it is difficult to identify studies in the literature which found no significant changes in HRV measures during hypoglycaemia, as was the case with the RNSH T1D Study. This may be the first study to demonstrate that autonomic activity remains relatively stable during hypoglycaemia in T1D, as determined by 10-minute HRV epochs in 15 subjects. It is possible changes in HRV in this sample were significant based on five-minute HRV epochs, however shorter epochs are less accurate and thus were not investigated in the statistical analysis. One recent study published by Lundqvist and colleagues (2021) reported no significant changes in HRV during

hypoglycaemia in a sample of T2D participants,⁵⁰⁸ though to date these findings have not been reported in T1D.

Recently, Bekkink and colleagues (2019) published an original research article with findings that are particularly relevant to this PhD thesis. They investigated changes in LF:HF, SDNN, and RMSSD during hypoglycaemia (n=66 events) in 23 participants (mean age 42 ± 11 years). In their study, mean LF:HF increased significantly (p<0.05) by 0.2 ± 0.4 , mean RMSSD decreased significantly by 4.4 ± 18.1 , and there was no change in SDNN.⁴⁹¹ These findings are promising as they indicate a potential pattern in HRV at the initiation of hypoglycaemia. The presence of specific increases in LF:HF corresponding with specific decreases in RMSSD may be detected non-invasively by a heart rate monitor, as demonstrated in this study, and used to detect the onset of hypoglycaemia. Such research is similar to that of the present study and aims to develop HRV as a non-invasive predictor of hypoglycaemia. However, as Bekkink and colleagues did not investigate LF power, HF power, or total power, it is difficult to compare the results with those of the RNSH T1D Study. Of the three HRV measures which were investigated in both studies, LF:HF and RMSSD changed significantly during hypoglycaemia according Bekkink and colleagues (2019) but did not change significantly according to the results of the RNSH T1D Study. This discrepancy will be explored further.

In their paper, Bekkink and colleagues (2019) reported the mean glucose level of the n=66 hypoglycaemic events to be 3.1 mmol/L, and ranging between 1.6 - 3.9 mmol/L.⁴⁹¹ In their methodology, the authors defined their hypoglycaemic events as any glucose level equal to or below 3.9 mmol/L. Considering this information, it is clear a more generous glucose cut-off was implemented in this study for defining hypoglycaemia, as in the RNSH T1D Study the glucose cut-off value for hypoglycaemia was < 3.0 mmol/L. The criterion for the RNSH T1D Study was more stringent as it incorporated findings from previous literature and based this cut-off value on the findings of numerous clinical studies. However, this more stringent cut-off value meant that the glycaemic events recorded in the present study were more severe than those of the study conducted by Bekkink and colleagues (2019). Despite the added clinical significance, changes in mean HRV measures were not statistically significant in the present study but were significant in the study from Bekkink and colleagues who assessed a larger, but less severe sample of hypoglycaemic events. Differences in the approach of the statistical analysis may have affected these different findings, for example in the present study a one-way ANOVA with repeated measures was used, but Bekkink and colleagues (2019) implemented paired sample ttests to analyse mean changes of HRV measures before and during hypoglycaemia.

One other key observation was made in the paper published by Bekkink and colleagues. They reported that 39% of their sample had impaired awareness of hypoglycaemia, reflecting a

reduced autonomic response to hypoglycaemia. Despite this, the authors were able to identify significant changes in LF:HF and RMSSD during hypoglycaemia. They reported that changes in HRV were not different between subjects with impaired awareness of hypoglycaemia compared with those with intact awareness, though initially they expected there would be because hypoglycaemia unawareness reflects decreased sympathetic nervous system activity during hypoglycaemia. The authors did not explain this finding, though limitations in the scope and design of their study may have limited their ability to comment on why impaired awareness is of particular importance to this area of research as it may limit the potential for heart rate monitors to detect autonomic responses to hypoglycaemia in theory, though findings from some studies, such as the one from Bekkink and colleagues, indicate this may not be the case in real-world settings.

4.3.2.1 Clinically Significant Hypoglycaemia

To further explore the limitations of the present research, as well as other research in this area, it is relevant to discuss the impact of clinically significant hypoglycaemia and impaired awareness of hypoglycaemia. The term clinically significant hypoglycaemia strictly refers to hypoglycaemic events which are associated with symptoms and impaired functioning. For individuals without diabetes, clinically significant hypoglycaemia generally occurs when BGL \leq 3.9 mmol/L, as BGL does not fall below this level under optimal physiological conditions, according to American Diabetes Association consensus.⁴⁹⁵ At this level, symptoms may arise such as sweating, confusion, and blurred vision. For people who experience hypoglycaemia more often, such as people with T1D who take insulin to counter their chronic hyperglycaemia, the onset of symptoms due to low BGL may not occur around the threshold of 3.9 mmol/L. Due to impaired awareness of hypoglycaemia, characterized by a reduced autonomic response to low BGL in as many as 30% of all T1D subjects, people with T1D may not experience symptoms until their BGL falls as low as 2.3 mmol/L, and some may not experience symptoms at all. Problematically, because of the pathophysiology of this unawareness and how it develops, it is likely that most people with T1D are affected by it to a certain degree. This unawareness may be graded, and as such some people are affected more strongly whilst others are affected only minimally by it. It is clear clinically significant hypoglycaemia may not occur in many people with T1D at the threshold of \leq 3.9 mmol/L, and there are different views in the literature on how hypoglycaemia should be defined.

A 12-week study on n=3,912 subjects demonstrated that less than 40% of the hypoglycaemia episodes experienced by people with diabetes are symptomatic when hypoglycaemia is defined as BGL \leq 3.9 mmol/L (Figure 4.10). However, as discussed, this threshold does not equate with symptoms of hypoglycaemia in people with T1D. Amiel and colleagues (2008) argue that defining hypoglycaemia as simply any glucose value equal to or below 3.9 mmol/L leads to overestimation of clinically significant hypoglycaemia.⁵⁰⁹ The members of the International Hypoglycaemia Study Group agree that BGL < 3.0 mmol/L is sufficient to indicate clinically relevant or important hypoglycaemia.⁵¹⁰ According to Swinnen and colleagues (2009), when hypoglycaemia is defined by a more clinically relevant threshold, such as < 3.0 mmol/L, roughly 70% of subjects experience autonomic symptoms, and only 30% do not.494 This is also demonstrated in Figure 4.10. This 30% is similar to the rate of hypoglycaemia unawareness observed in large samples of people with T1D and is the proportion of people with T1D who experience no autonomic symptoms at almost any level of low BGL. In the present study, it was therefore accepted that lowering the threshold further when defining hypoglycaemia would not significantly affect the observed proportion of subjects who experience symptoms, and thus < 3.0 mmol/L was an appropriate threshold for defining clinically significant hypoglycaemia. It was accepted that there will always be a proportion of people with diabetes who do not experience symptoms of hypoglycaemia due to hypoglycaemia unawareness.⁴⁹⁴ This is the main reason subjects were not screened for impaired awareness of hypoglycaemia when participating. In a real-world scenario, this unawareness is likely to be present in about 30% of subjects, and changes in HRV that precede hypoglycaemia may be identified even in the absence of symptoms, i.e., HRV may predict hypoglycaemia even when it is not clinically significant. The research of Bekkink and colleagues (2019) suggests as much, as they implemented a glucose cut-off of \leq 3.9 mmol/L when defining hypoglycaemia in their study. However, the results of the present study do not indicate that HRV measures may predict hypoglycaemia.



Figure 4.10 Proportion of asymptomatic hypoglycaemia at different definitions.

Figure 4.10 presents the different thresholds for hypoglycaemia used in the literature and the percentage of subjects who reported no symptoms at each of the thresholds. BGL \leq 3.9 mmol/L (in red) is distinguishable as the pathological state of hypoglycaemia, as BGL does not exist below this level in non-pathological states. As indicated by the red column, over 60% of subjects have no symptoms of hypoglycaemia when it is defined as BGL \leq 3.9 mmol/L. Defining hypoglycaemia by a symptomatic standard is more relevant in clinical practice, for example BGL \leq 2.9 mmol/L (in blue). This is the cut-off used in the present study, though for consistency it was described as '< 3.0 mmol/L' in this thesis. Adapted from Swinnen and colleagues (2009).⁴⁹⁴

4.3.2.2 Neurophysiology of Hypoglycaemia

The autonomic nervous system is responsible for maintaining glucose homeostasis, which includes responding to hypoglycaemia. The detection of low BGL leads to efferent activation of the branches of the splanchnic nerve which connect to the pancreas, liver, and adrenal medulla. This results in secretion of glucagon from the pancreas, glucose from the liver, and catecholamines from the adrenal medulla.⁵¹¹ As described previously, the neurophysiological response to hypoglycaemia involves autonomic symptoms, including sweating, heart palpitations, and trembling. If BGL continues to decline, such as when insulin medication continues to drive the uptake of glucose into cells from the blood, then neuroglycopenic

symptoms may predominate, including confusion, dizziness, and difficulty concentrating.⁸⁹ A clinical study on n=10 young adults demonstrated that awareness of hypoglycaemia was significantly higher in subjects during clamped hypoglycaemia with adrenergic blockade (phentolamine and propranolol) compared to subjects with pan-autonomic blockade (phentolamine, propranolol and atropine).⁵¹² The authors, Towler and colleagues (1993), concluded that awareness of hypoglycaemia was largely the result of autonomic symptoms (shakiness, heart pounding, nervous) rather than neuroglycopenic symptoms (confusion, drowsiness), reinforcing the idea that the autonomic response to hypoglycaemia acts as an early warning system. In individuals with reduced awareness of hypoglycaemia, this is driven by reduced autonomic response to low BGL. The absence of autonomic symptoms lowers the threshold for clinically significant hypoglycaemia, and individuals may not realise they are experiencing a hypoglycaemic event until BGL becomes low enough that it causes problems with concentration and decision making. This raises an important concern for the RNSH T1D Study and similar research. HRV measures may be unable to reliable predict the onset of hypoglycaemia because the physiological variable they are assessing – autonomic activity – may remain unchanged in a large proportion of those who experience hypoglycaemia, as documented in this sample. Based on the literature, it is likely that autonomic changes were present in this sample, so the lack of significant findings may be due to the small sample size or the definition of hypoglycaemia. Hypoglycaemia was defined by a short duration of 10 minutes instead of 15 minutes, and subjects were not required to validate hypoglycaemia by confirmation finger stick BGL. As such, some of the events defined as hypoglycaemia in these analyses may have not met the standardized criteria for hypoglycaemia.

4.3.3 Changes in Autonomic Activity during Hyperglycaemia

Hyperglycaemia is an independent risk factor for cardiovascular disease as well as numerous microvascular complications of diabetes. The ability to non-invasively predict the onset of hyperglycaemia may facilitate early intervention and prevention of hyperglycaemia, and thus reduce long term complications of diabetes. There is substantially less research in the area of non-invasive HRV measures predicting the onset of hyperglycaemia compared to hypoglycaemia. The RNSH T1D Study investigated changes in frequency-domain HRV measures during clinically relevant hyperglycaemia, as defined by BGL \geq 12.0 mmol/L, and presented novel non-significant findings. There were no significant changes in autonomic activity, as assessed by HRV, at the onset of hyperglycaemia, as assessed by CGM. This section aims to provide greater context and meaning to these findings.

In a clinical study of n=30 young females without chronic illness, Majeed and Yar (2020) demonstrated that LFnu and LF:HF were significantly higher during hyperglycaemia compared to the fasting state.⁵¹³ However, in the present study, HRV measures were compared before, during, and after hyperglycaemia onset, rather than compared with fasting levels. Additionally, there are marked differences in the neurophysiological response to suboptimal BGL in people without diabetes compared to people with T1D, and impaired awareness of hypoglycaemia is just one factor involved in these differences. Lundqvist and colleagues (2021) investigated autonomic and hormonal changes during hypoglycaemia and hyperglycaemia in n=15 overweight subjects with T2D and n=15 lean weight subjects with T2D.⁵⁰⁸ Though there are marked physiological and pathological differences between T1D and T2D, it may be relevant to compare the results of the present study with the results of the lean weight group with T2D. In this group, the authors observed no change in HRV frequency-domain HRV measures during hyperglycaemia (BGL ≥ 13.0 mmol/L), including LF power, HF power, and total power, and LF:HF. This is consistent with the findings of the RNSH T1D Study.

For clinical studies on people with T1D, there are few significant findings. According to a 96hour study on n=37 individuals with T1D, hyperglycaemia (BGL \geq 15.0 mmol/L) is not associated with clinically relevant cardiac arrhythmias,⁵¹⁴ and the authors suggest that cardiac autonomic control may not be impaired during hyperglycaemia. Based on the work of Berkelaar and colleagues (2013), hyperglycaemia (BGL \geq 10.0 mmol/L) is not correlated with HRV measures, including RMSSD and HF power, or cardiac vagal tone,⁵¹⁵ and the authors did not assess any other HRV measure relevant to the present study. As discussed, there are few clinical studies in this area, and more research is required to develop a better understanding of autonomic changes during hyperglycaemia. Future studies should consider clinical relevance when defining hyperglycaemia, as inconsistencies in how hyperglycaemia is defined by different researchers may explain differences in the statistical results. Specifically for frequency-domain HRV measures, more studies with larger sample sizes are required to form a consensus.

4.3.3.1 Neurophysiology of Hyperglycaemia

The response of the autonomic nervous system to increasing BGL has been discussed in a previous chapter. To summarise, the autonomic nervous system effectively decreases BGL by increasing insulin release from the pancreas, as well as by directly innervating the liver. The elevation of BGL above the optimal limit is most commonly caused by a lack of insulin response to food or caloric intake. In T1D, the near or total deficiency of circulating insulin leads to large glycaemic excursions in the postprandial state, as somatic cells in the body are unable to take up

glucose from the blood without insulin signalling. This results in chronically high levels of blood glucose, even hours after the ingestion of a meal. In a clinical study on n=20 newly diagnosed people with T2D, Marfella and colleagues (2000) observed acute increases in systolic blood pressure, diastolic blood pressure, heart rate, and plasma catecholamines during hyperglycaemia.⁵¹⁶ They concluded that, in the absence of diabetes complications including autonomic neuropathy, hyperglycaemia causes an autonomic-mediated hemodynamic response. This research is part of a growing body of evidence showing that hyperglycaemia is a risk factor for cardiovascular events, and this applies to people without diabetes as well.⁵¹⁷

Early literature from Legler and colleagues (1982) established that cortisol levels increase after a meal,⁵¹⁸ and this has been ratified by randomized clinical research from Reynolds and colleagues (2001).⁵¹⁹ However, hyperglycaemia in people with T1D is not always the result of a postprandial response, as it may result when the effects of glucose-lowering medications wear off. Lundqvist and colleagues (2020) contend that the hormonal response to hyperglycaemia, including cortisol release, varies between individuals, but in general the autonomic response to rising BGL is characterised by a stress response.⁵⁰⁸ In people with autonomic neuropathy or diminished autonomic tone, this stress response, as measured partly by LF power, is smaller.⁵⁰⁸ This is similar to hypoglycaemia, in which decreasing BGL leads to a 'fight or flight' stress response mediated by the sympathetic branch of the autonomic nervous system. It may be inferred that the sympathetic nervous system responds to both high or low BGL to maintain glucose homeostasis, and inability to maintain BGL in the target range may predispose an individual to further glycaemic excursions, such as repeated hypoglycaemia risk in people with impaired awareness of hypoglycaemia. An important conclusion to be made from this is that even if HRV measures cannot accurately predict BGL in an individual with diabetes, as suggested by the results of the RNSH T1D Study, they may still be useful in identifying individuals with diminished autonomic response to glycaemic excursions. This information may empower individuals to more stringently monitor their BGL using CGM or more stringently manage their condition with lifestyle interventions.

4.3.4 Strengths and Limitations

The potential to non-invasively predict glycaemic events represents a significant landmark in improving the quality of life and health outcomes for people with diabetes. A major strength of the RNSH T1D Study is that it provided further insight into a novel area of research. In terms of study design, the 24-hour monitoring period was an effective means of capturing multiple glycaemic events for each participant. Shorter monitoring periods would have reduced the

amount of data points captured and reduced the statistical power of the analysis. There were two key elements of the methodology which were important in strengthening the scope of this research. During the monitoring period, participants continued their routine work and life schedules, including medications and meals, with as little interruption as possible. Additionally, the criteria used to define glycaemic events were based on values that were clinically relevant. Due to the two key design elements, the findings presented in this research are applicable to real-world clinical settings. Based on these results, it can be concluded that HRV measures do not change significantly immediately before, during, or after the onset of glycaemic events in people with T1D during their regular work and life routines. Though clinical studies have observed significant differences in hypoglycaemia induced by clamp techniques, the RNSH T1D Study may be the first study to demonstrate no statistical significance in a real-world setting.

Some limitations of the RNSH T1D Study have been provided already and will be consolidated in this section. In this study, participants were recruited based on their likelihood of experiencing a glycaemic event within 24-hours of attending the clinic at RNSH. This was to increase the number of glycaemic events captured during the monitoring period which would improve statistical power, as the average person with T1D does not experience many glycaemic events in a 24-hour period. This in turn likely led to the recruitment of more subjects with impaired awareness of hypoglycaemia, as this condition is more prevalent in people with T1D who experience recurring episodes of hypoglycaemia (see Section 1.5.1.3). Due to the pathophysiology of impaired awareness of hypoglycaemia, subjects with this condition were likely to have a reduced autonomic response to hypoglycaemia, which could explain why mean HRV measures did not change significantly in this cohort during hypoglycaemic events. As subjects were not screened for impaired awareness of hypoglycaemia, it is unknown how many in this study had reduced autonomic response during hypoglycaemia, and this limits the conclusions which can be made here. Future studies should consider screening for this condition, as well as restrict the proportion of recruited subjects with this condition to a level reflective of the general T1D population, roughly 30%. In people with T2D who depend on insulin, this proportion is roughly 10%. Future studies which attempt to outright exclude participants based on the presence of impaired awareness of hypoglycaemia may not be able to apply their findings to the broader population of people with diabetes. Duration of diabetes for each subject was not recorded by the researchers in error. This would have been useful since a longer mean duration of diabetes in subjects may have indicated an increased risk or presence of autonomic neuropathy.

Though there have been significant improvements with recent models, there are still known lagtimes in the detection of glycaemic excursions by CGM systems. Participants in this study utilised several different CGM system models to record their glucose levels continuously. The difference between when participants were undergoing a glycaemic excursion compared to when the excursion was detected by the glucose monitor may have been as much as seven minutes on average, though based on current data this varies between CGM system models. For a rough example, a participants BGL may have decreased to 2.9 mmol/L at 10:00am during the study. As CGM systems estimate BGL based on glucose levels in the interstitial fluid, instead of directly measuring glucose levels in the blood, it is unlikely the CGM system would record this value of 2.9 mmol/L until approximately 10:07am (see Section 1.7.2). When analysing the data, the hypoglycaemic event would be registered as 10:07am, and changes in autonomic activity would be calculated from heart rate epochs 10-minutes before, during, and after this exact time. However, there was no corresponding lag-time with the heart rate monitors when recording autonomic activity. As such, there was likely a mismatch between the time when the glycaemic events were recorded compared to when they actually occurred in participants. This is a difficult problem to address, as a predictive model using HRV measures to predict glycaemic events will require large amounts of data, which currently can only be provided by a CGM system. This may change with technological advances, however.

Improvements should be made on this study design in future studies. Using a cut-off for hyperglycaemia as BGL \geq 10.0 mmol/L may provide more data points to analyse, though clinical relevance of these events should be considered. A participant questionnaire could be provided to collect subjective experiences of glycaemic events, indicating clinical relevance if there are signs or symptoms concurrent with each event. Hypoglycaemic events should be validated by a finger stick confirmation assessment. Improvements could have been made to the autonomic monitoring aspects of this study. Firstly, future research may calculate HRV measures from the full 24 hours of monitoring and compare against other groups. Secondly, the FirstBeat heart rate monitor may have lacked sensitivity required to detect changes in autonomic activity during glycaemic events, so a Holter monitor may be considered as an alternative as it is the hold standard for HRV measurement. Overall, this study did not provide evidence for a change in HRV measures during glycaemic events and did not indicate that HRV measures may act as an 'early warning' system for hypoglycaemic or hyperglycaemia in T1D. The evidence does not support the hypothesis stated in Chapter 1. CGM remains the gold standard for identifying these events in the absence of symptoms.

Chapter 5. Conclusion

The scope of diabetes is significant. The cost to individuals and to global healthcare systems is substantial, and due to its nature as a 'Disease of Civilization', the threat of T2D, the main type of diabetes, is not likely to ease without meaningful social and cultural changes. The initial chapter of this thesis introduced numerous concerns about the current standards in treating and managing diabetes from the perspective of the medical science community. According to scientific consensus, currently available glucose monitoring tools are inadequate given the scope of diabetes. The improvement of these tools, or replacement with superior ones, has been a major priority for decades, as has improvement of pharmacological treatments. Currently available tools have improved markedly in recent years, but core problems, such as their invasive nature, have yet to be addressed. There is growing interest in non-invasive approaches to glucose monitoring, such as tools that estimate BGL based on saliva or sweat samples, as there is evidence that non-invasive alternatives to current invasive measures are more efficient. To summarize, better efficiency is associated with higher quality glucose monitoring and management of diabetes, and this in turn is associated with better quality of life for the many people living with diabetes. However, to date there is no non-invasive measure of blood glucose that is commercially successful. All clinically relevant glucose monitoring tools are invasive, and this has been identified as a core problem in the scientific literature.

HRV is an emerging technology in this area with novel applications. There is strong evidence that HRV measures can predict diabetes complications, and the literature advocates for the use of routine HRV assessments in clinical settings for monitoring of complications such as autonomic neuropathy. HRV is also an excellent marker of autonomic control of the heart. The autonomic nervous system is responsible for regulating various functions throughout the body, including the regulation of blood glucose. It achieves this through innervation of various organs, such as the liver and pancreas, to release specific hormones in response to fluctuating glucose levels. Though the action of the autonomic nervous system on these endocrine organs cannot be measured easily, its action on the heart can be, and it can be measured non-invasively. The level of autonomic innervation that an individual exerts on their heart is inversely related to their glycaemia. This is due to various reasons, such as the fact that total autonomic activity is lower in people with poor physical fitness, and these people tend to have high levels of blood glucose. However, it is also related to the acute response of the autonomic nervous system to food intake. The autonomic nervous system prepares the body for food intake through innervation of various organs and nerve pathways. Though the mechanism is not well understood, changes in autonomic modulation of the heart are associated with changes in blood glucose. It was this physiological principle which first led to the development of the novel hypothesis that

autonomic innervation of the heart, as measured by HRV, may estimate glycaemia. Therefore, HRV measures may be a non-invasive alternative to current glucose monitoring.

This novel hypothesis was tested by Rothberg and colleagues (2016) in people with and people without diabetes. The authors set out to determine whether HRV measures were related to BGL in a way that could lead to the development of them as non-invasive markers of BGL. Their work was continued by Jarman and colleagues (2021) in subjects without chronic illness, and in the UTS Pilot Study over a shorter time period. These studies aimed to accumulate sufficient clinical data to justify further exploration of HRV measures as non-invasive markers of BGL. The UTS T2D Study was the main study conducted as part of this PhD candidature and aimed to consolidate years of exploratory research into a study on people without diabetes and people with T2D, the most common type of diabetes. While several HRV measures accounted for the variability in BGL, no one measure was significant in a multivariate model, and therefore HRV is limited in its capacity to diagnose diabetes or to predict BGL from HRV measures alone.

The ability to predict glycaemia using HRV measures represents a landmark in diabetes management, as it would improve the quality of life and health outcomes for those living with diabetes and reduce the overall burden of the condition on individuals and society. The ability to predict overall glycaemia is also important in the screening and diagnosis of diabetes, and undiagnosed diabetes is cause for significant disability and morbidity worldwide. Early detection of diabetes would be facilitated with the introduction of a convenient and reliable non-invasive measure of blood glucose, and early detection and intervention is associated with reduced risk of diabetes complications and increased quality of life. HRV may not just be relevant in the prediction of fasting and postprandial glycaemia. Another study conducted as part of this PhD candidature, the RNSH T1D Study, aimed to justify the use of HRV measures in the prediction of glycaemic events, which are prevalent in T1D. Though T1D affects only a minor percentage of people with diabetes compared to T2D, people living with T1D are dependent on insulin to survive, and this reliance on exogenous insulin predisposes individuals to hypoglycaemia. Due to the limitations of CGM systems, the ability to reliably predict glycaemic events using HRV measures would significantly improve glucose monitoring for those living with T1D, and this would in turn improve their quality of life as well as potentially reduce the costs associated with managing their condition. The results of the RNSH T1D Study are relevant in a real-world clinical setting, however they do not indicate that HRV measures can predict glycaemic events, such as hypoglycaemia. The effect of impaired awareness of hypoglycaemia on recruited subjects may cause problems for researchers attempting to develop a real-world application of this technology. Nevertheless, some future directions for this novel area of research have been provided in this thesis. The next section will address some of the broader

future directions for researchers in this area, as well as discuss the research which has been growing in recognition in recent years.

5.1 Future Directions

Limitations in the design of these studies conducted as part of this PhD candidature may have contributed to the lack of significant findings and the contradictions compared with current literature. If repeated, several improvements should be made to these studies. For the UTS T2D Study which investigated correlations between HRV measures and BGL in people with T2D and people without, subjects should be seated during the full assessment period and meal size should be standardized. Participants with T2D should be excluded if they have HbA1c above 7.5% or autonomic neuropathy, and this should be assessed during the screening process and prior to study participation. Alcohol consumption should also be limited in the exclusion criteria. For the RNSH T1D Study, hypoglycaemia should be defined in accordance with the standardized reporting guidelines, which means a minimum of 15 minutes in duration and validated by a finger stick BGL. Subjects should be carefully screened or assessed for autonomic neuropathy, and a questionnaire should be provided to subjects to collect information on clinical significance, or the presence of symptoms, during glycaemic events. Using a lower threshold for hypglycaemic cut-off, such as 10.0 mmol/L, may help to capture more events, and calculating HRV measures from the full 24-hours of recording may provide additional insights. Studies should be designed around these improvements.

This thesis investigated the potential of HRV as a non-invasive alternative to invasive glucose monitoring, and thus priority was given to lightweight, portable devices when selecting a device capable of determining HRV. There are several heart rate monitors available in the form of wearable smartwatches. Whilst these are convenient and lightweight to use, concerns have been raised regarding their ability to accurately estimate HRV. Smartwatches do not directly measure the electrical activity of the heart, but rather detect an individual's radial pulse under the skin, usually using a technique known as photoplethysmography. Regardless of the method used to detect a person's pulse, the use of pulse rate variability as an estimate of HRV remains a contentious topic. Pulse rate variability is only sufficiently accurate as an estimate of HRV in participants that are young, at rest, and without chronic illness.⁵²⁰ Physical or mental activity have been shown to exaggerate differences in pulse rate variability and HRV, and short-term variability is somewhat overestimated in pulse rate variability. A promising exception to these is the Polar S810, which has demonstrated excellent agreement with an ECG in providing time-domain HRV parameters for three-minute⁵²¹ and five-minute recordings.⁵²² Given that there is

no evidence to support the viability of a smartwatch to estimate frequency-domain HRV parameters and for long-term recordings, smartwatches were not considered for use within this study. In future, however, the accuracy of these wearable devices may be improved to a point where they can provide reliable estimates of glycaemia. This would be a significant improvement to current management of BGL and diabetes. Modern smartwatches are already capable of tracking stress, sleep hygiene, hear rate, and more, and one day may also be capable of tracking blood glucose.

There is growing interest in the applications of machine learning in predicting glycaemia. In their 2018 paper, Zhu and colleagues presented the results of their neural network which had been trained on 115,200 glucose data points from CGM systems attached to n=6 subjects across 40 days.⁵²³ Their model outperformed many existing algorithms, despite the fact machine learning models usually require training data with a much larger size to be effective. However, they noted that glucose levels predicted by their model lagged behind the true glucose levels observed in their subjects , and the error in prediction was high when subjects consumed a meal or took their insulin medication. A neural network developed by Zecchin and colleagues (2011) provided competitive predictive value for its time by learning carbohydrate (meal) intake alongside learning glucose information.⁴⁸³ An algorithm developed by Li (2019) achieved minimal time lag in a real subject dataset by also training their model using carbohydrate data, suggesting that it is an industry standard to manage or account for the effect of meals on glycaemia.⁵²⁴ Meals are an inevitable source of glucose spikes, and thus glycaemic variability, in people living with diabetes. Goldner and colleagues (2018), who trained their machine learning model on nearly two million BGL measurements from 14,706 people with T2D, have indicated their model could be improved by incorporating meal behaviours into their training set, such as carbohydrate amount.⁵²⁵ Pustozerov and colleagues (2020) advocate the importance of predicting postprandial BGL in diabetes, as machine learning algorithms may predict the magnitude of a glucose spike after consuming a meal.⁵²⁶

In terms of predicting low BGL, Wang and colleagues (2020) created a machine learning model by Bayesian inference capable of predicting hypoglycaemia in T1D with 96% accuracy and 92% precision, which has potential clinical utility.⁵²⁷ As mentioned, there are few studies on the applications of machine learning in SMBG. However, a probabilistic model developed by Sudharsan, Peeples, and Shomali (2014) had a sensitivity of 92% and a specificity of 70% in predicting a hypoglycaemic event within the next 24 hours.⁴⁸⁶ They achieved this using only 10 SMBG values per week, which is a reasonable number of blood glucose assessments for someone living with diabetes, based on current standards. A novel algorithm presented by Cichosz and colleagues has shown promising results by combining information from a Holter

monitor with concurrent values from a continuous glucose monitoring system. The algorithm detected 16/16 hypoglycaemic events in subjects with T1D with a sensitivity of 79% and a specificity of 99%.⁴⁵² This study represents an appealing line of research, as it aims to overcome some of the current problems people with diabetes face. Current SMBG is costly,³⁰⁵ though a portable ECG, such as a Holter monitor, may present a reduced financial burden and, at the very least, would provide a non-invasive option. The R-waves from an ECG, used in the determination of HRV parameters, have distinct profiles that make them suitable for detection by computer algorithms.⁵²⁰ A real-world application of this technology would incorporate the predictive model within the monitoring device, allowing for real-time analysis of autonomic activity. Consequently, an inexpensive portable heart rate monitor capable of estimating blood glucose may be commercialised.

It is clear from emerging research that a non-invasive, accurate measure of BGL will almost certainly involve advanced machine learning.⁵²⁸ The findings of the multiple studies conducted as part of this thesis add to the growing body of evidence that HRV measures may be an ideal marker of glycaemia. However, further research is needed. Continuous glucose monitoring (CGM) remains the gold-standard for measurement of BGL in diabetes, however HRV represents a promising area which can add to detection of diabetes and glycaemic events in tandem with CGM. Though the scope of this PhD limits the applications of these findings, it can be concluded that the future of HRV is promising in this area of research. The UTS Pilot Study provided a framework and preliminary data which was used to design a larger study. Following this, the UTS T2D Study demonstrated that HRV measures obtained whilst individuals are eating may be used to predict their postprandial BGL, a novel finding. The RNSH T1D Study investigated changes in HRV measures during glycaemic events, and though no statistically significant findings were observed, HRV measures may still be relevant in the management of T1D. There are certain confounders which future researchers should consider the impact of when investigating the clinical applications of HRV measures in predicting glycaemic events.

Overall, these findings presented in this thesis indicate that HRV possesses clinical relevance and real-world utility in the management of diabetes, and further investigation is warranted. Future studies should focus on gathering data in real-world situations, such as allowing subjects to continue their regular routines, including medications and meals. Studies may also improve sample power, which is a common limitation in this area of study, by pooling subjects from their individual groups and assessing correlations between HRV measures and BGL across a wider, unified sample. A useful follow-up study to this PhD research might investigate correlations between the change in HRV and change in BGL in postprandial states. Further, large amounts of data points may be used to train more sophisticated machine learning tools, such as deep learning artificial intelligence, and clinically relevant thresholds of glycaemic events should be considered when developing therapeutic technologies. It is important that researchers in the scientific community empathise with those living with the condition they are attempting to improve, and to consider their specific needs when developing potential solutions. This has been a core focus of this thesis, as empathy will invariably facilitate the real-world applications of clinical research.

Chapter 6. References

1. Eaton SB, Konner M. Paleolithic Nutrition - a consideration of its nature and current implications. N Engl J Med. 1985; 312: 283-289.

2. Macaulay V, Richards M, Hickey E, et al. The emerging tree of West Eurasian mtDNAs: a synthesis of control-region sequences and RFLPs. Am J Hum Genet. 1999; 64: 232-249.

3. Bollyky TJ, Templin T, Andridge C, et al. Understanding the relationships between noncommunicable diseases, unhealthy lifestyles, and country wealth. Health Aff (Millwood). 2015; 34: 1464-1471.

4. Li Y, Ley SH, Tobias DK, et al. Birth weight and later life adherence to unhealthy lifestyles in predicting type 2 diabetes: prospective cohort study. BMJ. 2015; 351: h3672.

5. Bennett P, Burch T, Miller M. Diabetes mellitus in American (Pima) indians. The Lancet. 1971; 298: 125-128.

6. Zimmet P, Arblaster M, Thoma K. The effect of westernization on native populations. Studies on a Micronesian community with a high diabetes prevalence. Aust N Z J Med. 1978; 8: 141-146.

7. Balick MJ, Lee RA, De Gezelle JM, et al. Traditional lifestyles, transition, and implications for healthy aging: An Example from the remote island of Pohnpei, Micronesia. PLoS One. 2019; 14: e0213567.

8. O'dea K. Marked improvement in carbohydrate and lipid metabolism in diabetic Australian Aborigines after temporary reversion to traditional lifestyle. Diabetes. 1984; 33: 596-603.

9. Frassetto LA, Schloetter M, Mietus-Synder M, et al. Metabolic and physiologic improvements from consuming a paleolithic, hunter-gatherer type diet. Eur J Clin Nutr. 2009; 63: 947.

10. Carrera-Bastos P, Fontes-Villalba M, O'Keefe JH, et al. The western diet and lifestyle and diseases of civilization. Res Rep Clin Cardio. 2011; 2: 15-35.

11. Kopp W. How western diet and lifestyle drive the pandemic of obesity and civilization diseases. Diabetes Metab Syndr Obes Targ Ther. 2019; 12: 2221.

12. Luke A, Cooper RS. Physical activity does not influence obesity risk: time to clarify the public health message. Int J Epidemiol. 2013; 42: 1831-1836.

13. Cordain L, Eaton SB, Sebastian A, et al. Origins and evolution of the Western diet: health implications for the 21st century. Am J Clin Nutr. 2005; 81: 341-354.

14. Fung TT, Rimm EB, Spiegelman D, et al. Association between dietary patterns and plasma biomarkers of obesity and cardiovascular disease risk. Am J Clin Nutr. 2001; 73: 61-67.

15. Herman WH, Zimmet P. Type 2 diabetes: an epidemic requiring global attention and urgent action. Diabetes Care. 2012; 35: 943-944.

16. Zuo H, Shi Z, Hussain A. Prevalence, trends and risk factors for the diabetes epidemic in China: a systematic review and meta-analysis. Diabetes Res Clin Pract. 2014; 104: 63-72.

17. World Health Organization. WHO methods and data sources for country-level causes of death 2000-2019. Geneva, Switzerland: World Health Organization; 2020.

18. Mathers CD, Stevens GA, Boerma T, et al. Causes of international increases in older age life expectancy. The Lancet. 2015; 385: 540-548.

19. Salomon JA, Wang H, Freeman MK, et al. Healthy life expectancy for 187 countries, 1990–2010: a systematic analysis for the Global Burden Disease Study 2010. The Lancet. 2013; 380: 2144-2162.

20. Go AS, Mozaffarian D, Roger VL, et al. Executive summary: heart disease and stroke statistics—2013 update: a report from the American Heart Association. Circulation. 2013; 127: 143-152.

21. Stambler I, Jin K, Lederman S, et al. Aging health and R&D for healthy longevity must be included into the WHO work program. Aging Dis. 2018; 9: 331.

22. Budreviciute A, Damiati S, Sabir DK, et al. Management and Prevention Strategies for Noncommunicable Diseases (NCDs) and Their Risk Factors. Front Pub Health. 2020; 8: 788.

23. Esteghamati A, Larijani B, Aghajani MH, et al. Diabetes in Iran: prospective analysis from first nationwide diabetes report of National Program for Prevention and Control of Diabetes (NPPCD-2016). Sci Rep. 2017; 7: 1-10.

24. Jaacks LM, Siegel KR, Gujral UP, et al. Type 2 diabetes: A 21st century epidemic. Best Pract Res Clin Endocrinol Metab. 2016; 30: 331-343.

25. American Diabetes Association. Diagnosis and classification of diabetes mellitus. Diabetes Care. 2010; 33: 62-69.

26. Gabir MM, Hanson RL, Dabelea D, et al. The 1997 American Diabetes Association and 1999 World Health Organization criteria for hyperglycemia in the diagnosis and prediction of diabetes. Diabetes Care. 2000; 23: 1108-1112.

27. Dall TM, Mann SE, Zhang Y, et al. Distinguishing the economic costs associated with type 1 and type 2 diabetes. Population health management. 2009; 12: 103-110.

28. American Diabetes Association. Classification and diagnosis of diabetes. Diabetes Care. 2016; 39: 13-22.

29. World Health Organization. Global report on diabetes. Geneva, Switzerland; 2016.

30. Danaei G, Finucane MM, Lu Y, et al. National, regional, and global trends in fasting plasma glucose and diabetes prevalence since 1980: systematic analysis of health examination surveys and epidemiological studies with 370 country-years and 2.7 million participants. The Lancet. 2011; 378: 31-40.

31. Cho N, Shaw J, Karuranga S, et al. IDF Diabetes Atlas: Global estimates of diabetes prevalence for 2017 and projections for 2045. Diabetes Res Clin Pract. 2018; 138: 271-281.

32. Lašaitė L, Ostrauskas R, Žalinkevičius R, et al. Diabetes distress in adult type 1 diabetes mellitus men and women with disease onset in childhood and in adulthood. J Diabetes Complications. 2016; 30: 133-137.

33. Leśniowska J, Schubert A, Wojna M, et al. Costs of diabetes and its complications in Poland. The European Journal of Health Economics. 2014; 15: 653-660.

34. Lee CMY, Colagiuri R, Magliano DJ, et al. The cost of diabetes in adults in Australia. Diabetes Res Clin Pract. 2013; 99: 385-390.

35. Estruch R, Ros E, Salas-Salvadó J, et al. Primary prevention of cardiovascular disease with a Mediterranean diet. N Engl J Med. 2013; 368: 1279-1290.

36. Sofi F, Cesari F, Abbate R, et al. Adherence to Mediterranean diet and health status: metaanalysis. BMJ. 2008; 337: a1344.

37. Musaiger AO. Overweight and obesity in eastern mediterranean region: prevalence and possible causes. J Obes. 2011; 2011.

38. National Diabetes Services Scheme. Diabetes data snapshots - All types of diabetes 2021 [December 2021:[Available from: <u>https://www.ndss.com.au/wp-content/uploads/ndss-data-snapshot-202112-all-types-of-diabetes.pdf</u>.

39. Australian Bureau of Statistics. National, state and territory population 2021 [Available from: https://www.abs.gov.au/statistics/people/population/national-state-and-territory-population/dec-2021#key-statistics.

40. Bullard KM, Cowie CC, Lessem SE, et al. Prevalence of diagnosed diabetes in adults by diabetes type—United States, 2016. Morb Mort Weekly Rep. 2018; 67: 359.

41. Merger SR, Kerner W, Stadler M, et al. Prevalence and comorbidities of double diabetes. Diabetes Res Clin Pract. 2016; 119: 48-56.

42. Bhatia K, Syal R, editors. Predictive analysis using hybrid clustering in diabetes diagnosis. 2017 Recent Developments in Control, Automation & Power Engineering (RDCAPE); 2017: IEEE.

43. Mamtani M, Kulkarni H, Dyer TD, et al. Waist circumference independently associates with the risk of insulin resistance and type 2 diabetes in Mexican American families. PLoS One. 2013; 8: e59153.

44. Forlenza GP, Moran A, Nathan B. Other specific types of diabetes. 2021.

45. Shidler KL, Letourneau LR, Novak LM. Uncommon Presentations of Diabetes: Zebras in the Herd. Clin Diabetes. 2020; 38: 78-92.

46. Chiang JL, Kirkman MS, Laffel LM, et al. Type 1 diabetes through the life span: a position statement of the American Diabetes Association. Diabetes Care. 2014; 37: 2034-2054.

47. ADA. Diagnosis and classification of diabetes mellitus. Diabetes Care. 2014; 37: S81-S90.

48. Grayson BE, Seeley RJ, Sandoval DA. Wired on sugar: the role of the CNS in the regulation of glucose homeostasis. Nat Rev Neurosci. 2013; 14: 24-37.

49. Atkinson MA, Eisenbarth GS, Michels AW. Type 1 diabetes. The Lancet. 2014; 383: 69-82.

50. Roep BO. The role of T-cells in the pathogenesis of Type 1 diabetes: from cause to cure. Diabetologia. 2003; 46: 305-321.

51. Saberzadeh-Ardestani B, Karamzadeh R, Basiri M, et al. Type 1 diabetes mellitus: cellular and molecular pathophysiology at a glance. Cell Journal (Yakhteh). 2018; 20: 294.

52. Pugliese A. The multiple origins of type 1 diabetes. Diabet Med. 2013; 30: 135-146.

53. Rother KI. Diabetes treatment—bridging the divide. The New England journal of medicine. 2007; 356: 1499.

54. Noble JA, Erlich HA. Genetics of type 1 diabetes. Cold Spring Harb Perspect Med. 2012; 2: a007732.

55. Risch N. Assessing the role of HLA-linked and unlinked determinants of disease. Am J Hum Genet. 1987; 40: 1.

56. Noble JA, Valdes AM, Varney MD, et al. HLA class I and genetic susceptibility to type 1 diabetes: results from the Type 1 Diabetes Genetics Consortium. Diabetes. 2010; 59: 2972-2979.

57. Richardson SJ, Rodriguez-Calvo T, Gerling IC, et al. Islet cell hyperexpression of HLA class I antigens: a defining feature in type 1 diabetes. Diabetologia. 2016; 59: 2448-2458.

58. Pociot F, Akolkar B, Concannon P, et al. Genetics of type 1 diabetes: what's next? Diabetes. 2010; 59: 1561-1571.

59. Stankov K, Benc D, Draskovic D. Genetic and epigenetic factors in etiology of diabetes mellitus type 1. Pediatrics. 2013; 132: 1112-1122.

60. Bahar SA, Razieh K, Mohsen B, et al. Type 1 diabetes mellitus: cellular and molecular pathophysiology at a glance. 2018.

61. Azzopardi P, Brown AD, Zimmet P, et al. Type 2 diabetes in young Indigenous Australians in rural and remote areas: diagnosis, screening, management and prevention. Med J Aust. 2012; 197: 32-36.

62. American Diabetes Association. Screening for type 2 diabetes. Diabetes Care. 2004; 27: s11-s14.

63. Stratton IM, Adler AI, Neil HAW, et al. Association of glycaemia with macrovascular and microvascular complications of type 2 diabetes (UKPDS 35): prospective observational study. BMJ. 2000; 321: 405-412.

64. Cole JB, Florez JC. Genetics of diabetes mellitus and diabetes complications. Nature Reviews: Nephrology. 2020; 16: 377-390.

65. Aune D, Norat T, Leitzmann M, et al. Physical activity and the risk of type 2 diabetes: a systematic review and dose–response meta-analysis. Eur J Epidemiol. 2015; 30: 529-542.

66. Eckel N, Mühlenbruch K, Meidtner K, et al. Characterization of metabolically unhealthy normalweight individuals: Risk factors and their associations with type 2 diabetes. Metabolism. 2015; 64: 862-871.

67. Diabetes Prevention Program Research Group. Reduction in the incidence of type 2 diabetes with lifestyle intervention or metformin. N Engl J Med. 2002; 2002: 393-403.

68. Espeland M. Reduction in weight and cardiovascular disease risk factors in individuals with type 2 diabetes: one-year results of the look AHEAD trial. Diabetes Care. 2007.

69. del Cañizo Gómez FJ, Andrés MNM. Strict control of modifiable cardiovascular risk factors in patients with type 2 diabetes mellitus. Med Clin (Barc). 2008; 130: 641-644.

70. Gæde P, Vedel P, Larsen N, et al. Multifactorial intervention and cardiovascular disease in patients with type 2 diabetes. N Engl J Med. 2003; 348: 383-393.

71. Martín-Timón I, Sevillano-Collantes C, Segura-Galindo A, et al. Type 2 diabetes and cardiovascular disease: have all risk factors the same strength? World J Diabetes. 2014; 5: 444.

72. Butalia S, Kaplan GG, Khokhar B, et al. Environmental risk factors and type 1 diabetes: past, present, and future. Can J Diabetes. 2016; 40: 586-593.

73. Cooke A. Review series on helminths, immune modulation and the hygiene hypothesis: how might infection modulate the onset of type 1 diabetes? Immunology. 2009; 126: 12-17.

74. Harding JL, Pavkov ME, Magliano DJ, et al. Global trends in diabetes complications: a review of current evidence. Diabetologia. 2019; 62: 3-16.

75. Punthakee Z, Goldenberg R, Katz P. Definition, classification and diagnosis of diabetes, prediabetes and metabolic syndrome. Can J Diabetes. 2018; 42: S10-S15.

76. Kim S. Burden of hospitalizations primarily due to uncontrolled diabetes: implications of inadequate primary health care in the United States. Diabetes Care. 2007; 30: 1281-1282.

77. Wexler DJ, Meigs JB, Cagliero E, et al. Prevalence of hyper-and hypoglycemia among inpatients with diabetes: a national survey of 44 US hospitals. Diabetes Care. 2007; 30: 367-369.

78. Varni JW, Delamater AM, Hood KK, et al. Diabetes symptoms predictors of health-related quality of life in adolescents and young adults with type 1 or type 2 diabetes. Qual Life Res. 2018; 27: 2295-2303.

79. King BR, Howard NJ, Verge CF, et al. A diabetes awareness campaign prevents diabetic ketoacidosis in children at their initial presentation with type 1 diabetes. Pediatr Diabetes. 2012; 13: 647-651.

80. Marcovecchio L. Complications of acute and chronic hyperglycemia. 2017.

81. Farsani SF, Brodovicz K, Soleymanlou N, et al. Incidence and prevalence of diabetic ketoacidosis (DKA) among adults with type 1 diabetes mellitus (T1D): a systematic literature review. BMJ. 2017; 7: e016587.

82. Umpierrez G, Korytkowski M. Diabetic emergencies—ketoacidosis, hyperglycaemic hyperosmolar state and hypoglycaemia. Nat Rev Endocrinol. 2016; 12: 222.

83. Chua HR, Schneider A, Bellomo R. Bicarbonate in diabetic ketoacidosis-a systematic review. Ann Inten Care. 2011; 1: 1-12.

84. Pasquel FJ, Umpierrez GE. Hyperosmolar hyperglycemic state: a historic review of the clinical presentation, diagnosis, and treatment. Diabetes Care. 2014; 37: 3124-3131.

85. Yale J-F, Paty B, Senior PA. Hypoglycemia. Can J Diabetes. 2018; 42: S104-S108.

86. Khunti K, Alsifri S, Aronson R, et al. Rates and predictors of hypoglycaemia in 27 585 people from 24 countries with insulin-treated type 1 and type 2 diabetes: the global HAT study. Diabetes Obes Metab. 2016; 18: 907-915.

87. Richard T, Issa R, Andres E, et al. latrogenic hypoglycemia–related hospital admissions identified through databases: economic burden and causes. Int J Clin Pharm. 2019; 41: 1159-1165.

88. Tourkmani AM, Alharbi TJ, Rsheed AMB, et al. Hypoglycemia in Type 2 Diabetes Mellitus patients: A review article. Diabetes Metab Syndr. 2018; 12: 791-794.

89. Rubega M, Sparacino G. Neurological changes in hypoglycemia. Diabetes Technol Ther. 2017; 19: 73-75.

90. Cryer PE. Symptoms of hypoglycemia, thresholds for their occurrence, and hypoglycemia unawareness. Endocrinol Metab Clin North Am. 1999; 28: 495-500.

91. Rombopoulos G, Hatzikou M, Latsou D, et al. The prevalence of hypoglycemia and its impact on the quality of life (QoL) of type 2 diabetes mellitus patients (The HYPO Study). Hormones. 2013; 12: 550-558.

92. Cryer PE. Mechanisms of hypoglycemia-associated autonomic failure and its component syndromes in diabetes. Diabetes. 2005; 54: 3592-3601.

93. International Hypoglycaemia Study Group. Minimizing hypoglycemia in diabetes. Diabetes Care. 2015; 38: 1583-1591.

94. Paes T, Rolim LC, Sallum Filho C, et al. Awareness of hypoglycemia and spectral analysis of heart rate variability in type 1 diabetes. J Diabetes Complications. 2020; 34: 107617.

95. Van Meijel LA, De Vegt F, Abbink EJ, et al. High prevalence of impaired awareness of hypoglycemia and severe hypoglycemia among people with insulin-treated type 2 diabetes: The Dutch Diabetes Pearl Cohort. BMJ Open Diabetes Research and Care. 2020; 8: e000935.

96. Little SA, Leelarathna L, Walkinshaw E, et al. Recovery of hypoglycemia awareness in longstanding type 1 diabetes: a multicenter 2× 2 factorial randomized controlled trial comparing insulin pump with multiple daily injections and continuous with conventional glucose self-monitoring (HypoCOMPaSS). Diabetes Care. 2014; 37: 2114-2122.

97. Cryer PE. latrogenic hypoglycemia as a cause of hypoglycemia-associated autonomic failure in IDDM: a vicious cycle. Diabetes. 1992; 41: 255-260.

98. Rickels MR. Hypoglycemia-associated autonomic failure, counterregulatory responses, and therapeutic options in type 1 diabetes. Ann N Y Acad Sci. 2019; 1454: 68.

99. Weinger K, Jacobson AM, Draelos MT, et al. Blood glucose estimation and symptoms during hyperglycemia and hypoglycemia in patients with insulin-dependent diabetes mellitus. Am J Med. 1995; 98: 22-31.

100. Gonder-Frederick L, Zrebiec J, Bauchowitz A, et al. Detection of hypoglycemia by children with type 1 diabetes 6 to 11 years of age and their parents: a field study. Pediatrics. 2008; 121: e489-e495.

101. Danne T, Nimri R, Battelino T, et al. International consensus on use of continuous glucose monitoring. Diabetes Care. 2017; 40: 1631-1640.

102. Goto A, Arah OA, Goto M, et al. Severe hypoglycaemia and cardiovascular disease: systematic review and meta-analysis with bias analysis. BMJ. 2013; 347.

103. Beagley J, Guariguata L, Weil C, et al. Global estimates of undiagnosed diabetes in adults. Diabetes Res Clin Pract. 2014; 103: 150-160.

104. Michaelides C, Daja M, Peterson D, et al. An HbA1c mapping tool helps identify where interventions and strategies for change need to be targeted. The Mapping Glycaemic Control Across Australia (MGCAA) project. 2012.

105. Cardoso CR, Salles GF. Predictors of development and progression of microvascular complications in a cohort of Brazilian type 2 diabetic patients. J Diabetes Complications. 2008; 22: 164-170.

106. Motala AA, Pirie FJ, Gouws E, et al. Microvascular complications in South African patients with long duration diabetes mellitus. S Afr Med J. 2001; 91: 987-992.

107. Zoungas S, Woodward M, Li Q, et al. Impact of age, age at diagnosis and duration of diabetes on the risk of macrovascular and microvascular complications and death in type 2 diabetes. Diabetologia. 2014; 57: 2465-2474.

108. Chawla A, Chawla R, Jaggi S. Microvasular and macrovascular complications in diabetes mellitus: distinct or continuum? Indian J Endocrinol Metab. 2016; 20: 546.

109. Al-Wakeel JS, Hammad D, Al Suwaida A, et al. Microvascular and macrovascular complications in diabetic nephropathy patients referred to nephrology clinic. Saudi J Kidney Dis Transpl. 2009; 20: 77.

110. Krentz AJ, Clough G, Byrne CD. Interactions between microvascular and macrovascular disease in diabetes: pathophysiology and therapeutic implications. Diabetes Obes Metab. 2007; 9: 781-791.

111. Inzucchi SE, Bergenstal RM, Buse JB, et al. Management of hyperglycemia in type 2 diabetes: a patient-centered approach. Diabetes Care. 2012; 35: 1364-1379.

112. Pettus JH, Zhou FL, Shepherd L, et al. Incidences of severe hypoglycemia and diabetic ketoacidosis and prevalence of microvascular complications stratified by age and glycemic control in US adult patients with type 1 diabetes: a real-world study. Diabetes Care. 2019; 42: 2220-2227.

113. Brownlee M. The pathobiology of diabetic complications a unifying mechanism. Diabetes. 2005; 54: 1615-1625.

114. Pop-Busui R, Evans GW, Gerstein HC, et al. Effects of cardiac autonomic dysfunction on mortality risk in the Action to Control Cardiovascular Risk in Diabetes (ACCORD) trial. Diabetes Care. 2010; 33: 1578-1584.

115. Vinik AI, Erbas T, Casellini CM. Diabetic cardiac autonomic neuropathy, inflammation and cardiovascular disease. J Diabetes Invest. 2013; 4: 4-18.
116. Wulsin LR, Horn PS, Perry JL, et al. Autonomic imbalance as a predictor of metabolic risks, cardiovascular disease, diabetes, and mortality. J Clin Endocrinol Metab. 2015; 100: 2443-2448.

117. Ewing D, Irving J, Kerr F, et al. Cardiovascular responses to sustained handgrip in normal subjects and in patients with diabetes mellitus: a test of autonomic function. Clin Sci. 1974; 46: 295-306.

118. Said G. Handbook of Clinical Neurology. Said G, Krarup C, editors2013.

119. Dafaalla MD, Nimir MN, Mohammed MI, et al. Risk factors of diabetic cardiac autonomic neuropathy in patients with type 1 diabetes mellitus: a meta-analysis. Open Heart. 2016; 3: e000336. 120. Van Dam PS, Cotter MA, Bravenboer B, et al. Pathogenesis of diabetic neuropathy: focus on neurovascular mechanisms. Eur J Pharmacol. 2013; 719: 180-186.

121. Gibbons GW, Shaw PM, editors. Diabetic vascular disease: characteristics of vascular disease unique to the diabetic patient. Semin Vasc Surg; 2012: Elsevier.

122. Ndip A, Lavery LA, Boulton AJ. Diabetic foot disease in people with advanced nephropathy and those on renal dialysis. Curr Diab Rep. 2010; 10: 283-290.

123. Morbach S, Furchert H, Gröblinghoff U, et al. Long-Term Prognosis of Diabetic Foot Patients and Their Limbs Amputation and death over the course of a decade. Diabetes Care. 2012; 35: 2021-2027. 124. Kamenov ZA, Traykov LD. Diabetic autonomic neuropathy. Diabetes: Springer; 2013. p. 176-193.

125. Bray GA. Obesity, a disorder of nutrient partitioning: the MONA LISA hypothesis. J Nutr. 1991; 121: 1146-1162.

126. Lagani V, Koumakis L, Chiarugi F, et al. A systematic review of predictive risk models for diabetes complications based on large scale clinical studies. J Diabetes Complications. 2013; 27: 407-413.

127. Balcıoğlu AS, Müderrisoğlu H. Diabetes and cardiac autonomic neuropathy: clinical manifestations, cardiovascular consequences, diagnosis and treatment. World J Diabetes. 2015; 6: 80.

128. Spallone V. Update on the impact, diagnosis and management of cardiovascular autonomic neuropathy in diabetes: what is defined, what is new, and what is unmet. Diabetes Metab J. 2019; 43: 3-30.

129. Nonogaki K. New insights into sympathetic regulation of glucose and fat metabolism. Diabetologia. 2000; 43: 533-549.

130. Morton GJ, Muta K, Kaiyala KJ, et al. Evidence that the sympathetic nervous system elicits rapid, coordinated, and reciprocal adjustments of insulin secretion and insulin sensitivity during cold exposure. Diabetes. 2017; 66: 823-834.

131. Shimazu T. Innervation of the liver and glucoregulation: roles of the hypothalamus and autonomic nerves. Nutrition (Burbank, Los Angeles County, Calif). 1996; 12: 65-66.

132. Nonogaki K, Iguchi A. Role of central neural mechanisms in the regulation of hepatic glucose metabolism. Life Sci. 1997; 60: 797-807.

133. Gregg EW, Sattar N, Ali MK. The changing face of diabetes complications. The Lancet Diabetes & endocrinology. 2016; 4: 537-547.

134. Kantharidis P, Wang B, Carew RM, et al. Diabetes complications: the microRNA perspective. Diabetes. 2011; 60: 1832-1837.

135. Salvotelli L, Stoico V, Perrone F, et al. Prevalence of neuropathy in type 2 diabetic patients and its association with other diabetes complications: The Verona Diabetic Foot Screening Program. J Diabetes Complications. 2015; 29: 1066-1070.

136. Pickup J, Sutton A. Severe hypoglycaemia and glycaemic control in Type 1 diabetes: meta-analysis of multiple daily insulin injections compared with continuous subcutaneous insulin infusion. Diabet Med. 2008; 25: 765-774.

137. Ray KK, Seshasai SRK, Wijesuriya S, et al. Effect of intensive control of glucose on cardiovascular outcomes and death in patients with diabetes mellitus: a meta-analysis of randomised controlled trials. The Lancet. 2009; 373: 1765-1772.

138. Shima K, Komatsu M, Kawahara K, et al. Stringent glycaemic control prolongs survival in diabetic patients with end-stage renal disease on haemodialysis. Nephrology. 2010; 15: 632-638.

139. Group AC. Intensive blood glucose control and vascular outcomes in patients with type 2 diabetes. N Engl J Med. 2008; 358: 2560-2572.

140. Hayward RA, Reaven PD, Wiitala WL, et al. Follow-up of glycemic control and cardiovascular outcomes in type 2 diabetes. N Engl J Med. 2015; 372: 2197-2206.

141. Holman RR, Paul SK, Bethel MA, et al. 10-year follow-up of intensive glucose control in type 2 diabetes. N Engl J Med. 2008; 359: 1577-1589.

142. Nathan DM, Zinman B, Cleary PA, et al. Modern-day clinical course of type 1 diabetes mellitus after 30 years' duration: the diabetes control and complications trial/epidemiology of diabetes interventions and complications and Pittsburgh epidemiology of diabetes complications experience (1983-2005). Arch Intern Med. 2009; 169: 1307.

143. Ceriello A, Monnier L, Owens D. Glycaemic variability in diabetes: clinical and therapeutic implications. The Lancet Diabetes & endocrinology. 2019; 7: 221-230.

144. Wilmot EG, Choudhary P, Leelarathna L, et al. Glycaemic variability: The under-recognized therapeutic target in type 1 diabetes care. Diabetes Obes Metab. 2019; 21: 2599-2608.

145. Muggeo M, Zoppini G, Bonora E, et al. Fasting plasma glucose variability predicts 10-year survival of type 2 diabetic patients: the Verona Diabetes Study. Diabetes Care. 2000; 23: 45-50.

146. Kalopita S, Liatis S, Thomakos P, et al. Relationship between autonomic nervous system function and continuous interstitial glucose measurement in patients with type 2 diabetes. J Diabetes Res. 2014; 2014: 1-6.

147. Nusca A, Tuccinardi D, Albano M, et al. Glycemic variability in the development of cardiovascular complications in diabetes. Diabetes Metab Res Rev. 2018; 34: e3047.

148. Giugliano D, Ceriello A, Paolisso G. Oxidative stress and diabetic vascular complications. Diabetes Care. 1996; 19: 257-267.

149. Niedowicz DM, Daleke DL. The role of oxidative stress in diabetic complications. Cell Biochem Biophys. 2005; 43: 289-330.

150. Wu J-S, Yang Y-C, Lin T-S, et al. Epidemiological evidence of altered cardiac autonomic function in subjects with impaired glucose tolerance but not isolated impaired fasting glucose. J Clin Endocrinol Metab. 2007; 92: 3885-3889.

151. Jin HY, Baek HS, Park TS. Morphologic changes in autonomic nerves in diabetic autonomic neuropathy. Diabetes Metab J. 2015; 39: 461-467.

152. Ewing DJ, Martyn CN, Young RJ, et al. The value of cardiovascular autonomic function tests: 10 years experience in diabetes. Diabetes Care. 1985; 8: 491-498.

153. Pafili K, Trypsianis G, Papazoglou D, et al. Simplified diagnosis of cardiovascular autonomic neuropathy in type 2 diabetes using Ewing's battery. The review of diabetic studies: RDS. 2015; 12: 213.

154. Zhou J, Zhou S. Inflammation: therapeutic targets for diabetic neuropathy. Mol Neurobiol. 2014; 49: 536-546.

155. Bierhaus A, Nawroth P. Multiple levels of regulation determine the role of the receptor for AGE (RAGE) as common soil in inflammation, immune responses and diabetes mellitus and its complications. Diabetologia. 2009; 52: 2251-2263.

156. Negi G, Kumar A, Sharma SS. Melatonin modulates neuroinflammation and oxidative stress in experimental diabetic neuropathy: effects on NF-κB and Nrf2 cascades. J Pineal Res. 2011; 50: 124-131.

157. Olokoba AB, Obateru OA, Olokoba LB. Type 2 diabetes mellitus: a review of current trends. Oman Med J. 2012; 27: 269.

158. Yaribeygi H, Farrokhi FR, Butler AE, et al. Insulin resistance: Review of the underlying molecular mechanisms. J Cell Physiol. 2019; 234: 8152-8161.

159. Riddle MC, Cefalu WT, Evans PH, et al. Consensus report: definition and interpretation of remission in type 2 diabetes. Diabetologia. 2021: 1-8.

160. Bergenstal RM, Tamborlane WV, Ahmann A, et al. Effectiveness of sensor-augmented insulinpump therapy in type 1 diabetes. N Engl J Med. 2010; 363: 311-320.

161. Chamberlain JJ, Rhinehart AS, Shaefer CF, et al. Diagnosis and Management of Diabetes: Synopsis of the 2016 American Diabetes Association Standards of Medical Care in DiabetesSynopsis of the 2016 ADA Standards of Medical Care in Diabetes. Ann Intern Med. 2016; 164: 542-552.

162. Chellappan DK, Sivam NS, Teoh KX, et al. Gene therapy and type 1 diabetes mellitus. Biomed Pharmacother. 2018; 108: 1188-1200.

163. Asche C, LaFleur J, Conner C. A review of diabetes treatment adherence and the association with clinical and economic outcomes. Clin Ther. 2011; 33: 74-109.

164. Coppelli A, Giannarelli R, Vistoli F, et al. The beneficial effects of pancreas transplant alone on diabetic nephropathy. Diabetes Care. 2005; 28: 1366-1370.

165. White SA, Shaw JA, Sutherland DE. Pancreas transplantation. The Lancet. 2009; 373: 1808-1817. 166. Bruni A, Gala-Lopez B, Pepper AR, et al. Islet cell transplantation for the treatment of type 1 diabetes: recent advances and future challenges. Diabetes Metab Syndr Obes Targ Ther. 2014; 7: 211.

167. Lemos JR, Baidal DA, Ricordi C, et al. Survival after islet transplantation in subjects with type 1 diabetes: twenty-year follow-up. Diabetes Care. 2021; 44: e67-e68.

168. Esmeijer K, Hoogeveen EK, van den Boog PJ, et al. Superior long-term survival for simultaneous pancreas-kidney transplantation as renal replacement therapy: 30-year follow-up of a nationwide cohort. Diabetes Care. 2020; 43: 321-328.

169. Gruessner AC, Sutherland DE, Gruessner RW. Long-term outcome after pancreas transplantation. Current opinion in organ transplantation. 2012; 17: 100-105.

170. Ihnat M, Thorpe J, Ceriello A. Hypothesis: the" metabolic memory"-the new challenge of diabetes. Clinical Diabetology. 2007; 8: 289-294.

171. Hemmingsen B, Gimenez-Perez G, Mauricio D, et al. Diet, physical activity or both for prevention or delay of type 2 diabetes mellitus and its associated complications in people at increased risk of developing type 2 diabetes mellitus. Cochrane Database of Systematic Reviews. 2017.

172. Dyson P, Kelly T, Deakin T, et al. Diabetes UK evidence-based nutrition guidelines for the prevention and management of diabetes. Diabet Med. 2011; 28: 1282-1288.

173. Pawlak R. Vegetarian diets in the prevention and management of diabetes and its complications. Diabetes Spectr. 2017; 30: 82-88.

174. Rice K, Te Hiwi B, Zwarenstein M, et al. Best practices for the prevention and management of diabetes and obesity-related chronic disease among Indigenous peoples in Canada: a review. Can J Diabetes. 2016; 40: 216-225.

175. Jalving AC, Gant CM, Binnenmars SH, et al. Glycaemic control in the diabetes and Lifestyle Cohort Twente: A cross-sectional assessment of lifestyle and pharmacological management on Hba1c target achievement. Diabetes Obes Metab. 2018; 20: 2494-2499.

176. Gregg EW, Chen H, Wagenknecht LE, et al. Association of an intensive lifestyle intervention with remission of type 2 diabetes. JAMA. 2012; 308: 2489-2496.

177. Pournaras D, Aasheim E, Søvik T, et al. Effect of the definition of type II diabetes remission in the evaluation of bariatric surgery for metabolic disorders. Journal of British Surgery. 2012; 99: 100-103.

178. Hostalek U, Gwilt M, Hildemann S. Therapeutic use of metformin in prediabetes and diabetes prevention. Drugs. 2015; 75: 1071-1094.

179. Esteghamati A, Eskandari D, Mirmiranpour H, et al. Effects of metformin on markers of oxidative stress and antioxidant reserve in patients with newly diagnosed type 2 diabetes: a randomized clinical trial. Clin Nutr. 2013; 32: 179-185.

180. Faure P, Wiernsperger N, Polge C, et al. Impairment of the antioxidant properties of serum albumin in patients with diabetes: protective effects of metformin. Clin Sci. 2008; 114: 251-256.

181. Sanchez-Rangel E, Inzucchi SE. Metformin: clinical use in type 2 diabetes. Diabetologia. 2017; 60: 1586-1593.

182. Manzella D, Grella R, Esposito K, et al. Blood pressure and cardiac autonomic nervous system in obese type 2 diabetic patients: effect of metformin administration. Am J Hypertens. 2004; 17: 223-227.

183. Billeter A, Scheurlen K, Probst P, et al. Meta-analysis of metabolic surgery versus medical treatment for microvascular complications in patients with type 2 diabetes mellitus. Journal of British Surgery. 2018; 105: 168-181.

184. Chang S-H, Stoll CR, Song J, et al. The effectiveness and risks of bariatric surgery: an updated systematic review and meta-analysis, 2003-2012. JAMA surgery. 2014; 149: 275-287.

185. Chaudhury A, Duvoor C, Reddy Dendi VS, et al. Clinical review of antidiabetic drugs: implications for type 2 diabetes mellitus management. Front Endocrinol (Lausanne). 2017; 8: 6.

186. Jeschke MG, Kraft R, Emdad F, et al. Glucose control in severely thermally injured pediatric patients: what glucose range should be the target? Ann Surg. 2010; 252.

187. Heikkinen S, Argmann CA, Champy MF, et al. Evaluation of glucose homeostasis. Curr Protoc Mol Biol. 2007; 77: 29B. 23.21-29B. 23.22.

188. Kelly TN, Bazzano LA, Fonseca VA, et al. Systematic review: glucose control and cardiovascular disease in type 2 diabetes. Ann Intern Med. 2009; 151: 394-403.

189. Roder PV, Wu B, Liu Y, et al. Pancreatic regulation of glucose homeostasis. Exp Mol Med. 2016; 48: 1-19.

190. Vervloet M, Spreeuwenberg P, Bouvy M, et al. Lazy Sunday afternoons: the negative impact of interruptions in patients' daily routine on adherence to oral antidiabetic medication. A multilevel analysis of electronic monitoring data. Eur J Clin Pharmacol. 2013; 69: 1599-1606.

191. Bastyr Er, Stuart CA, Brodows RG, et al. Therapy focused on lowering postprandial glucose, not fasting glucose, may be superior for lowering HbA1c. IOEZ Study Group. Diabetes Care. 2000; 23: 1236-1241.

192. Jiang J, Zhao L, Lin L, et al. Postprandial blood glucose outweighs fasting blood glucose and HbA1c in screening coronary heart disease. Sci Rep. 2017; 7: 1-7.

193. Shaye K, Amir T, Shlomo S, et al. Fasting glucose levels within the high normal range predict cardiovascular outcome. Am Heart J. 2012; 164: 111-116.

194. Bowen ME, Xuan L, Lingvay I, et al. Doc, I Just Ate: interpreting random blood glucose values in patients with unknown glycemic status. J Gen Intern Med. 2018; 33: 142-144.

195. Cavalot F, Petrelli A, Traversa M, et al. Postprandial blood glucose is a stronger predictor of cardiovascular events than fasting blood glucose in type 2 diabetes mellitus, particularly in women: lessons from the San Luigi Gonzaga Diabetes Study. J Clin Endocrinol Metab. 2006; 91: 813-819.

196. Amiel SA, Pursey N, Higgins B, et al. Diagnosis and management of type 1 diabetes in adults: summary of updated NICE guidance. BMJ. 2015; 351.

197. Akasaka T, Sueta D, Tabata N, et al. Effects of the mean amplitude of glycemic excursions and vascular endothelial dysfunction on cardiovascular events in nondiabetic patients with coronary artery disease. Journal of the American Heart Association. 2017; 6: e004841.

198. Carlson AL, Mullen DM, Bergenstal RM. Clinical use of continuous glucose monitoring in adults with type 2 diabetes. Diabetes Technol Ther. 2017; 19: S-4-S-11.

199. Koschinsky T, Jungheim K, Heinemann L. Glucose sensors and the alternate site testing-like phenomenon: relationship between rapid blood glucose changes and glucose sensor signals. Diabetes Technol Ther. 2003; 5: 829-842.

200. Al-Zahrani A, Alshareef R, Farahat F, et al. Assessing the Accuracy of Different Glucometers Based on the Laboratory Reference Method. Clin Lab. 2020: 1.

201. Andelin M, Kropff J, Matuleviciene V, et al. Assessing the accuracy of continuous glucose monitoring (CGM) calibrated with capillary values using capillary or venous glucose levels as a reference. J Diabetes Sci Technol. 2016; 10: 876-884.

202. Villena Gonzales W, Mobashsher AT, Abbosh A. The progress of glucose monitoring—A review of invasive to minimally and non-invasive techniques, devices and sensors. Sensors. 2019; 19: 800.

203. World Health Organization. WHO guidelines on drawing blood: best practices in phlebotomy. Geneva, Switzerland; 2010.

204. Madeo B, Scaltriti S, Romano S, et al., editors. Self-monitoring of blood glucose in type 2 diabetes mellitus: effect on glycemic control in non-insulin-treated patients. Endocrine Abstracts; 2011: Bioscientifica.

205. Klonoff DC, Blonde L, Cembrowski G, et al. Consensus report: the current role of self-monitoring of blood glucose in non-insulin-treated type 2 diabetes. J Diabetes Sci Technol. 2011; 5: 1529-1548.

206. Halldorsdottir S, Warchal-Windham ME, Wallace JF, et al. Accuracy evaluation of five blood glucose monitoring systems: the North American Comparator Trial. J Diabetes Sci Technol. 2013; 7: 1294-1304.

207. Tack C, Pohlmeier H, Behnke T, et al. Accuracy evaluation of five blood glucose monitoring systems obtained from the pharmacy: a European multicenter study with 453 subjects. Diabetes Technol Ther. 2012; 14: 330-337.

208. Tunis SL, Willis WD, Foos V. Self-monitoring of blood glucose (SMBG) in patients with type 2 diabetes on oral anti-diabetes drugs: cost-effectiveness in France, Germany, Italy, and Spain. Curr Med Res Opin. 2010; 26: 163-175.

209. Farmer A, Wade A, Goyder E, et al. Impact of self monitoring of blood glucose in the management of patients with non-insulin treated diabetes: open parallel group randomised trial. BMJ. 2007; 335: 132.

210. Allemann S, Houriet C, Diem P, et al. Self-monitoring of blood glucose in non-insulin treated patients with type 2 diabetes: a systematic review and meta-analysis. Curr Med Res Opin. 2009; 25: 2903-2913.

211. Nagelkerk J, Reick K, Meengs L. Perceived barriers and effective strategies to diabetes self-management. J Adv Nurs. 2006; 54: 151-158.

212. Nair KM, Levine MA, Lohfeld LH, et al. "I take what I think works for me": a qualitative study to explore patient perception of diabetes treatment benefits and risks. J Popul Ther Clin Pharmacol. 2007; 14.

213. Sacks DB. A1C versus glucose testing: a comparison. Diabetes Care. 2011; 34: 518-523.

214. Fisher WA, Cornman DH, Kohut T, et al. What primary care providers can do to address barriers to self-monitoring of blood glucose. Clin Diabetes. 2013; 31: 34-42.

215. Hallberg SJ, McKenzie AL, Williams PT, et al. Effectiveness and safety of a novel care model for the management of type 2 diabetes at 1 year: an open-label, non-randomized, controlled study. Diabetes Ther. 2018; 9: 583-612.

216. Machry RV, Rados DV, de Gregorio GR, et al. Self-monitoring blood glucose improves glycemic control in type 2 diabetes without intensive treatment: A systematic review and meta-analysis. Diabetes Res Clin Pract. 2018; 142: 173-187.

217. Renard E. Monitoring glycemic control: the importance of self-monitoring of blood glucose. Am J Med. 2005; 118: 12-19.

218. Farmer A, Wade A, French DP, et al. Blood glucose self-monitoring in type 2 diabetes: a randomised controlled trial. Health Technol Assess. 2009; 13: 1-50.

219. Clar C, Barnard K, Cummins E, et al. Self-monitoring of blood glucose in type 2 diabetes: systematic review. Health technology assessment (Winchester, England). 2010; 14: 1-140.

220. St John A, Davis WA, Price CP, et al. The value of self-monitoring of blood glucose: a review of recent evidence. J Diabetes Complications. 2010; 24: 129-141.

221. Tunis SL, Minshall ME. Self-monitoring of blood glucose in type 2 diabetes: cost-effectiveness in the united states. Am J Manag Care. 2008; 14: 131-140.

222. Puhr S, Derdzinski M, Welsh JB, et al. Real-world hypoglycemia avoidance with a continuous glucose monitoring system's predictive low glucose alert. Diabetes Technol Ther. 2019; 21: 155-158.

223. Wolpert HA. Use of continuous glucose monitoring in the detection and prevention of hypoglycemia. SAGE Publications; 2007.

224. Juvenile Diabetes Research Foundation Continuous Glucose Monitoring Study Group. Continuous glucose monitoring and intensive treatment of type 1 diabetes. The New England Journal of Medicine. 2008; 2008: 1464-1476.

225. Boyne MS, Silver DM, Kaplan J, et al. Timing of changes in interstitial and venous blood glucose measured with a continuous subcutaneous glucose sensor. Diabetes. 2003; 52: 2790-2794.

226. Pickup JC, Hussain F, Evans ND, et al. In vivo glucose monitoring: the clinical reality and the promise. Biosens Bioelectron. 2005; 20: 1897-1902.

227. Medtronic MiniMed I. Sensors & Transmitters 2017 [Available from: <u>https://www.medtronicdiabetes.com/customer-support/sensors-and-transmitters-</u> support/calibration-sensor.

228. Hommel E, Olsen B, Battelino T, et al. Impact of continuous glucose monitoring on quality of life, treatment satisfaction, and use of medical care resources: analyses from the SWITCH study. Acta Diabetol. 2014; 51: 845-851.

229. Beck RW, Riddlesworth TD, Ruedy K, et al. Continuous glucose monitoring versus usual care in patients with type 2 diabetes receiving multiple daily insulin injections: a randomized trial. Ann Intern Med. 2017; 167: 365-374.

230. Group JDRFCGMS. Effectiveness of continuous glucose monitoring in a clinical care environment: evidence from the Juvenile Diabetes Research Foundation continuous glucose monitoring (JDRF-CGM) trial. Diabetes Care. 2010; 33: 17-22.

231. Raccah D, Sulmont V, Reznik Y, et al. Incremental value of continuous glucose monitoring when starting pump therapy in patients with poorly controlled type 1 diabetes: the RealTrend study. Diabetes Care. 2009; 32: 2245-2250.

232. Nalysnyk L, Hernandez-Medina M, Krishnarajah G. Glycaemic variability and complications in patients with diabetes mellitus: evidence from a systematic review of the literature. Diabetes Obes Metab. 2010; 12: 288-298.

233. Zammitt NN, Frier BM. Hypoglycemia in type 2 diabetes. Diabetes Care. 2005; 28: 2948-2961.

234. Smith-Palmer J, Brändle M, Trevisan R, et al. Assessment of the association between glycemic variability and diabetes-related complications in type 1 and type 2 diabetes. Diabetes Res Clin Pract. 2014; 105: 273-284.

235. Klimontov VV, Myakina NE. Glucose variability indices predict the episodes of nocturnal hypoglycemia in elderly type 2 diabetic patients treated with insulin. Diabetes Metab Syndr. 2017; 11: 119-124.

236. Heinemann L, Fleming GA, Petrie JR, et al. Insulin pump risks and benefits: a clinical appraisal of pump safety standards, adverse event reporting and research needs. A joint statement of the European Association for the Study of Diabetes and the American Diabetes Association Diabetes Technology Working Group. Diabetologia. 2015; 58: 862-870.

237. Powers MA, Bardsley J, Cypress M, et al. Diabetes self-management education and support in type 2 diabetes: a joint position statement of the American Diabetes Association, the American Association of Diabetes Educators, and the Academy of Nutrition and Dietetics. The Diabetes Educator. 2017; 43: 40-53.

238. Mian Z, Hermayer KL, Jenkins A. Continuous glucose monitoring: review of an innovation in diabetes management. Am J Med Sci. 2019; 358: 332-339.

239. Hepburn DA, Deary IJ, Frier BM, et al. Symptoms of acute insulin-induced hypoglycemia in humans with and without IDDM: factor-analysis approach. Diabetes Care. 1991; 14: 949-957.

240. Edelman SV. Regulation catches up to reality: nonadjunctive use of continuous glucose monitoring data. J Diabetes Sci Technol. 2017; 11: 160-164.

241. Lal RA, Maahs DM. Clinical Use of Continuous Glucose Monitoring in Pediatrics. Diabetes Technol Ther. 2017; 19: S-37-S-43.

242. Rodbard D. Continuous glucose monitoring: a review of recent studies demonstrating improved glycemic outcomes. Diabetes Technol Ther. 2017; 19: S-25-S-37.

243. Kropff J, Choudhary P, Neupane S, et al. Accuracy and longevity of an implantable continuous glucose sensor in the PRECISE study: a 180-day, prospective, multicenter, pivotal trial. Diabetes Care. 2017; 40: 63-68.

244. Christiansen MP, Klaff LJ, Brazg R, et al. A prospective multicenter evaluation of the accuracy of a novel implanted continuous glucose sensor: PRECISE II. Diabetes Technol Ther. 2018; 20: 197-206.

245. Kovatchev B, Anderson S, Heinemann L, et al. Comparison of the numerical and clinical accuracy of four continuous glucose monitors. Diabetes Care. 2008; 31: 1160-1164.

246. Francescato M, Geat M, Stel G, et al. Accuracy of a portable glucose meter and of a continuous glucose monitoring device used at home by patients with type 1 diabetes. Clin Chim Acta. 2012; 413: 312-318.

247. Sonmez A, Yilmaz Z, Uckaya G, et al. The accuracy of home glucose meters in hypoglycemia. Diabetes Technol Ther. 2010; 12: 619-626.

248. Kropff J, Bruttomesso D, Doll W, et al. Accuracy of two continuous glucose monitoring systems: a head-to-head comparison under clinical research centre and daily life conditions. Diabetes Obes Metab. 2015; 17: 343-349.

249. Guerci B, Floriot M, Böhme P, et al. Clinical performance of CGMS in type 1 diabetic patients treated by continuous subcutaneous insulin infusion using insulin analogs. Diabetes Care. 2003; 26: 582-589.

250. Bailey TS, Alva S. Landscape of continuous glucose monitoring (CGM) and integrated CGM: accuracy considerations. Diabetes Technol Ther. 2021; 23: S-5-S-11.

251. Welsh JB, Gao P, Derdzinski M, et al. Accuracy, utilization, and effectiveness comparisons of different continuous glucose monitoring systems. Diabetes Technol Ther. 2019; 21: 128-132.

252. McQueen RB, Ellis SL, Campbell JD, et al. Cost-effectiveness of continuous glucose monitoring and intensive insulin therapy for type 1 diabetes. Cost Eff Res Allo. 2011; 9: 1-8.

253. Roze S, Isitt JJ, Smith-Palmer J, et al. Evaluation of the Long-Term Cost-Effectiveness of the Dexcom G6 Continuous Glucose Monitor versus Self-Monitoring of Blood Glucose in People with Type 1 Diabetes in Canada. ClinicoEconomics and Outcomes Research: CEOR. 2021; 13: 717.

254. Wan W, Skandari MR, Minc A, et al. Cost-effectiveness of continuous glucose monitoring for adults with type 1 diabetes compared with self-monitoring of blood glucose: the DIAMOND randomized trial. Diabetes Care. 2018; 41: 1227-1234.

255. Chaugule S, Graham C. Cost-effectiveness of G5 Mobile continuous glucose monitoring device compared to self-monitoring of blood glucose alone for people with type 1 diabetes from the Canadian societal perspective. J Med Econ. 2017; 20: 1128-1135.

256. Fonda SJ, Graham C, Munakata J, et al. The cost-effectiveness of real-time continuous glucose monitoring (RT-CGM) in type 2 diabetes. J Diabetes Sci Technol. 2016; 10: 898-904.

257. Schrangl P, Reiterer F, Heinemann L, et al. Limits to the evaluation of the accuracy of continuous glucose monitoring systems by clinical trials. Biosensors. 2018; 8: 50.

258. Chung ST, Ha J, Onuzuruike AU, et al. Time to glucose peak during an oral glucose tolerance test identifies prediabetes risk. Clin Endocrinol (Oxf). 2017; 87: 484-491.

259. Eyth E, Basit H, Smith CJ. Glucose tolerance test. StatPearls [Internet]. 2020.

260. Hulman A, Vistisen D, Glümer C, et al. Glucose patterns during an oral glucose tolerance test and associations with future diabetes, cardiovascular disease and all-cause mortality rate. Diabetologia. 2018; 61: 101-107.

261. Pareek M, Bhatt DL, Nielsen ML, et al. Enhanced predictive capability of a 1-hour oral glucose tolerance test: a prospective population-based cohort study. Diabetes Care. 2018; 41: 171-177.

262. Sherwani SI, Khan HA, Ekhzaimy A, et al. Significance of HbA1c test in diagnosis and prognosis of diabetic patients. Biomarker insights. 2016; 11: BMI. S38440.

263. Weykamp C. HbA1c: a review of analytical and clinical aspects. Ann Lab Med. 2013; 33: 393-400.

264. Gupta S, Jain U, Chauhan N. Laboratory diagnosis of HbA1c: a review. J Nanomed Res. 2017; 5: 00120.

265. Khan H, Sobki S, Khan S. Association between glycaemic control and serum lipids profile in type 2 diabetic patients: HbA 1c predicts dyslipidaemia. Clin Exp Med. 2007; 7: 24-29.

266. Beck RW, Connor CG, Mullen DM, et al. The fallacy of average: how using HbA1c alone to assess glycemic control can be misleading. Diabetes Care. 2017; 40: 994-999.

267. Kilpatrick ES, Rigby AS, Atkin SL. Variability in the relationship between mean plasma glucose and HbA1c: implications for the assessment of glycemic control. Clin Chem. 2007; 53: 897-901.

268. Ohde S, Deshpande GA, Yokomichi H, et al. HbA1c monitoring interval in patients on treatment for stable type 2 diabetes. A ten-year retrospective, open cohort study. Diabetes Res Clin Pract. 2018; 135: 166-171.

269. Tankova T, Chakarova N, Dakovska L, et al. Assessment of HbA1c as a diagnostic tool in diabetes and prediabetes. Acta Diabetol. 2012; 49: 371-378.

270. Nathan DM, Group DER. The diabetes control and complications trial/epidemiology of diabetes interventions and complications study at 30 years: overview. Diabetes Care. 2014; 37: 9-16.

271. Adler AI, Stevens RJ, Manley SE, et al. Development and progression of nephropathy in type 2 diabetes: the United Kingdom Prospective Diabetes Study (UKPDS 64). Kidney Int. 2003; 63: 225-232. 272. Hirsch IB, Brownlee M. Should minimal blood glucose variability become the gold standard of glycemic control? J Diabetes Complications. 2005; 19: 178-181.

273. Klonoff DC. Continuous glucose monitoring roadmap for 21st century diabetes therapy. Diabetes Care. 2005; 28: 1231-1239.

274. Bragd J, Adamson U, Bäcklund L, et al. Can glycaemic variability, as calculated from blood glucose self-monitoring, predict the development of complications in type 1 diabetes over a decade? Diabetes Metab. 2008; 34: 612-616.

275. Frontoni S, Di Bartolo P, Avogaro A, et al. Glucose variability: An emerging target for the treatment of diabetes mellitus. Diabetes Res Clin Pract. 2013; 102: 86-95.

276. Chehregosha H, Khamseh ME, Malek M, et al. A view beyond HbA1c: role of continuous glucose monitoring. Diabetes Ther. 2019; 10: 853-863.

277. Gunton JE, Cheung NW, Davis TME, et al. A new blood glucose management algorithm for type 2 diabetes: a position statement of the Australian Diabetes Society. Med J Aust. 2014; 201: 650-653.

278. O'sullivan C, Hynes N, Mahendran B, et al. Haemoglobin A1c (HbA1C) in non-diabetic and diabetic vascular patients. Is HbA1C an independent risk factor and predictor of adverse outcome? Eur J Vasc Endovasc Surg. 2006; 32: 188-197.

279. Kharal M, Al-Hajjaj A, Al-Ammri M, et al. Meeting the American Diabetic Association standards of diabetic care. Saudi J Kidney Dis Transpl. 2010; 21: 678.

280. Lavernia F, Adkins SE, Shubrook JH. Use of oral combination therapy for type 2 diabetes in primary care: meeting individualized patient goals. Postgrad Med. 2015; 127: 808-817.

281. Stone MA, Charpentier G, Doggen K, et al. Quality of care of people with type 2 diabetes in eight European countries: findings from the Guideline Adherence to Enhance Care (GUIDANCE) study. Diabetes Care. 2013; 36: 2628-2638.

282. American Diabetes Association. 2. Classification and diagnosis of diabetes. Diabetes Care. 2015; 38: S8-S16.

283. Costantino S, Paneni F, Battista R, et al. Impact of glycemic variability on chromatin remodeling, oxidative stress, and endothelial dysfunction in patients with type 2 diabetes and with target HbA1c levels. Diabetes. 2017; 66: 2472-2482.

284. Litwak L, Goh S-Y, Hussein Z, et al. Prevalence of diabetes complications in people with type 2 diabetes mellitus and its association with baseline characteristics in the multinational A 1 chieve study. Diabetol Metab Syndr. 2013; 5: 1.

285. Foster NC, Beck RW, Miller KM, et al. State of type 1 diabetes management and outcomes from the T1D exchange in 2016–2018. Diabetes Technol Ther. 2019; 21: 66-72.

286. Grenier J, Leiter LA, Langer A, et al. Glycaemic control and cardiovascular risk factor management in patients with diabetes with and without coronary artery disease: insights from the diabetes mellitus status in Canada survey. European Heart Journal–Quality of Care and Clinical Outcomes. 2016; 2: 277-284.

287. Giugliano D, Maiorino M, Bellastella G, et al. Clinical inertia, reverse clinical inertia, and medication non-adherence in type 2 diabetes. J Endocrinol Invest. 2019; 42: 495-503.

288. Lindner N, Kuwabara A, Holt T. Non-invasive and minimally invasive glucose monitoring devices: a systematic review and meta-analysis on diagnostic accuracy of hypoglycaemia detection. Systematic reviews. 2021; 10: 1-14.

289. Garg SK. Glucose monitoring: an important tool for improving glucose control and reducing hypoglycemia. Diabetes Technol Ther. 2008; 10: S-1-S-4.

290. Schnell O, Barnard K, Bergenstal R, et al. Clinical utility of SMBG: recommendations on the use and reporting of SMBG in clinical research. Diabetes Care. 2015; 38: 1627-1633.

291. Ng C, Lai P, Lee Y, et al. Barriers and facilitators to starting insulin in patients with type 2 diabetes: a systematic review. Int J Clin Pract. 2015; 69: 1050-1070.

292. Saha S. Compliance and barriers to self-monitoring of blood glucose in patients with gestational diabetes mellitus: A systematic review. International journal of health sciences. 2019; 13: 44.

293. Zheng M, Luo Y, Lin W, et al. Comparing effects of continuous glucose monitoring systems (CGMs) and self-monitoring of blood glucose (SMBG) amongst adults with type 2 diabetes mellitus: a systematic review protocol. Systematic reviews. 2020; 9: 1-6.

294. Hayes R, Fitzgerald J, Jacober S. Primary care physician beliefs about insulin initiation in patients with type 2 diabetes. Int J Clin Pract. 2008; 62: 860-868.

295. Fagherazzi G, Ravaud P. Digital diabetes: Perspectives for diabetes prevention, management and research. Diabetes Metab. 2019; 45: 322-329.

296. Nyomba BG, Berard L, Murphy LJ. The cost of self-monitoring of blood glucose is an important factor limiting glycemic control in diabetic patients. Diabetes Care. 2002; 25: 1244-1245.

297. Shrivastava SR, Shrivastava PS, Ramasamy J. Role of self-care in management of diabetes mellitus. J Diabetes Metab Disord. 2013; 12: 14.

298. Selvin E, Wang D, Lee AK, et al. Identifying trends in undiagnosed diabetes in US adults by using a confirmatory definition: a cross-sectional study. Ann Intern Med. 2017; 167: 769-776.

299. Devine F, Edwards T, Feldman SR. Barriers to treatment: describing them from a different perspective. Patient preference and adherence. 2018; 12: 129.

300. Davies MJ, D'Alessio DA, Fradkin J, et al. Management of hyperglycaemia in type 2 diabetes, 2018. A consensus report by the American Diabetes Association (ADA) and the European Association for the Study of Diabetes (EASD). Diabetologia. 2018; 61: 2461-2498.

301. Patton SR. Adherence to glycemic monitoring in diabetes. J Diabetes Sci Technol. 2015; 9: 668-675.

302. Peyrot M, Rubin R, Lauritzen T, et al. Psychosocial problems and barriers to improved diabetes management: results of the Cross-National Diabetes Attitudes, Wishes and Needs (DAWN) Study. Diabet Med. 2005; 22: 1379-1385.

303. Simmons JH, Chen V, Miller KM, et al. Differences in the management of type 1 diabetes among adults under excellent control compared with those under poor control in the T1D Exchange Clinic Registry. Diabetes Care. 2013; 36: 3573-3577.

304. Harris MI. Frequency of blood glucose monitoring in relation to glycemic control in patients with type 2 diabetes. Diabetes Care. 2001; 24: 979-982.

305. Ong WM, Chua SS, Ng CJ. Barriers and facilitators to self-monitoring of blood glucose in people with type 2 diabetes using insulin: a qualitative study. Patient preference and adherence. 2014; 8: 237-246.

306. Vashist SK. Non-invasive glucose monitoring technology in diabetes management: A review. Anal Chim Acta. 2012; 750: 16-27.

307. Tanaka N, Yabe D, Murotani K, et al. Mental distress and health-related quality of life among type 1 and type 2 diabetes patients using self-monitoring of blood glucose: A cross-sectional questionnaire study in Japan. J Diabetes Invest. 2018; 9: 1203-1211.

308. Vincze G, Barner JC, Lopez D. Factors associated with adherence to self-monitoring of blood glucose among persons with diabetes. The Diabetes Educator. 2004; 30: 112-125.

309. Bergenstal RM, Mullen DM, Strock E, et al. Randomized comparison of self-monitored blood glucose (BGM) versus continuous glucose monitoring (CGM) data to optimize glucose control in type 2 diabetes. J Diabetes Complications. 2022; 36: 108106.

310. Dicembrini I, Cosentino C, Monami M, et al. Effects of real-time continuous glucose monitoring in type 1 diabetes: a meta-analysis of randomized controlled trials. Acta Diabetol. 2021; 58: 401-410.

311. Block CD, Vertommen J, Manuel-y-Keenoy B, et al. Minimally-invasive and non-invasive continuous glucose monitoring systems: indications, advantages, limitations and clinical aspects. Curr Diabetes Rev. 2008; 4: 159-168.

312. Poolsup N, Suksomboon N, Kyaw AM. Systematic review and meta-analysis of the effectiveness of continuous glucose monitoring (CGM) on glucose control in diabetes. Diabetol Metab Syndr. 2013; 5: 1-14.

313. Pencic-Popovic B, Celic V, Stojcevski B, et al. Heart rate variability and increased risk for developing type 2 diabetes mellitus. Vojnosanit Pregl. 2014; 71.

314. Kilpatrick CR, Elliott MB, Pratt E, et al. Prevention of inpatient hypoglycemia with a real-time informatics alert. J Hosp Med. 2014; 9: 621-626.

315. Rice MJ, Coursin DB, Riou B. Continuous measurement of glucose: facts and challenges. Journal of the American Society of Anesthesiologists. 2012; 116: 199-204.

316. Srinivasan V, Pamula VK, Pollack MG, et al., editors. Clinical diagnostics on human whole blood, plasma, serum, urine, saliva, sweat, and tears on a digital microfluidic platform. Proc MicroTAS; 2003.

317. Park HD, Lee KJ, Yoon HR, et al. Design of a portable urine glucose monitoring system for health care. Comput Biol Med. 2005; 35: 275-286.

318. Lin T, Gal A, Mayzel Y, et al. Non-invasive glucose monitoring: a review of challenges and recent advances. Curr Trends Biomed Eng Biosci. 2017; 6: 1-8.

319. do Amaral CEF, Wolf B. Current development in non-invasive glucose monitoring. Med Eng Phys. 2008; 30: 541-549.

320. Oliver N, Toumazou C, Cass A, et al. Glucose sensors: a review of current and emerging technology. Diabet Med. 2009; 26: 197-210.

321. Olesberg JT, Liu L, Zee VV, et al. In vivo near-infrared spectroscopy of rat skin tissue with varying blood glucose levels. Anal Chem. 2006; 78: 215-223.

322. Lakowicz JR, Masters BR. Principles of fluorescence spectroscopy. Journal of Biomedical Optics. 2008; 13: 029901.

323. March W, Lazzaro D, Rastogi S. Fluorescent measurement in the non-invasive contact lens glucose sensor. Diabetes Technol Ther. 2006; 8: 312-317.

324. Tronstad C, Elvebakk O, Staal OM, et al. Non-invasive prediction of blood glucose trends during hypoglycemia. Anal Chim Acta. 2019; 1052: 37-48.

325. Leboulanger B, Guy RH, Delgado-Charro MB. Reverse iontophoresis for non-invasive transdermal monitoring. Physiol Meas. 2004; 25: R35.

326. Singh P, Maibach HI. Iontophoresis in drug delivery: basic principles and applications. Crit Rev Ther Drug Carrier Syst. 1994; 11: 161-213.

327. Potts RO, A Tamada J, J Tierney M. Glucose monitoring by reverse iontophoresis. Diabetes Metab Res Rev. 2002; 18.

328. Shokrekhodaei M, Quinones S. Review of non-invasive glucose sensing techniques: optical, electrical and breath acetone. Sensors. 2020; 20: 1251.

329. Bertachi A, Biagi L, Contreras I, et al., editors. Prediction of Blood Glucose Levels And Nocturnal Hypoglycemia Using Physiological Models and Artificial Neural Networks. Institut d'Informatica i Aplicacions; 2018.

330. Fralick M, Dai D, Pou-Prom C, et al. Using Machine Learning to Predict Severe Hypoglycemia in Hospital. Diabetes Obes Metab. 2021.

331. Pérez-Gandía C, Facchinetti A, Sparacino G, et al. Artificial neural network algorithm for online glucose prediction from continuous glucose monitoring. Diabetes Technol Ther. 2010; 12: 81-88.

332. Mercan ÖB, Doğan V, Kılıç V, editors. Time Series Analysis based Machine Learning Classification for Blood Sugar Levels. 2020 Medical Technologies Congress (TIPTEKNO); 2020: IEEE.

333. Hardin DS, Anderson W, Cattet J. Dogs can be successfully trained to alert to hypoglycemia samples from patients with type 1 diabetes. Diabetes Ther. 2015; 6: 509-517.

334. Los EA, Ramsey KL, Guttmann-Bauman I, et al. Reliability of trained dogs to alert to hypoglycemia in patients with type 1 diabetes. J Diabetes Sci Technol. 2017; 11: 506-512.

335. Jung M, editor Toward designing mobile software to predict hypoglycemia for patients with diabetes. 2016 IEEE/ACM International Conference on Mobile Software Engineering and Systems (MOBILESoft); 2016: IEEE.

336. Ruan Y, Bellot A, Moysova Z, et al. Predicting the risk of inpatient hypoglycemia with machine learning using electronic health records. Diabetes Care. 2020; 43: 1504-1511.

337. Lama L, Wilhelmsson O, Norlander E, et al. Machine learning for prediction of diabetes risk in middle-aged Swedish people. Heliyon. 2021; 7: e07419.

338. Foss-Freitas MC, Moreira GS, Antloga VP, et al. 930-P: Blood Glucose Levels Prediction Accuracy for T1DM Patients Using Neural Networks to Combine Insulin Doses, Food Nutrients, and Heart Rate. Am Diabetes Assoc; 2019.

339. Alonge KM, D'Alessio DA, Schwartz MW. Brain control of blood glucose levels: implications for the pathogenesis of type 2 diabetes. Diabetologia. 2021; 64: 5-14.

340. Bear MF, Connors BW, Paradiso MA. Neuroscience: exploring the brain. Fourth ed: Wolters Kluwer; 2016.

341. McCorry LK. Physiology of the autonomic nervous system. Am J Pharm Educ. 2007; 71.

342. Garfinkel SN, Critchley HD. Threat and the Body: How the Heart Supports Fear Processing. Trends Cogn Sci. 2016; 20: 34-46.

343. Scarpa A. Physiological arousal and its dysregulation in child maladjustment. Curr Dir Psychol Sci. 2015; 24: 345-351.

344. Thayer JF, Lane RD. A model of neurovisceral integration in emotion regulation and dysregulation. J Affect Disord. 2000; 61: 201-216.

345. Werth B, Meyer-Wyss B, Spinas G, et al. Non-invasive assessment of gastrointestinal motility disorders in diabetic patients with and without cardiovascular signs of autonomic neuropathy. Gut. 1992; 33: 1199-1203.

346. Holst JJ, Schwartz TW, Knuhtsen S, et al. Autonomic nervous control of the endocrine secretion from the isolated, perfused pig pancreas. J Auton Nerv Syst. 1986; 17: 71-84.

347. Laborde S, Mosley E, Thayer JF. Heart rate variability and cardiac vagal tone in psychophysiological research–recommendations for experiment planning, data analysis, and data reporting. Front Psychol. 2017; 8: 213.

348. Huikuri HV, Stein PK. Clinical application of heart rate variability after acute myocardial infarction. Front Physiol. 2012; 3.

349. Stoickov V, Ilic S, Ilic M, et al. Impact of diabetes on heart rate variability and left ventricular function in patients after myocardial infarction. Facta Universitatis, Series: Medicine and Biology. 2005; 12: 130-134.

350. Tarvainen MP, Laitinen TP, Lipponen JA, et al. Cardiac autonomic dysfunction in type 2 diabetes-effect of hyperglycemia and disease duration. Front Endocrinol (Lausanne). 2014; 5: 130.

351. Zygmunt A, Stanczyk J. Methods of evaluation of autonomic nervous system function. Arch Med Sci. 2010; 6: 11-18.

352. Thayer JF, Yamamoto SS, Brosschot JF. The relationship of autonomic imbalance, heart rate variability and cardiovascular disease risk factors. Int J Cardiol. 2010; 141: 122-131.

353. Neff RA, Wang J, Baxi S, et al. Respiratory sinus arrhythmia: endogenous activation of nicotinic receptors mediates respiratory modulation of brainstem cardioinhibitory parasympathetic neurons. Circ Res. 2003; 93: 565-572.

354. ZÍTEK P, BÍLA PK. Establishing Heart Rate Variability as. Computational Intelligence for Modelling, Control & Automation: Neural Networks & Advanced Control Strategies. 1999; 1: 305.

355. Berntson GG, Cacioppo JT, Quigley KS. Respiratory sinus arrhythmia: autonomic origins, physiological mechanisms, and psychophysiological implications. Psychophysiology. 1993; 30: 183-196.

356. Angelone A, Coulter JR NA. Respiratory sinus arrhythmia: a frequency dependent phenomenon. J Appl Physiol. 1964; 19: 479-482.

357. Goldberger AL. Is the normal heartbeat chaotic or homeostatic? Physiology. 1991; 6: 87-91.

358. Shaffer F, Ginsberg J. An overview of heart rate variability metrics and norms. Front Pub Health. 2017; 5: 258.

359. Parati G, Ochoa JE, Lombardi C, et al. Assessment and management of blood-pressure variability. Nature Reviews Cardiology. 2013; 10: 143-155.

360. Parati G, Stergiou GS, Dolan E, et al. Blood pressure variability: clinical relevance and application. The Journal of Clinical Hypertension. 2018; 20: 1133-1137.

361. Thomas BL, Claassen N, Becker P, et al. Validity of commonly used heart rate variability markers of autonomic nervous system function. Neuropsychobiology. 2019; 78: 14-26.

362. McCraty R, Shaffer F. Heart rate variability: new perspectives on physiological mechanisms, assessment of self-regulatory capacity, and health risk. Global advances in health and medicine. 2015; 4: 46-61.

363. Cygankiewicz I, Zareba W. Heart rate variability. Handb Clin Neurol. 2013; 117: 379-393.

364. Hage FG, Iskandrian AE. Cardiac autonomic denervation in diabetes mellitus. Circ Cardiovasc Imaging. 2011; 4: 79-81.

365. Pop-Busui R. Cardiac autonomic neuropathy in diabetes a clinical perspective. Diabetes Care. 2010; 33: 434-441.

366. Ortiz Guzmán JE, Mendoza Romero D, Calderón CA, et al. Spectral analysis of heart rate variability in trained young men: comparison of endurance and resistance training. Apunts Medicina de l" Esport (English Edition). 2012; 47: 41-47.

367. Fleischer J, Cichosz SL, Hoeyem P, et al. Glycemic Variability Is Associated With Reduced Cardiac Autonomic Modulation in Women With Type 2 Diabetes. Diabetes Care. 2015; 38: 682-688.

368. Meyer C, Grossmann R, Mitrakou A, et al. Effects of autonomic neuropathy on counterregulation and awareness of hypoglycemia in type 1 diabetic patients. Diabetes Care. 1998; 21: 1960-1966.

369. Duque A, Mediano MFF, De Lorenzo A, et al. Cardiovascular autonomic neuropathy in diabetes: Pathophysiology, clinical assessment and implications. World J Diabetes. 2021; 12: 855.

370. Ang L, Cowdin N, Mizokami-Stout K, et al. Update on the management of diabetic neuropathy. Diabetes Spectr. 2018; 31: 224-233.

371. Tarvainen MP, Cornforth DJ, Kuoppa P, et al., editors. Complexity of heart rate variability in type 2 diabetes-effect of hyperglycemia. 35th Annual International Conference of the IEEE EMBS; 2013; Osaka, Japan: IEEE Press.

372. Dagliati A, Marini S, Sacchi L, et al. Machine learning methods to predict diabetes complications. J Diabetes Sci Technol. 2018; 12: 295-302.

373. Jaiswal M, Urbina EM, Wadwa RP, et al. Reduced heart rate variability among youth with type 1 diabetes The SEARCH CVD study. Diabetes Care. 2013; 36: 157-162.

374. Javorka M, Javorková J, Tonhajzerová I, et al. Heart rate variability in young patients with diabetes mellitus and healthy subjects explored by Poincaré and sequence plots. Clin Physiol Funct Imaging. 2005; 25: 119-127.

375. Kudat H, Akkaya V, Sozen A, et al. Heart rate variability in diabetes patients. J Int Med Res. 2006; 34: 291-296.

376. Mirza M, Lakshmi A. A comparative study of heart rate variability in diabetic subjects and normal subjects. Int J Biomed Adv Res. 2012; 3: 640-644.

377. Brock C, Jessen N, Brock B, et al. Cardiac vagal tone, a non-invasive measure of parasympathetic tone, is a clinically relevant tool in type 1 diabetes mellitus. Diabet Med. 2017; 34: 1428-1434.

378. Wegeberg A-M, Lunde ED, Riahi S, et al. Cardiac vagal tone as a novel screening tool to recognize asymptomatic cardiovascular autonomic neuropathy: Aspects of utility in type 1 diabetes. Diabetes Res Clin Pract. 2020; 170: 108517.

379. Buccelletti E, Gilardi E, Scaini E, et al. Heart rate variability and myocardial infarction: systematic literature review and metanalysis. Eur Rev Med Pharmacol Sci. 2009; 13: 299-307.

380. Deshmukh T, Emerson P, Anderson P, et al. Cardiac autonomic neuropathy is not reversed by euglycemia following islet transplantation. Transplantation. 2021; 105: 1125-1129.

381. Burr RL. Interpretation of normalized spectral heart rate variability indices in sleep research: a critical review. Sleep. 2007; 30: 913-919.

382. Pichon A, Roulaud M, Antoine-Jonville S, et al. Spectral analysis of heart rate variability: interchangeability between autoregressive analysis and fast Fourier transform. J Electrocardiol. 2006; 39: 31-37.

383. Rothberg LJ, Lees T, Clifton-Bligh R, et al. Association between heart rate variability measures and blood glucose levels: implications for noninvasive glucose monitoring for diabetes. Diabetes Technol Ther. 2016; 18: 366-376.

384. Lee CMY, Goode B, Nørtoft E, et al. The cost of diabetes and obesity in Australia. J Med Econ. 2018; 21: 1001-1005.

385. Mozaffarian D. Dietary and policy priorities for cardiovascular disease, diabetes, and obesity: a comprehensive review. Circulation. 2016; 133: 187-225.

386. Al Busaidi N, Shanmugam P, Manoharan D. Diabetes in the Middle East: government health care policies and strategies that address the growing diabetes prevalence in the Middle East. Curr Diab Rep. 2019; 19: 8.

387. Greven WL, Beulens JW, Biesma DH, et al. Glycemic variability in inadequately controlled type 1 diabetes and type 2 diabetes on intensive insulin therapy: a cross-sectional, observational study. Diabetes Technol Ther. 2010; 12: 695-699.

388. Mäkimattila S, Schlenzka A, Mäntysaari M, et al. Predictors of abnormal cardiovascular autonomic function measured by frequence domain analysis of heart rate variability and conventional tests in patients with type 1 diabetes. Diabetes Care. 2000; 23: 1686-1693.

389. Jarman L, Elliott J, Lees T, et al. Heart Rate Variability as a Potential Non-invasive Marker of Blood Glucose Level. Hum Physiol. 2021; 47: 209-218.

390. D'alessio DA, Kieffer TJ, Taborsky Jr GJ, et al. Activation of the parasympathetic nervous system is necessary for normal meal-induced insulin secretion in rhesus macaques. J Clin Endocrinol Metab. 2001; 86: 1253-1259.

391. Havel PJ, Veith RC, Dunning BE, et al. Pancreatic noradrenergic nerves are activated by neuroglucopenia but not by hypotension or hypoxia in the dog. Evidence for stress-specific and regionally selective activation of the sympathetic nervous system. The Journal of clinical investigation. 1988; 82: 1538-1545.

392. Singh JP, Larson MG, O'Donnell CJ, et al. Association of hyperglycemia with reduced heart rate variability (The Framingham Heart Study). Am J Cardiol. 2000; 86: 309-312.

393. Weissman A, Lowenstein L, Peleg A, et al. Power spectral analysis of heart rate variability during the 100-g oral glucose tolerance test in pregnant women. Diabetes Care. 2006; 29: 571-574.

394. Stein P, Barzilay J, Domitrovich P, et al. The relationship of heart rate and heart rate variability to non-diabetic fasting glucose levels and the metabolic syndrome: The Cardiovascular Health Study. Diabet Med. 2007; 24: 855-863.

395. Jarczok MN, Li J, Mauss D, et al. Heart rate variability is associated with glycemic status after controlling for components of the metabolic syndrome. Int J Cardiol. 2013; 167: 855-861.

396. Klimontov VV, Myakina NE, Tyan NV. Heart rate variability is associated with interstitial glucose fluctuations in type 2 diabetic women treated with insulin. SpringerPlus. 2016; 5: 1-9.

397. Lutfi MF, Elhakeem RF. Effect of fasting blood glucose level on heart rate variability of healthy young adults. PLoS One. 2016; 11.

398. Hayano J, Yuda E. Pitfalls of assessment of autonomic function by heart rate variability. J Physiol Anthropol. 2019; 38: 1-8.

399. Cichosz SL, Fleischer J, Hoeyem P, et al. Assessment of postprandial glucose excursions throughout the day in newly diagnosed type 2 diabetes. Diabetes Technol Ther. 2013; 15: 78-83.

400. Ceriello A, Ihnat M. 'Glycaemic variability': a new therapeutic challenge in diabetes and the critical care setting. Diabet Med. 2010; 27: 862-867.

401. Ogawa A, Hayashi A, Kishihara E, et al. New indices for predicting glycaemic variability. 2012.

402. Freckmann G, Hagenlocher S, Baumstark A, et al. Continuous glucose profiles in healthy subjects under everyday life conditions and after different meals. J Diabetes Sci Technol. 2007; 1: 695-703.

403. Vilsbøll T, Krarup T, Madsbad S, et al. Both GLP-1 and GIP are insulinotropic at basal and postprandial glucose levels and contribute nearly equally to the incretin effect of a meal in healthy subjects. Regul Pept. 2003; 114: 115-121.

404. Emerging Risk Factors Collaboration. Diabetes mellitus, fasting blood glucose concentration, and risk of vascular disease: a collaborative meta-analysis of 102 prospective studies. The Lancet. 2010; 375: 2215-2222.

405. Pickering TG, Hall JE, Appel LJ, et al. Recommendations for blood pressure measurement in humans and experimental animals. Circulation. 2005; 111: 697-716.

406. World Health Organization. Definition and Diagnosis of Diabetes Mellitus and Intermediate Hyperglycemia. Geneva, Switzerland: World Health Organization; 2006.

407. Kardelen F, Akçurin G, Ertuğ H, et al. Heart rate variability and circadian variations in type 1 diabetes mellitus. Pediatr Diabetes. 2006; 7: 45-50.

408. Freckmann G, Schmid C, Baumstark A, et al. System accuracy evaluation of 43 blood glucose monitoring systems for self-monitoring of blood glucose according to DIN EN ISO 15197. J Diabetes Sci Technol. 2012; 6: 1060-1075.

409. Akintola AA, van de Pol V, Bimmel D, et al. Comparative analysis of the equivital EQ02 lifemonitor with holter ambulatory ECG device for continuous measurement of ECG, heart rate, and heart rate variability: A validation study for precision and accuracy. Front Physiol. 2016; 7: 391.

410. Kinnunen H, Rantanen A, Kenttä T, et al. Feasible assessment of recovery and cardiovascular health: accuracy of nocturnal HR and HRV assessed via ring PPG in comparison to medical grade ECG. Physiol Meas. 2020; 41: 04NT01.

411. Kim H-G, Cheon E-J, Bai D-S, et al. Stress and heart rate variability: a meta-analysis and review of the literature. Psychiatry Investig. 2018; 15: 235.

412. Electrophysiology Task Force of the European Society of Cardiology the North American Society of Pacing. Heart rate variability: standards of measurement, physiological interpretation, and clinical use. Circulation. 1996; 93: 1043-1065.

413. Combatalade DC. Basics of Heart Rate Variability Applied to Psychophysiology. Montreal, Canada; 2010.

414. Melzig CA, Weike AI, Hamm AO, et al. Individual differences in fear-potentiated startle as a function of resting heart rate variability: implications for panic disorder. Int J Psychophysiol. 2009; 71: 109-117.

415. McNames J, Aboy M. Reliability and accuracy of heart rate variability metrics versus ECG segment duration. Med Biol Eng Comput. 2006; 44: 747-756.

416. Singh D, Vinod K, Saxena SC, et al. Effects of RR segment duration on HRV spectrum estimation. Physiol Meas. 2004; 25: 721.

417. Tarvainen MP, Niskanen J-P, Lipponen JA, et al. Kubios HRV–heart rate variability analysis software. Comput Methods Programs Biomed. 2014; 113: 210-220.

418. Chemla D, Young J, Badilini F, et al. Comparison of fast Fourier transform and autoregressive spectral analysis for the study of heart rate variability in diabetic patients. Int J Cardiol. 2005; 104: 307-313.

419. Kanaley JA, Baynard T, Franklin RM, et al. The effects of a glucose load and sympathetic challenge on autonomic function in obese women with and without type 2 diabetes mellitus. Metabolism. 2007; 56: 778-785.

420. Poletto R, Janczak AM, Marchant-Forde RM, et al. Identification of low and high frequency ranges for heart rate variability and blood pressure variability analyses using pharmacological autonomic blockade with atropine and propranolol in swine. Physiol Behav. 2011; 103: 188-196.

421. ChuDuc H, NguyenPhan K, NguyenViet D. A review of heart rate variability and its applications. APCBEE Procedia. 2013; 7: 80-85.

422. Acharya UR, Joseph KP, Kannathal N, et al. Heart rate variability: a review. Med Biol Eng Comput. 2006; 44: 1031-1051.

423. Kobayashi H, Park B-J, Miyazaki Y. Normative references of heart rate variability and salivary alpha-amylase in a healthy young male population. J Physiol Anthropol. 2012; 31: 9.

424. Ramaekers D, Ector H, Aubert AE, et al. Heart rate variability and heart rate in healthy volunteers. Is the female autonomic nervous system cardioprotective? Eur Heart J. 1998; 19: 1334-1341.

425. Farshchi H, Taylor M, Macdonald I. Regular meal frequency creates more appropriate insulin sensitivity and lipid profiles compared with irregular meal frequency in healthy lean women. Eur J Clin Nutr. 2004; 58: 1071.

426. Kilinc O, Cincin A, Pehlivan A, et al. Assessment of Time and Frequency Domain Parameters of Heart Rate Variability and Interictal Cardiac Rhythm Abnormalities in Drug-naïve Patients with Idiopathic Generalized Epilepsy. J Epilepsy Res. 2016; 6: 22.

427. Wolever TM, Chiasson J-L, Csima A, et al. Variation of postprandial plasma glucose, palatability, and symptoms associated with a standardized mixed test meal versus 75 g oral glucose. Diabetes Care. 1998; 21: 336-340.

428. Hlebowicz J, Darwiche G, Björgell O, et al. Effect of cinnamon on postprandial blood glucose, gastric emptying, and satiety in healthy subjects. Am J Clin Nutr. 2007; 85: 1552-1556.

429. Östman E, Granfeldt Y, Persson L, et al. Vinegar supplementation lowers glucose and insulin responses and increases satiety after a bread meal in healthy subjects. Eur J Clin Nutr. 2005; 59: 983.

430. Tsuneki H, Ishizuka M, Terasawa M, et al. Effect of green tea on blood glucose levels and serum proteomic patterns in diabetic (db/db) mice and on glucose metabolism in healthy humans. BMC Pharmacol. 2004; 4: 1-10.

431. Bryans JA, Judd PA, Ellis PR. The effect of consuming instant black tea on postprandial plasma glucose and insulin concentrations in healthy humans. J Am Coll Nutr. 2007; 26: 471-477.

432. Agelink MW, Malessa R, Baumann B, et al. Standardized tests of heart rate variability: normal ranges obtained from 309 healthy humans, and effects of age, gender, and heart rate. Clin Auton Res. 2001; 11: 99-108.

433. Zhang J. Effect of age and sex on heart rate variability in healthy subjects. J Manipulative Physiol Ther. 2007; 30: 374-379.

434. Umetani K, Singer DH, McCraty R, et al. Twenty-four hour time domain heart rate variability and heart rate: relations to age and gender over nine decades. J Am Coll Cardiol. 1998; 31: 593-601.

435. Lu C-L, Zou X, Orr WC, et al. Postprandial changes of sympathovagal balance measured by heart rate variability. Dig Dis Sci. 1999; 44: 857-861.

436. Soni Y, Jain JK, Meena R, et al., editors. HRV analysis of young adults in pre-meal and post-meal stage. 2017 2nd IEEE International Conference on Recent Trends in Electronics, Information & Communication Technology (RTEICT); 2017: IEEE.

437. Zwillich C, Martin B, Hofeldt F, et al. Lack of effects of beta sympathetic blockade on the metabolic and respiratory responses to carbohydrate feeding. Metabolism. 1981; 30: 451-456.

438. Choi CJ, Choi WS, Kim KS. The changes in heart rate variability between morning and afternoon. Korean journal of family medicine. 2008; 29: 579-584.

439. Armstrong RG, Kenny GP, Green G, et al. Diurnal variation in heart rate variability before and after maximal exercise testing. Chronobiol Int. 2011; 28: 344-351.

440. Paolisso G, Manzella D, Ferrara N, et al. Glucose ingestion affects cardiac ANS in healthy subjects with different amounts of body fat. Am J Physiol Endocrinol Metab. 1997; 273: E471.

441. Tentolouris N, Tsigos C, Perea D, et al. Differential effects of high-fat and high-carbohydrate isoenergetic meals on cardiac autonomic nervous system activity in lean and obese women. Metabolism. 2003; 52: 1426-1432.

442. Nuttall FQ, Gannon MC, Wald JL, et al. Plasma glucose and insulin profiles in normal subjects ingesting diets of varying carbohydrate, fat, and protein content. J Am Coll Nutr. 1985; 4: 437-450.

443. Rosén LA, Silva LOB, Andersson UK, et al. Endosperm and whole grain rye breads are characterized by low post-prandial insulin response and a beneficial blood glucose profile. Nutr J. 2009; 8: 1-11.

444. Brand-Miller JC, Stockmann K, Atkinson F, et al. Glycemic index, postprandial glycemia, and the shape of the curve in healthy subjects: analysis of a database of more than 1000 foods. Am J Clin Nutr. 2009; 89: 97-105.

445. Morgan LM, Shi J-W, Hampton SM, et al. Effect of meal timing and glycaemic index on glucose control and insulin secretion in healthy volunteers. Br J Nutr. 2012; 108: 1286-1291.

446. Esposito K, Ciotola M, Carleo D, et al. Post-meal glucose peaks at home associate with carotid intima-media thickness in type 2 diabetes. J Clin Endocrinol Metab. 2008; 93: 1345-1350.

447. Tsukuda K, Kikuchi M, Irie S, et al. Evaluation of the 24-hour profiles of physiological insulin, glucose, and C-peptide in healthy Japanese volunteers. Diabetes Technol Ther. 2009; 11: 499-508.

448. Elizondo-Montemayor L, Hernández-Brenes C, Ramos-Parra PA, et al. High hydrostatic pressure processing reduces the glycemic index of fresh mango puree in healthy subjects. Food Funct. 2015; 6: 1352-1360.

449. Hätönen KA, Virtamo J, Eriksson JG, et al. Protein and fat modify the glycaemic and insulinaemic responses to a mashed potato-based meal. Br J Nutr. 2011; 106: 248-253.

450. Foreman YD, Brouwers MC, Berendschot TT, et al. The oral glucose tolerance test-derived incremental glucose peak is associated with greater arterial stiffness and maladaptive arterial remodeling: the Maastricht Study. Cardiovasc Diabetol. 2019; 18: 1-12.

451. Nilsson A, Radeborg K, Björck I. Effects on cognitive performance of modulating the postprandial blood glucose profile at breakfast. Eur J Clin Nutr. 2012; 66: 1039-1043.

452. Cichosz SL, Frystyk J, Hejlesen OK, et al. A novel algorithm for prediction and detection of hypoglycemia based on continuous glucose monitoring and heart rate variability in patients with type 1 diabetes. J Diabetes Sci Technol. 2014; 8: 731-737.

453. Gill G, Woodward A, Casson I, et al. Cardiac arrhythmia and nocturnal hypoglycaemia in type 1 diabetes—the 'dead in bed'syndrome revisited. Diabetologia. 2009; 52: 42.

454. Craig A, Hancock K, Craig M. The lifestyle appraisal questionnaire: A comprehensive assessment of health and stress. Psychology and Health. 1996; 11: 331-343.

455. Dobbs WC, Fedewa MV, MacDonald HV, et al. The accuracy of acquiring heart rate variability from portable devices: a systematic review and meta-analysis. Sports Med. 2019; 49: 417-435.

456. Georgiou K, Larentzakis AV, Khamis NN, et al. Can wearable devices accurately measure heart rate variability? A systematic review. Folia medica. 2018; 60: 7-20.

457. Jain JK, Maheshwari R. Correlation Between Poincare Plot Indices and Linear–Nonlinear Heart Rate Variability During Fasting and Postprandial States. Applications of Computing, Automation and Wireless Systems in Electrical Engineering: Springer; 2019. p. 881-889. 458. Omech B, Tshikuka J-G, Amone-P'Olak K, et al. The utilization of HbA1c test in the screening and diagnosis of type 2 diabetes mellitus: An outpatient clinics survey in Botswana. Journal of Diabetes and Endocrinology. 2017; 8: 1-8.

459. Jensen-Urstad K, Storck N, Bouvier F, et al. Heart rate variability in healthy subjects is related to age and gender. Acta Physiol Scand. 1997; 160: 235-241.

460. O'Brien I, O'Hare P, Corrall R. Heart rate variability in healthy subjects: effect of age and the derivation of normal ranges for tests of autonomic function. Br Heart J. 1986; 55: 348-354.

461. Huo L, Magliano DJ, Rancière F, et al. Impact of age at diagnosis and duration of type 2 diabetes on mortality in Australia 1997–2011. Diabetologia. 2018; 61: 1055-1063.

462. Nanayakkara N, Ranasinha S, Gadowski A, et al. Age, age at diagnosis and diabetes duration are all associated with vascular complications in type 2 diabetes. J Diabetes Complications. 2018; 32: 279-290.

463. Knowles SR, Apputhurai P, O'Brien CL, et al. Exploring the relationships between illness perceptions, self-efficacy, coping strategies, psychological distress and quality of life in a cohort of adults with diabetes mellitus. Psychol Health Med. 2020; 25: 214-228.

464. Kim MK, Han K, Park Y-M, et al. Associations of variability in blood pressure, glucose and cholesterol concentrations, and body mass index with mortality and cardiovascular outcomes in the general population. Circulation. 2018; 138: 2627-2637.

465. Khandoker AH, Al-Angari HM, Khalaf K, et al. Association of diabetes related complications with heart rate variability among a diabetic population in the UAE. PLoS One. 2017; 12: e0168584.

466. Imai C, Li L, Hardie R-A, et al. Adherence to guideline-recommended HbA1c testing frequency and better outcomes in patients with type 2 diabetes: a 5-year retrospective cohort study in Australian general practice. BMJ Quality & Safety. 2021.

467. Koenig J, Thayer JF. Sex differences in healthy human heart rate variability: a meta-analysis. Neurosci Biobehav Rev. 2016; 64: 288-310.

468. Kautzky-Willer A, Brazzale AR, Moro E, et al. Influence of increasing BMI on insulin sensitivity and secretion in normotolerant men and women of a wide age span. Obesity. 2012; 20: 1966-1973.

469. Faerch K, Borch-Johnsen K, Vaag A, et al. Sex differences in glucose levels: a consequence of physiology or methodological convenience? The Inter99 study. Diabetologia. 2010; 53: 858-865.

470. Dinger MK, Waigandt A. Dietary intake and physical activity behaviors of male and female college students. Am J Health Promot. 1997; 11: 360-362.

471. Pryer JA, Nichols R, Elliott P, et al. Dietary patterns among a national random sample of British adults. J Epidemiol Community Health. 2001; 55: 29-37.

472. WY G, MT MN, MS Z, et al. Differences in eating behaviours, dietary intake and body weight status between male and female Malaysian University students. Malays J Nutr. 2011; 17.

473. Liao D, Cai J, Brancati FL, et al. Association of vagal tone with serum insulin, glucose, and diabetes mellitus—The ARIC Study. Diabetes Res Clin Pract. 1995; 30: 211-221.

474. Benichou T, Pereira B, Mermillod M, et al. Heart rate variability in type 2 diabetes mellitus: A systematic review and meta–analysis. PLoS One. 2018; 13: e0195166.

475. Myakina NE, Klimontov VV, Safarov DM, editors. Glucose Fluctuations and Heart Rate Variability in Type 2 Diabetic Patients: The Results of Simultaneous Glucose and ECG Monitoring. Diabetes; 2015: American Diabetes Association.

476. Burchfiel CM, Sharp DS, Curb JD, et al. Physical activity and incidence of diabetes: the Honolulu Heart Program. Am J Epidemiol. 1995; 141: 360-368.

477. Franco M, Ordunez P, Caballero B, et al. Impact of energy intake, physical activity, and population-wide weight loss on cardiovascular disease and diabetes mortality in Cuba, 1980–2005. Am J Epidemiol. 2007; 166: 1374-1380.

478. Soares-Miranda L, Sattelmair J, Chaves P, et al. Physical activity and heart rate variability in older adults: the Cardiovascular Health Study. Circulation. 2014; 129: 2100-2110.

479. da Silva DF, Bianchini JAA, Antonini VDS, et al. Parasympathetic cardiac activity is associated with cardiorespiratory fitness in overweight and obese adolescents. Pediatr Cardiol. 2014; 35: 684-690.

480. Nederkoorn C, Smulders F, Jansen A. Cephalic phase responses, craving and food intake in normal subjects. Appetite. 2000; 35: 45-55.

481. Quilliot D, Zannad F, Ziegler O. Impaired response of cardiac autonomic nervous system to glucose load in severe obesity. Metabolism. 2005; 54: 966-974.

482. Rowe JW, Young JB, Minaker KL, et al. Effect of insulin and glucose infusions on sympathetic nervous system activity in normal man. Diabetes. 1981; 30: 219-225.

483. Zecchin C, Facchinetti A, Sparacino G, et al., editors. A new neural network approach for shortterm glucose prediction using continuous glucose monitoring time-series and meal information. 2011 Annual International Conference of the IEEE Engineering in Medicine and Biology Society; 2011: IEEE.

484. Xie J, Wang Q. Benchmarking Machine Learning Algorithms on Blood Glucose Prediction for Type I Diabetes in Comparison with Classical Time-Series Models. IEEE Trans Biomed Eng. 2020; 67: 3101-3124.

485. Rodríguez-Rodríguez I, Chatzigiannakis I, Rodríguez J-V, et al. Utility of big data in predicting short-term blood glucose levels in type 1 diabetes mellitus through machine learning techniques. Sensors. 2019; 19: 4482.

486. Sudharsan B, Peeples M, Shomali M. Hypoglycemia prediction using machine learning models for patients with type 2 diabetes. J Diabetes Sci Technol. 2014; 9: 86-90.

487. Parati G, Saul JP, Di Rienzo M, et al. Spectral analysis of blood pressure and heart rate variability in evaluating cardiovascular regulation: a critical appraisal. Hypertension. 1995; 25: 1276-1286.

488. Rossi P, Andriesse G, Oey P, et al. Stomach distension increases efferent muscle sympathetic nerve activity and blood pressure in healthy humans. J Neurol Sci. 1998; 161: 148-155.

489. Cichosz SL, Frystyk J, Tarnow L, et al. Are changes in heart rate variability during hypoglycemia confounded by the presence of cardiovascular autonomic neuropathy in patients with diabetes? Diabetes Technol Ther. 2017; 19: 91-95.

490. Tsuji H, Larson MG, Venditti FJ, et al. Impact of reduced heart rate variability on risk for cardiac events: the Framingham Heart Study. Circulation. 1996; 94: 2850-2855.

491. Bekkink MO, Koeneman M, de Galan BE, et al. Early detection of hypoglycemia in type 1 diabetes using heart rate variability measured by a wearable device. Diabetes Care. 2019; 42: 689-692.

492. Battelino T, Danne T, Bergenstal RM, et al. Clinical targets for continuous glucose monitoring data interpretation: recommendations from the international consensus on time in range. Diabetes Care. 2019; 42: 1593-1603.

493. Heller SR, Buse JB, Ratner R, et al. Redefining hypoglycemia in clinical trials: validation of definitions recently adopted by the American Diabetes Association/European Association for the Study of Diabetes. Diabetes Care. 2020; 43: 398-404.

494. Swinnen SG, Mullins P, Miller M, et al. Changing the glucose cut-off values that define hypoglycaemia has a major effect on reported frequencies of hypoglycaemia. Diabetologia. 2009; 52: 38.

495. American Diabetes Association. Defining and reporting hypoglycemia in diabetes: a report from the American Diabetes Association Workgroup on Hypoglycemia. Diabetes Care. 2005; 28: 1245-1249.

496. Bonds DE, Miller ME, Bergenstal RM, et al. The association between symptomatic, severe hypoglycaemia and mortality in type 2 diabetes: retrospective epidemiological analysis of the ACCORD study. BMJ. 2010; 340.

497. NICE-Sugar Study Investigators. Hypoglycemia and risk of death in critically ill patients. N Engl J Med. 2012; 367: 1108-1118.

498. ORIGIN Trial Investigators. Does hypoglycaemia increase the risk of cardiovascular events? A report from the ORIGIN trial. Eur Heart J. 2013; 34: 3137-3144.

499. Winterstein AG, Jeon N, Staley B, et al. Development and validation of an automated algorithm for identifying patients at high risk for drug-induced hypoglycemia. Am J Health Syst Pharm. 2018; 75: 1714-1728.

500. Heathers JA. Everything Hertz: methodological issues in short-term frequency-domain HRV. Front Physiol. 2014; 5: 177.

501. Burr RL. Interpretation of normalized spectral heart rate variability indices in sleep research: a critical review. Sleep. 2007; 30: 913-919.

502. Wang H-M, Huang S-C. SDNN/RMSSD as a surrogate for LF/HF: a revised investigation. Mod Sim Eng. 2012; 2012.

503. Guo Y, Logan HL, Glueck DH, et al. Selecting a sample size for studies with repeated measures. BMC Med Res Methodol. 2013; 13: 1-8.

504. Muller KE, Barton CN. Approximate power for repeated-measures ANOVA lacking sphericity. Journal of the American Statistical Association. 1989; 84: 549-555.

505. Cohen J. A power primer. Psychol Bull. 1992; 112: 155.

506. Cichosz SL, Frystyk J, Tarnow L, et al. Combining information of autonomic modulation and CGM measurements enables prediction and improves detection of spontaneous hypoglycemic events. J Diabetes Sci Technol. 2014; 9: 132-137.

507. Muntner P, Shimbo D, Carey RM, et al. Measurement of blood pressure in humans: a scientific statement from the American Heart Association. Hypertension. 2019; 73: e35-e66.

508. Lundqvist MH, Almby K, Wiklund U, et al. Altered hormonal and autonomic nerve responses to hypo-and hyperglycaemia are found in overweight and insulin-resistant individuals and may contribute to the development of type 2 diabetes. Diabetologia. 2021; 64: 641-655.

509. Amiel S, Dixon T, Mann R, et al. Hypoglycaemia in type 2 diabetes. Diabet Med. 2008; 25: 245-254.

510. Group IHS. Glucose concentrations of less than 3.0 mmol/L (54 mg/dL) should be reported in clinical trials: a joint position statement of the American Diabetes Association and the European Association for the Study of Diabetes. Diabetes Care. 2017; 40: 155-157.

511. Niijima A. Neural mechanisms in the control of blood glucose concentration. J Nutr. 1989; 119: 833-840.

512. Towler DA, Havlin CE, Craft S, et al. Mechanism of awareness of hypoglycemia: perception of neurogenic (predominantly cholinergic) rather than neuroglycopenic symptoms. Diabetes. 1993; 42: 1791-1798.

513. Majeed F, Yar T. Comparison of cardiovascular autonomic activity (heart rate variability and baroreceptor sensitivity) in young healthy females during fasting and hyperglycaemia. Diabetes Metab Syndr. 2020; 14: 1511-1518.

514. Bernjak A, Novodvorsky P, Chow E, et al. Cardiac arrhythmias and electrophysiologic responses during spontaneous hyperglycaemia in adults with type 1 diabetes mellitus. Diabetes Metab. 2021; 47: 101237.

515. Berkelaar M, Eekhoff E, Simonis-Bik A, et al. Effects of induced hyperinsulinaemia with and without hyperglycaemia on measures of cardiac vagal control. Diabetologia. 2013; 56: 1436-1443.

516. Marfella R, Nappo F, De Angelis L, et al. Hemodynamic effects of acute hyperglycemia in type 2 diabetic patients. Diabetes Care. 2000; 23: 658-663.

517. Marfella R, Nappo F, De Angelis L, et al. The effect of acute hyperglycaemia on QTc duration in healthy man. Diabetologia. 2000; 43: 571-575.

518. Legler M, Brandenberger G, Hietter B, et al. Diurnal cortisol peaks and their relationships to meals. J Clin Endocrinol Metab. 1982; 55: 757-761.

519. Reynolds RM, Walker BR, Syddall HE, et al. Elevated plasma cortisol in glucose-intolerant men: differences in responses to glucose and habituation to venepuncture. J Clin Endocrinol Metab. 2001; 86: 1149-1153.

520. Schäfer A, Vagedes J. How accurate is pulse rate variability as an estimate of heart rate variability?: A review on studies comparing photoplethysmographic technology with an electrocardiogram. Int J Cardiol. 2013; 166: 15-29.

521. Weippert M, Kumar M, Kreuzfeld S, et al. Comparison of three mobile devices for measuring R– R intervals and heart rate variability: Polar S810i, Suunto t6 and an ambulatory ECG system. Eur J Appl Physiol. 2010; 109: 779-786.

522. Porto LGG, Junqueira Jr LF. Comparison of time-domain short-term heart interval variability analysis using a wrist-worn heart rate monitor and the conventional electrocardiogram. Pacing Clin Electrophysiol. 2009; 32: 43-51.

523. Zhu T, Li K, Herrero P, et al., editors. A Deep Learning Algorithm for Personalized Blood Glucose Prediction. Department of Electronic and Electrical Engineering; 2018.

524. Li K, Daniels J, Liu C, et al. Convolutional recurrent neural networks for glucose prediction. IEEE J Biomedical Health Info. 2019; 24: 603-613.

525. Goldner DR, Osborn CY, Sears LE, et al. A machine-learning model accurately predicts projected blood glucose. Am Diabetes Assoc; 2018.

526. Pustozerov EA, Tkachuk AS, Vasukova EA, et al. Machine learning approach for postprandial blood glucose prediction in gestational diabetes mellitus. IEEE Access. 2020; 8: 219308-219321.

527. Wang W, Wang S, Wang X, et al. A Glucose-Insulin Mixture Model and Application to Short-Term Hypoglycemia Prediction in the Night Time. IEEE Trans Biomed Eng. 2020; 68: 834-845.

528. Cichosz SL, Johansen MD, Hejlesen O. Toward big data analytics: review of predictive models in management of diabetes and its complications. J Diabetes Sci Technol. 2016; 10: 27-34.